# $\mathbb{Z}/m\mathbb{Z}$ -GRADED LIE ALGEBRAS AND PERVERSE SHEAVES, III: GRADED DOUBLE AFFINE HECKE ALGEBRA

#### GEORGE LUSZTIG AND ZHIWEI YUN

ABSTRACT. In this paper we construct representations of certain graded double affine Hecke algebras (DAHA) with possibly unequal parameters from geometry. More precisely, starting with a simple Lie algebra  $\mathfrak{g}$  together with a  $\mathbb{Z}/m\mathbb{Z}$ -grading  $\bigoplus_{i \in \mathbb{Z}/m\mathbb{Z}} \mathfrak{g}_i$  and a block of  $\mathcal{D}_{G_{\underline{0}}}(\mathfrak{g}_i)$  as introduced in [J. Represent. Theory 21 (2017), pp. 277-321], we attach a graded DAHA and construct its action on the direct sum of spiral inductions in that block. This generalizes results of Vasserot [Duke Math J. 126 (2005), pp. 251-323] and Oblomkov-Yun [Adv. Math 292 (2016), pp. 601-706] which correspond to the case of the principal block.

#### CONTENTS

1.	Introduction	87
2.	Relative affine Weyl group and graded DAHA	92
3.	Facets and spirals	99
4.	The graded DAHA action on spiral inductions	106
References		118

#### 1. INTRODUCTION

#### 1.1. Background.

1.1.1. Let G be a simply-connected almost simple group over an algebraically closed field k with char(k) = 0. Let  $\mathfrak{g} = \text{Lie } G$ . Let  $m \geq 1$  be an integer and let

$$\mathfrak{g} = igoplus_{i \in \mathbb{Z}/m\mathbb{Z}} \mathfrak{g}_i$$

be a  $\mathbb{Z}/m\mathbb{Z}$ -grading on  $\mathfrak{g}$ . For an integer  $n \in \mathbb{Z}$ , let  $\underline{n}$  be its image in  $\mathbb{Z}/m\mathbb{Z}$ .

For the rest of the paper, we fix  $\eta \in \mathbb{Z} - \{0\}$ . Let  $G_{\underline{0}} \subset G$  be the connected reductive subgroup with Lie algebra  $\mathfrak{g}_{\underline{0}}$ . Let  $\mathfrak{g}_{\underline{\eta}}^{nil}$  be the cone of nilpotent elements in  $\mathfrak{g}_{\underline{\eta}}$ .<sup>1</sup> In the series of papers starting with [9] and [10], we are interested in the structure of the equivariant derived category  $\mathcal{D}_{G_{\underline{0}}}(\mathfrak{g}_{\eta}^{nil})$ . This paper mainly relies

Received by the editors April 2, 2017, and, in revised form, June 4, 2018.

<sup>2010</sup> Mathematics Subject Classification. Primary 20G99, 20C08.

The first author was partially supported by the NSF grant DMS-1566618.

The second author was supported by the NSF grant DMS-1302071 (with extension as DMS-1736600) and the Packard Foundation.

<sup>&</sup>lt;sup>1</sup>In [9, 10] we use  $\delta$  to denote  $\underline{\eta}$ ; in this paper we will reserve  $\delta$  for an imaginary root in an affine root system.

on  $[9, \S1-4]$  and  $[10, \S10]$ . We will use notation from [9], part of which will be reviewed in  $\S1.5$ .

1.1.2. Block decomposition and admissible systems. The main result of [9] is a direct sum decomposition of  $\mathcal{D}_{G_{\underline{0}}}(\mathfrak{g}_{\underline{\eta}}^{nil})$  into blocks in the style of the generalized Springer correspondence:

(1.1) 
$$\mathcal{D}_{G_{\underline{0}}}(\mathfrak{g}_{\underline{\eta}}^{nil}) = \bigoplus_{\xi \in \underline{\mathfrak{T}}_{\eta}} \mathcal{D}_{G_{\underline{0}}}(\mathfrak{g}_{\underline{\eta}}^{nil})_{\xi}.$$

The blocks are in bijection with the set  $\underline{\mathfrak{T}}_{\eta}$  of  $G_{\underline{0}}$ -conjugacy classes of *admissible* systems. Roughly speaking, an admissible system is a tuple  $\dot{\xi} = (M, M_0, \mathfrak{m}, \mathfrak{m}_*, \mathcal{L})$ , where M is a reductive subgroup of G which is the fixed point subgroup of a finiteorder automorphism of G (a pseudo-Levi subgroup, see §2.2),  $\mathfrak{m} = \bigoplus_{n \in \mathbb{Z}} \mathfrak{m}_n$  is a  $\mathbb{Z}$ -grading on the Lie algebra  $\mathfrak{m}$  of M such that  $\mathfrak{m}_n \subset \mathfrak{g}_n$ ,  $M_0$  is the connected reductive subgroup of M with Lie algebra  $\mathfrak{m}_0$ , and  $\mathcal{L}$  is an irreducible cuspidal  $M_0$ -equivariant local system (with  $\overline{\mathbb{Q}}_{\ell}$ -coefficients) on the open  $M_0$ -orbit  $\mathfrak{m}_\eta$  of  $\mathfrak{m}_\eta$ . For the precise definition of an admissible system, see [9, §3.1] or §3.4.8.

There is a unique  $G_{\underline{0}}$ -conjugacy class of admissible systems  $(M, M_0, \mathfrak{m}, \mathfrak{m}_*, \mathcal{L})$  in which M is a torus. In this case, M is a maximal torus in  $G_{\underline{0}}$ , the  $\mathbb{Z}$ -grading on  $\mathfrak{m}$  is concentrated in degree 0, and the local system  $\mathcal{L}$  is the skyscraper sheaf supported on  $\mathfrak{m}_{\eta} = \{0\}$ . Let  $\xi_0 \in \underline{\mathfrak{T}}_{\eta}$  be the  $G_{\underline{0}}$ -conjugacy class of such admissible systems. We call  $\mathcal{D}_{G_{\underline{0}}}(\mathfrak{g}_{\underline{\eta}}^{nil})_{\xi_0}$  the *principal block*. This block contains the intersection cohomology sheaves  $\mathrm{IC}(\mathcal{O})$  of all  $G_{\underline{0}}$ -orbits  $\mathcal{O} \subset \mathfrak{g}_{\eta}^{nil}$  with constant local systems.

1.1.3. Spiral induction. Let  $\epsilon = \eta/|\eta| \in \{1, -1\}$  be the sign of  $\eta$ . In [9, §2] we defined the notion of  $\epsilon$ -spirals for the  $\mathbb{Z}/m\mathbb{Z}$ -graded  $\mathfrak{g}$ . The definitions of spirals and its splittings will be recalled in §3.3.

Fix a  $G_{\underline{0}}$ -conjugacy class  $\xi$  of admissible systems, and let  $(M, M_0, \mathfrak{m}, \mathfrak{m}_*, \mathcal{L}) \in \xi$ . Let  $\mathfrak{P}^{\xi}$  be the set of  $\epsilon$ -spirals  $\mathfrak{p}_*$  of the  $\mathbb{Z}/m\mathbb{Z}$ -graded  $\mathfrak{g}$  such that some (equivalently any) splitting of it is  $G_{\underline{0}}$ -conjugate to  $(M, M_0, \mathfrak{m}, \mathfrak{m}_*)$ . For each  $\mathfrak{p}_* \in \mathfrak{P}^{\xi}$  with splitting  $(L, L_0, \mathfrak{l}, \mathfrak{l}_*)$  which is  $G_{\underline{0}}$ -conjugate to  $(M, M_0, \mathfrak{m}, \mathfrak{m}_*)$ , there is a canonical  $L_0$ equivariant cuspidal local system on  $\mathring{\mathfrak{l}}_\eta$  (the open  $L_0$ -orbit on  $\mathfrak{l}_\eta$ ) which corresponds to  $\mathcal{L}$  under any element  $g \in G_{\underline{0}}$  that conjugates  $(L, L_0, \mathfrak{m}, \mathfrak{m}_*)$  to  $(M, M_0, \mathfrak{m}, \mathfrak{m}_*)$  (see §4.1.3). We denote this local system on  $\mathring{\mathfrak{l}}_\eta$  still by  $\mathcal{L}$ . We have the (unnormalized) spiral induction (see §4.1.3)

$$\mathbf{I}_{\mathfrak{p}_*} = {}^{\epsilon} \mathrm{Ind}_{\mathfrak{p}_{\eta}}^{\mathfrak{g}_{\eta}}(\mathcal{L}^{\sharp}) \in \mathcal{D}_{G_0}(\mathfrak{g}_{\eta}^{nil}).$$

By definition, the full triangulated subcategory  $\mathcal{D}_{G_{\underline{0}}}(\mathfrak{g}_{\underline{\eta}}^{nil})_{\xi} \subset \mathcal{D}_{G_{\underline{0}}}(\mathfrak{g}_{\underline{\eta}}^{nil})$  that appears in the decomposition (1.1) is generated by those simple perverse sheaves that appear as direct summands (up to shifts) of  $\mathbf{I}_{\mathfrak{p}_*}$  for various  $\mathfrak{p}_* \in \mathfrak{P}^{\xi}$ .

The graded dimensions of  $\operatorname{Ext}^*(\mathbf{I}_{\mathfrak{p}_*}, \mathbf{I}_{\mathfrak{p}'_*})$  are computed in [9, Prop. 6.4]. In this paper, we will exhibit a large symmetry on the direct sum of all spiral inductions  $\mathbf{I}_{\mathfrak{p}_*}$  belonging to the fixed block  $\mathcal{D}_{G_{\mathfrak{Q}}}(\mathfrak{g}^{nil}_{\eta})_{\xi}$ .

## 1.2. Main results.

1.2.1. We fix a  $G_{\underline{0}}$ -conjugacy class of admissible systems  $\xi \in \underline{\mathfrak{T}}_{\eta}$  and let  $\dot{\xi} = (M, M_0, \mathfrak{m}, \mathfrak{m}_*, \mathcal{L})$  be an admissible system in  $\xi$ . Attached to  $\xi$  we will introduce an affine Coxeter group  $W_{\mathrm{aff}}^{\xi}$  with simple reflections  $I^{\xi}$  and a reflection representation on a  $\mathbb{Q}$ -vector space  $\mathbb{E}_{\Diamond}^{\xi}$ . We will also introduce a graded double affine Hecke algebra (DAHA)  $\mathbb{H}_c(W_{\mathrm{aff}}^{\xi})$  with parameters  $\{c_i\}_{i\in I^{\xi}}$ . The underlying vector space of  $\mathbb{H}_c(W_{\mathrm{aff}}^{\xi})$  is

$$\mathbb{H}_{c}(W_{\mathrm{aff}}^{\xi}) = \mathbb{Q}[u] \otimes \mathrm{Sym}(\mathbb{E}_{\diamondsuit}^{\xi,*}) \otimes \mathbb{Q}[W_{\mathrm{aff}}^{\xi}].$$

Finally, for  $\nu \in \mathbb{Q}$ , let  $\mathbb{H}_{c,\nu}(W_{\text{aff}}^{\xi})$  be the specialization of  $\mathbb{H}_{c}(W_{\text{aff}}^{\xi})$  by sending  $u \mapsto -\nu$  and  $\mathbb{E}_{\diamond}^{\xi,*} \ni \delta \mapsto 1$ . For more details see §2.5.

The affine Dynkin graphs of  $W_{\text{aff}}^{\xi}$  are exactly those appearing as the " $\flat - \sharp$ diagrams" in the tables of [6, §7] and [8, §11]. The parameters c that appear in the graded DAHA are also specified in loc. cit. as the first label of each node of the  $\flat - \sharp$ -diagrams. From these tables one can list exactly which graded DAHAs appear as our  $\mathbb{H}_{c}(W_{\text{aff}}^{\xi})$ .

1.2.2. Let  $\underline{\mathfrak{P}}^{\xi} = G_{\underline{0}} \setminus \mathfrak{P}^{\xi}$ . It can be shown that  $\mathbf{I}_{\mathfrak{p}_*}$  depends only on the  $G_{\underline{0}}$ -conjugacy class of the  $\epsilon$ -spiral  $\mathfrak{p}_*$  (see §4.1.4). Let  $[\mathfrak{p}_*] \in \underline{\mathfrak{P}}^{\xi}$ ; then  $\mathbf{I}_{[\mathfrak{p}_*]}$  is well-defined. The set  $\underline{\mathfrak{P}}^{\xi}$  admits a combinatorial description in terms of alcoves in an affine space  $\mathbb{E}$  modulo a finite group action (see Corollary 3.4.10). In the notation of [10, §10],  $\underline{\mathfrak{P}}^{\xi}$  is in bijection with the  $\mathcal{W}$ -orbits of alcoves in  $\mathbf{E}$  (alcoves are defined using the hyperplanes  $\{\mathfrak{H}_{\alpha,n,N}\}$  introduced in loc. cit.; for precise statements, see §3.4.11 and Lemma 3.4.12).

The main theorem of this paper is the following.

1.2.3. **Theorem.** Let  $\xi \in \underline{\mathfrak{T}}_{\eta}$ . Consider the infinite direct sum of perverse sheaves in  $\mathcal{D}_{G_{\underline{0}}}(\mathfrak{g}_{\eta}^{nil})_{\xi}$ 

$$\mathbb{I}_{\xi} := \bigoplus_{[\mathfrak{p}_*] \in \underline{\mathfrak{P}}^{\xi}} {}^p \mathrm{H}\mathbf{I}_{[\mathfrak{p}_*]}.$$

The notation  ${}^{p}\mathrm{H}(-)$  means the direct sum of perverse cohomology sheaves; see §1.5.5. Then there is a  $\mathbb{Z}/2\mathbb{Z}$ -graded action of  $\mathbb{H}_{c,n/m}(W_{\mathrm{aff}}^{\xi})$  on  $\mathbb{I}_{\xi}$ .

A  $\mathbb{Z}/2\mathbb{Z}$ -graded action of  $\mathbb{H}_{c,\eta/m}(W_{\mathrm{aff}}^{\xi})$  on the infinite direct sum  $\mathbb{I}_{\xi}$  means the following. For every element  $h \in \mathbb{H}_{c,\eta/m}(W_{\mathrm{aff}}^{\xi})$  and  $[\mathfrak{p}_*], [\mathfrak{p}'_*] \in \mathfrak{P}^{\xi}$ , there is a map of perverse sheaves  $h_{[\mathfrak{p}_*], [\mathfrak{p}'_*]} : {}^{p}\mathrm{HI}_{[\mathfrak{p}_*]} \to {}^{p}\mathrm{HI}_{[\mathfrak{p}'_*]}$  preserving the  $\mathbb{Z}/2\mathbb{Z}$ -gradings such that

- (1) For fixed  $[\mathfrak{p}_*]$ , we have  $h_{[\mathfrak{p}_*],[\mathfrak{p}'_*]} = 0$  for all but finitely many  $[\mathfrak{p}'_*]$ .
- (2) The assignment  $\mathbb{H}_{c,\eta/m}(W_{\mathrm{aff}}^{\xi}) \to \mathrm{Hom}({}^{p}\mathrm{HI}_{[\mathfrak{p}_{*}]}, {}^{p}\mathrm{HI}_{[\mathfrak{p}_{*}']})$  given by  $h \mapsto h_{[\mathfrak{p}_{*}],[\mathfrak{p}_{*}']}$  is  $\mathbb{Q}$ -linear.
- (3) Let  $\mathbf{1} \in \mathbb{H}_{c,\eta/m}(W_{\text{aff}}^{\xi})$  be the identity element; then  $\mathbf{1}_{[\mathfrak{p}_*],[\mathfrak{p}'_*]}$  is the identity if  $[\mathfrak{p}_*] = [\mathfrak{p}'_*]$  and zero otherwise.
- (4) For  $h, h' \in \mathbb{H}_{c,\eta/m}(W_{\mathrm{aff}}^{\xi})$  and  $[\mathfrak{p}_*], [\mathfrak{p}'_*], [\mathfrak{p}''_*] \in \underline{\mathfrak{P}}^{\xi}$ , we have  $h'_{[\mathfrak{p}'_*], [\mathfrak{p}''_*]} \circ h_{[\mathfrak{p}_*], [\mathfrak{p}'_*]}$ =  $(h'h)_{[\mathfrak{p}_*], [\mathfrak{p}''_*]}$ .

1.2.4. By the decomposition theorem, we know that each  ${}^{p}\mathbf{HI}_{[\mathfrak{p}_{*}]}$  is a semisimple perverse sheaf. For any simple perverse sheaf S in  $\mathcal{D}_{G_{\underline{0}}}(\mathfrak{g}_{\underline{\eta}}^{nil})_{\xi}$ , let  $[{}^{p}\mathbf{HI}_{[\mathfrak{p}_{*}]}:S] = \mathrm{Hom}(S, {}^{p}\mathbf{HI}_{[\mathfrak{p}_{*}]})$  be the multiplicity space of S in  ${}^{p}\mathbf{HI}_{[\mathfrak{p}_{*}]}$ . A priori  $[{}^{p}\mathbf{HI}_{[\mathfrak{p}_{*}]}:S]$  is  $\mathbb{Z}/2\mathbb{Z}$ -graded; however, by [10, Theorem 14.10(d)],  $[{}^{p}\mathbf{HI}_{[\mathfrak{p}_{*}]}:S]$  is concentrated in degrees of the same parity as dim  $\mathrm{Supp}(S)$ .

1.2.5. Corollary (of Theorem 1.2.3). For any simple perverse sheaf S in the block  $\mathcal{D}_{G_0}(\mathfrak{g}_{\eta}^{nil})_{\xi}$ , there is an action of  $\mathbb{H}_{c,\eta/m}(W_{\text{aff}}^{\xi})$  on the vector space

(1.2) 
$$[\mathbb{I}_{\xi} : \mathcal{S}] := \bigoplus_{[\mathfrak{p}_{*}] \in \mathfrak{P}^{\xi}} [{}^{p}\mathrm{HI}_{[\mathfrak{p}_{*}]} : \mathcal{S}].$$

In the decomposition (1.2), each direct summand  $[{}^{p}\mathrm{HI}_{[\mathfrak{p}_{*}]}:\mathcal{S}]$  is in fact a generalized eigenspace for the polynomial part of  $\mathbb{H}_{c,\eta/m}(W_{\mathrm{aff}}^{\xi})$ , and the eigenvalue can be described explicitly as a point in  $\mathbb{E}^{\xi}$ ; see Proposition 4.3.4.

We make some conjectures about the  $\mathbb{H}_{c,\eta/m}(W_{\text{aff}}^{\xi})$ -modules we construct.

1.2.6. Conjecture. In the notation of Corollary 1.2.5, the multiplicity space  $[\mathbb{I}_{\xi} : S]$  is a simple  $\mathbb{H}_{c,\eta/m}(W_{\mathrm{aff}}^{\xi}) \otimes_{\mathbb{Q}} \overline{\mathbb{Q}}_{\ell}$ -module.

This conjecture is true when  $\xi = \xi_0$  (corresponding to the principal block), by the result of Vasserot [13].

1.2.7. Fix a homomorphism  $\sigma : \mu_e \hookrightarrow \operatorname{Out}(G)$ , a positive integer m, and a tuple  $\dot{\xi} = (M, M_0, \mathfrak{m}, \mathfrak{m}_*, \mathcal{L})$  where  $(M, M_0, \mathfrak{m}, \mathfrak{m}_*)$  is a graded pseudo-Levi subgroup of  $(G, \sigma)$  (see Definition 2.2.1), and  $\mathcal{L}$  is an  $M_0$ -equivariant cuspidal local system on  $\mathfrak{m}_{\eta}$ . Then the graded DAHA  $\mathbb{H}_{c,\eta/m}(W_{\operatorname{aff}}^{\xi})$  is defined independent of the choice of the  $\mathbb{Z}/m\mathbb{Z}$ -grading on  $\mathfrak{g}$ . An  $\mathbb{H}_{c,\eta/m}(W_{\operatorname{aff}}^{\xi}) \otimes \overline{\mathbb{Q}}_{\ell}$ -module V is called *integrable* if the polynomial part  $\operatorname{Sym}(\mathbb{E}_{\Delta}^{\xi,*}) \otimes \overline{\mathbb{Q}}_{\ell}$  acts on V locally finitely.

For a  $\mathbb{Z}/m\mathbb{Z}$ -grading  $\hat{\theta}$  on  $\mathfrak{g}$  whose outer class is  $\sigma$  (in the sense of §3.1.1), the tuple  $\dot{\xi}$  may or may not be an admissible system for  $(G, \theta)$ . If it is,  $\mathbb{I}_{\xi}$  is defined. To emphasize its dependence on  $\theta$ , we denote it by  $\mathbb{I}_{\xi,\theta}$ . We expect that all integrable simple  $\mathbb{H}_{c,\eta/m}(W_{\mathrm{aff}}^{\xi}) \otimes \overline{\mathbb{Q}}_{\ell}$ -modules are of the form  $[\mathbb{I}_{\xi,\theta} : S]$  for various  $\mathbb{Z}/m\mathbb{Z}$ -gradings  $\theta$  on  $\mathfrak{g}$  such that  $\dot{\xi}$  is admissible with respect to  $\theta$  and various simple perverse sheaves S in  $\mathcal{D}_{G_{\underline{0}}}(\mathfrak{g}_{\underline{\eta}}^{nil})_{\xi}$ .

1.2.8. Mimicking the construction of standard modules for the graded affine Hecke algebra in [5], one can define standard modules for the graded DAHA  $\mathbb{H}_{c,\eta/m}(W_{\text{aff}}^{\xi})$  by taking stalks of  $\mathbb{I}_{\xi}$ . Now if Conjecture 1.2.6 holds, we would get a formula for the multiplicities of the simple modules in the standard modules in terms of the stalks of the simple perverse sheaves in the block  $\mathcal{D}_{G_0}(\mathfrak{g}_{\underline{\eta}}^{nil})_{\xi}$ , which are computable algorithmically as shown in [10].

1.3. Organization of the paper. In §2, the discussion is independent of the  $\mathbb{Z}/m\mathbb{Z}$ -grading on  $\mathfrak{g}$ . We introduce the notion of a pseudo-Levi subgroup M of  $(G, \sigma)$  (where  $\sigma : \mu_e \to \operatorname{Out}(G)$ ) and describe its relation with building theory. For such an M that admits a cuspidal local system, with the extra choice of a facet in the building, we will introduce an affine Coxeter group and a graded DAHA with possibly unequal parameters. In particular, the graded DAHA attached to M is independent of the  $\mathbb{Z}/m\mathbb{Z}$ -grading on  $\mathfrak{g}$ .

In §3, the  $\mathbb{Z}/m\mathbb{Z}$ -grading on  $\mathfrak{g}$  starts to play a role. We review the notion of spirals and describe them again in terms of building theory. In particular, we get a combinatorial description of  $\underline{\mathfrak{P}}^{\xi}$  in Corollary 3.4.10. This section is closely related to [10, §10].

In §4 we prove Theorem 1.2.3. In §4.3 we first construct the action of the polynomial part of  $\mathbb{H}_{c,\eta/m}(W_{\mathrm{aff}}^{\xi})$  on each  $\mathbf{I}_{\mathfrak{p}_*}$  using Chern classes coming from equivariant parameters. Then, in §4.5, for each finite subgroup  $W_J \subset W_{\mathrm{aff}}^{\xi}$  generated by a proper subset J of simple reflections, we construct actions of the corresponding subalgebra  $\mathbb{H}_{c,\eta/m}(W_J)$  on  $\mathbb{I}_{\xi}$  (in fact on a finite direct sum of the  ${}^{p}\mathbf{HI}_{[\mathfrak{p}_*]}$ ). This construction is essentially the construction in [5], which we review in §4.4. Equivariant localization with respect to a torus action is used in the passage from the results in [5] to our setting. Finally the action of  $\mathbb{H}_{c,\eta/m}(W_{\mathrm{aff}}^{\xi})$  is obtained by gluing the actions of  $\mathbb{H}_{c,\eta/m}(W_J)$  for various J (the  $W_J$ 's generate  $W_{\mathrm{aff}}^{\xi}$ ).

## 1.4. Variant and related work.

1.4.1. The  $\mathbb{Z}$ -graded case. The construction in this paper can be applied to the situation of a  $\mathbb{Z}$ -graded Lie algebra  $\mathfrak{g}$ . In this case, for each  $G_0$ -conjugacy class  $\xi = (M, M_0, \mathfrak{m}, \mathfrak{m}, \mathfrak{L})$  of admissible systems, we have a finite Coxeter group  $W^{\xi}$  and a graded affine Hecke algebra  $\mathbb{H}_c(W^{\xi})$  as introduced in [5] for the cuspidal pair  $(M, \mathcal{L})$  for G (independent of the  $\mathbb{Z}$ -grading on  $\mathfrak{g}$ ). There exists an action of  $\mathbb{H}_c(W^{\xi})/(u-1)$  on the direct sum of parabolic inductions from  $(M, M_0, \mathfrak{m}, \mathfrak{m}, \mathfrak{L})$ . This is essentially done in §4.5, and it follows directly from the results in [5] by equivariant localization.

1.4.2. Relation with loop algebras. A natural way to study the  $\mathbb{Z}/m\mathbb{Z}$ -graded Lie algebra  $\mathfrak{g}$  is to turn the  $\mathbb{Z}/m\mathbb{Z}$ -grading into a  $\mathbb{Z}$ -grading on its loop algebra  $L\mathfrak{g} = \mathfrak{g}((z))$  (or its twisted form if  $\sigma$  is non-trivial). Under such a translation, spirals can be interpreted using parahoric subalgebras of  $L\mathfrak{g}$ , and spiral induction can be understood using parahoric induction. We discuss this relationship in §3.4.13, but otherwise we avoid mentioning loop algebras or parahoric induction in this paper in order to make our approach as elementary as possible. However, we are implicitly taking the loop algebra point of view: for example we make a choice of a lifting of the element  $\overline{x} \in \mathbb{X}_*(T^{\mathrm{ad}}) \otimes \mathbb{Z}/m\mathbb{Z}$  giving rise to the  $\mathbb{Z}/m\mathbb{Z}$ -grading to an element  $x \in \mathbb{X}_*(T^{\mathrm{ad}})$  giving rise to a  $\mathbb{Z}$ -grading on  $L\mathfrak{g}$ .

1.4.3. Relation with [13] and [11]. In [13], when the grading on  $\mathfrak{g}$  is inner, a similar action of  $\mathbb{H}_c(W_{\mathrm{aff}})$  for the principal block  $\mathcal{D}_{G_{\underline{0}}}(\mathfrak{g}_{\underline{\eta}}^{nil})_{\xi_0}$  was constructed from the point of view of the loop algebra. Moreover, [13] gives a complete classification of integrable simple  $\mathbb{H}_{c,\eta/m}(W_{\mathrm{aff}})$ -modules: they are in bijection with simple perverse sheaves in the principal block  $\mathcal{D}_{G_{\underline{0}}}(\mathfrak{g}_{\eta}^{nil})_{\xi_0}$ .

In [11], an action of  $\mathbb{H}_{c}(W_{\mathrm{aff}})$  on the cohomology of homogeneous affine Springer fibers was constructed. Spirals did not explicitly appear in these works but the relevant varieties  $X^{A} = G_{\underline{0}} \times \mathfrak{p}_{\eta}^{A}$  (Hessenberg varieties) showed up as torus fixed points on homogeneous affine Springer fibers in the affine flag variety (see [11, §5.4]). The action of  $\mathbb{H}_{c}(W_{\mathrm{aff}})$  on the cohomology of homogeneous affine Springer fibers constructed in [11] is related to the one constructed in this paper via a Fourier transform (between  $\mathcal{D}_{G_{0}}(\mathfrak{g}_{\eta})$  and  $\mathcal{D}_{G_{0}}(\mathfrak{g}_{-\eta})$ ) and equivariant localization.

# 1.5. Notation.

1.5.1. Throughout the paper, k is an algebraically closed field of characteristic zero. All schemes are schemes of finite type over k in this paper unless otherwise claimed.

1.5.2. Let G be an almost simple, simply-connected algebraic group over k. Let  $\mathfrak{g} = \text{Lie } G$ .

1.5.3. Let  $\ell$  be a prime different from char(k). When we talk about complexes of sheaves on a scheme X, we always mean  $\overline{\mathbb{Q}}_{\ell}$ -complexes on X for the étale topology. Let  $\mathcal{D}(X)$  be the bounded derived category of  $\overline{\mathbb{Q}}_{\ell}$ -complexes on X. If an algebraic group H acts on X, let  $\mathcal{D}_H(X)$  be the derived category of H-equivariant  $\overline{\mathbb{Q}}_{\ell}$ -complexes on X as developed in [1]. We will use the notion of complexes of sheaves on algebraic stacks  $\mathfrak{X}$  but only for quotient stacks of the form  $\mathfrak{X} = [X/H]$ , in which case  $\mathcal{D}(\mathfrak{X}) = \mathcal{D}_H(X)$  by definition. All sheaf-theoretic functors are derived.

1.5.4. Let X be a scheme. For a local system  $\mathcal{L}$  on some locally closed, equidimensional smooth subscheme  $j: Y \hookrightarrow X$ , we let  $\mathcal{L}^{\sharp} = j_{!*}(\mathcal{L}[\dim Y])[-\dim Y] \in \mathcal{D}(X)$ . Note that  $\mathcal{L}^{\sharp}|_{Y} = \mathcal{L}$ .

1.5.5. For a scheme X of finite type over k with an action of an algebraic group H, and any object  $\mathcal{K} \in \mathcal{D}_H(X)$ , we define the perverse sheaves

$${}^{p}\mathrm{H}^{\mathrm{ev}}\mathcal{K} := \bigoplus_{n \text{ even}} {}^{p}\mathrm{H}^{n}\mathcal{K}; \quad {}^{p}\mathrm{H}^{\mathrm{odd}}\mathcal{K} := \bigoplus_{n \text{ odd}} {}^{p}\mathrm{H}^{n}\mathcal{K};$$
$${}^{p}\mathrm{H}\mathcal{K} := {}^{p}\mathrm{H}^{\mathrm{ev}}\mathcal{K} \oplus {}^{p}\mathrm{H}^{\mathrm{odd}}\mathcal{K} = \bigoplus_{n \in \mathbb{Z}} {}^{p}\mathrm{H}^{n}\mathcal{K}.$$

We consider  ${}^{p}H\mathcal{K}$  as a  $\mathbb{Z}/2\mathbb{Z}$ -graded perverse sheaf.

1.5.6. For a subalgebra  $\mathfrak{h} \subset \mathfrak{g}$ , we denote by  $e^{\mathfrak{h}}$  the (smooth) connected subgroup of G with Lie algebra  $\mathfrak{h}$ , whenever it exists.

1.5.7. For an algebraic group H and a subgroup H' of it which acts on a scheme X, we denote by  $H \stackrel{H'}{\times} X$  the (stack) quotient of  $H \times X$  under the H'-action given by  $h' \cdot (h, x) = (hh'^{-1}, h'x)$  for  $h' \in H', h \in H$  and  $x \in X$ . In the examples we consider, the quotient  $H \stackrel{H'}{\times} X$  is always representable by a scheme.

1.5.8. For a  $\mathbb{Z}$ -graded Lie algebra  $\mathfrak{h} = \bigoplus_{n \in \mathbb{Z}} \mathfrak{h}_n$ , we often abbreviate the collection of the graded pieces  $\{\mathfrak{h}_n | n \in \mathbb{Z}\}$  by  $\mathfrak{h}_*$ .

1.5.9. For a positive integer n,  $\mu_n$  denotes the diagonalizable group over k of nth roots of unity.

#### 2. Relative affine Weyl group and graded DAHA

In this section, for a pseudo-Levi subgroup M of  $(G, \sigma)$  which admits a cuspidal local system supported on a nilpotent orbit, we will introduce an affine Coxeter group and a graded DAHA with possibly unequal parameters. The results in this section are essentially contained in [6] and [5].

#### 2.1. Affine root system.

2.1.1. We fix a pinning  $E = (B_0, T_0, \cdots)$  of G, where  $B_0$  is a Borel subgroup and  $T_0$  a maximal torus of G contained in  $B_0$ . We identify Out(G) with the pinned automorphism group of G. Fix an injective homomorphism

(2.1) 
$$\sigma: \mu_e \hookrightarrow \operatorname{Out}(G).$$

Then  $e \in \{1, 2, 3\}$ . Let  $\mu_e$  act on G via pinned automorphisms through  $\sigma$ . Let  $G^{\sigma}$  be the fixed point subgroup of  $\sigma(\mu_e)$ .

2.1.2. Let  $T = T_0^{\sigma}$  be the fixed points of  $\sigma(\mu_e)$ ; then T is a maximal torus of  $G^{\sigma}$ , and  $T_0 = C_G(T)$ . Let  $\mathfrak{t} = \text{Lie } T$ . Let  $\Phi(G, T) \subset \mathbb{X}_*(T)$  be the root system of G with respect to T. Note that  $\Phi(G, T)$  is not necessarily reduced, and it is non-reduced precisely when  $G = \text{SL}_{2n+1}$  and e = 2.

2.1.3. Let  $W_G$  be the Weyl group of G with respect to  $T_0$ . Let W be the Weyl group of the root system  $\Phi(G,T)$ . Then W is also the Weyl group of  $G^{\sigma}$  with respect to T. We have  $W = W_G^{\sigma}$ , the fixed point subgroup of  $\sigma(\mu_e)$ .

2.1.4. The pinned action of  $\sigma(\mu_e)$  gives a decomposition  $\mathfrak{g} = \bigoplus_{i \in \mathbb{Z}/e\mathbb{Z}} \mathfrak{g}^i$ . For each  $\alpha \in \Phi(G,T)$  and  $i \in \mathbb{Z}/e\mathbb{Z}$ , the root space  $\mathfrak{g}^i(\alpha)$  is either zero or one dimensional. Let

$$\Phi(G,T) = \{ (\alpha,i) \in \Phi(G,T) \times \mathbb{Z}/e\mathbb{Z} | \mathfrak{g}^i(\alpha) \neq 0 \}.$$

2.1.5. Let  $\mathfrak{a} = \mathbb{X}_*(T) \otimes_{\mathbb{Z}} \mathbb{Q}$ . The Killing form on  $\mathfrak{g}$  gives  $\mathfrak{a}_{\mathbb{R}} = \mathfrak{a} \otimes \mathbb{R}$  a Euclidean structure. Let  $\Phi_{\mathrm{aff}}$  be the set of affine functions  $\widetilde{\alpha} = \alpha + \frac{n}{e}$  on  $\mathfrak{a}$ , for  $(\alpha, n) \in \Phi(G, T) \times \mathbb{Z}$  such that  $(\alpha, n \mod e) \in \widetilde{\Phi}(G, T)$ . This is the set of real affine roots attached to the pair  $(G, \sigma)$ . The vanishing locus of each element  $\widetilde{\alpha} \in \Phi_{\mathrm{aff}}$  gives an affine hyperplane  $H_{\widetilde{\alpha}} \subset \mathfrak{a}$ . Let  $\mathfrak{H}$  be the collection of such affine hyperplanes. These hyperplanes give a stratification of  $\mathfrak{a}$  into facets. Let  $\mathfrak{F}$  be the set of facets in  $\mathfrak{a}$ .

2.1.6. **Definition.** A relevant affine subspace  $\mathbb{E}$  of  $\mathfrak{a}$  is the affine subspace spanned by some facet F. Let  $\mathfrak{E}$  be the set of relevant affine subspaces of  $\mathfrak{a}$ .

2.1.7. The affine Weyl group  $W_{\text{aff}}$  attached to  $(G, \sigma)$  is the group of affine isometries of  $\mathfrak{a}$  generated by the orthogonal reflections across affine hyperplanes  $H \in \mathfrak{H}$ . For any facet  $F \subset \mathfrak{a}$ , let  $W_F \subset W_{\text{aff}}$  be the stabilizer of the facet F under  $W_{\text{aff}}$ . Then  $W_F$  fixes F pointwise, and it is a finite Weyl group generated by the reflections across those hyperplanes  $H \in \mathfrak{H}$  that contain F. Note that  $W_F$  only depends on the affine subspace spanned by F. Therefore, if  $\mathbb{E} \in \mathfrak{E}$ , we may define  $W_{\mathbb{E}}$  to be  $W_F$  for any facet F that spans  $\mathbb{E}$ .

2.1.8. Let  $\mathfrak{a}_{\diamond} = \mathfrak{a} \oplus \mathbb{Q}d$  with dual space  $\mathfrak{a}^{\diamond,*} = \mathbb{Q}\delta \oplus \mathfrak{a}^*$ , such that  $\langle \delta, d \rangle = 1$ . We think of  $\delta$  as the generator of the imaginary roots in  $\Phi_{\mathrm{aff}}$ . The affine action of  $W_{\mathrm{aff}}$  on  $\mathfrak{a}$  extends canonically to a linear action of  $W_{\mathrm{aff}}$  on  $\mathfrak{a}_{\diamond}$ :  ${}^w(v+zd) := w(zv) + zd$ , where  $v \in \mathfrak{a}, z \in \mathbb{Q}, w \in W_{\mathrm{aff}}$  and w(-) denotes the action of w on  $\mathfrak{a}$ .

2.1.9. Let  $T^{\mathrm{ad}} = T/Z_G^{\sigma}$ , where  $Z_G^{\sigma} = Z_G \cap G^{\sigma}$  is also the center of  $G^{\sigma}$ . The embedding  $\mathbb{X}_*(T^{\mathrm{ad}}) \subset \mathfrak{a}$  gives a lattice in  $\mathfrak{a}$ . We have

$$\mathbb{X}_*(T^{\mathrm{ad}}) = \{\lambda \in \mathfrak{a} | \langle \alpha, \lambda \rangle \in \mathbb{Z}, \forall \alpha \in \Phi(G, T) \}$$

The extended affine Weyl group  $\widetilde{W}$  is the group of affine isometries of  $\mathfrak{a}$  that stabilize the stratification set  $\mathfrak{H}$  of affine hyperplanes. The translation part of  $\widetilde{W}$  is  $\mathbb{X}_*(T^{\mathrm{ad}})$ and  $\widetilde{W}/\mathbb{X}_*(T^{\mathrm{ad}}) \cong W$ .

2.2. **Pseudo-Levi subgroups.** Let  $\operatorname{Aut}_{\sigma}(G) \subset \operatorname{Aut}(G)$  be the preimage of  $\sigma(\mu_e)$ under the projection  $\pi_{\operatorname{Out}}$ :  $\operatorname{Aut}(G) \to \operatorname{Out}(G)$ . Let  $\mu_e^* \subset \mu_e$  be the primitive eth roots of unity (so  $\mu_1^* = \mu_1 = \{1\}$ ). Let  $\operatorname{Aut}_{\sigma}^*(G)$  be the preimage of  $\sigma(\mu_e^*)$  under  $\pi_{\operatorname{Out}}$ .

2.2.1. **Definition.** A pseudo-Levi subgroup of  $(G, \sigma)$  is the fixed point subgroup of G of the form  $M = G^{\tau}$  for some semisimple automorphism  $\tau \in \operatorname{Aut}^*_{\sigma}(G)$ .

In the case e = 3, we may restrict  $\tau$  to be in a fixed non-neutral component of  $\operatorname{Aut}_{\sigma}^*(G)$  (among the two choices), and they give the same notion of pseudo-Levi subgroups, because inversion on  $\operatorname{Aut}(G)$  interchanges the two choices. By Steinberg's theorem, a pseudo-Levi subgroup M is a connected reductive group.

## 2.2.2. Lemma.

- (1) Any semisimple element  $\tau \in \operatorname{Aut}^*_{\sigma}(G) = G^{\operatorname{ad}} \rtimes \sigma(\mu_e^*)$  may be conjugated to  $T^{\operatorname{ad}} \times \sigma(\mu_e^*)$  under G.
- (2) If  $M = G^{\tau}$  for some  $\tau \in T^{\mathrm{ad}} \times \sigma(\mu_e^*)$ , then T is a maximal torus of M.

Proof. (1) Recall the fixed pinning  $E = (B_0, T_0, \cdots)$  of G and  $T = T_0^{\sigma}$ . By [12, 7.5, 7.6], there exists a Borel subgroup  $B_1$  of G and a maximal torus  $T_1$  of G with  $T_1 \subset B_1$  such that  $\tau$  normalizes  $B_1$  and  $T_1$ . Let  $g \in G^{\mathrm{ad}}$  be such that  $\mathrm{Ad}(g)(B_1, T_1) = (B_0, T_0)$ ; then  $g\tau g^{-1}$  normalizes  $(B_0, T_0)$ . Replacing  $\tau$  by  $g\tau g^{-1}$ , we may assume that  $\tau$  normalizes  $(B_0, T_0)$ . Then there is an element  $s \in T_0^{\mathrm{ad}}$  such that  $s^{-1}\tau$  fixed the pinning E, i.e.,  $s^{-1}\tau = \sigma(\zeta)$ , where  $\zeta \in \mu_e^*$  such that  $\sigma(\zeta)$  is the image of  $\tau$  in  $\mathrm{Out}(G)$ . We then have  $\tau = s\sigma(\zeta)$  for some  $s \in T_0^{\mathrm{ad}}$ . Let  $T_{0,\sigma(\zeta)}^{\mathrm{ad}}$  be the coinvariants of the  $\sigma(\zeta)$ -action on  $T_0^{\mathrm{ad}}$ ; then the natural map  $T^{\mathrm{ad}} = T_0^{\mathrm{ad},\sigma(\zeta)} \hookrightarrow T_0^{\mathrm{ad}} \twoheadrightarrow T_{0,\sigma(\zeta)}^{\mathrm{ad}}$  is surjective. Let  $s' \in T^{\mathrm{ad}}$  have the same image as s in  $T_{0,\sigma(\zeta)}^{\mathrm{ad}}$ ; then there exists  $s_1 \in T_0^{\mathrm{ad}}$  such that  $s's^{-1} = s_1 \cdot \sigma(\zeta)s_1^{-1}$  (we denote the action of  $\sigma(\zeta)$  on  $G^{\mathrm{ad}}$  by  $g \mapsto \sigma(\zeta)g$ ). Then we have  $s_1\tau s_1^{-1} = s_1s\sigma(\zeta)s_1^{-1} = s_1 \cdot \sigma(\zeta)s_1^{-1} \cdot s\sigma(\zeta) = s'\sigma(\zeta) \in T^{\mathrm{ad}} \times \sigma(\mu_e^*)$ , as desired.

(2) We have  $C_M(T) = C_G(T) \cap M = T_0 \cap M = T_0^{\tau}$ . Since  $\tau = s\sigma(\zeta)$  for some  $s \in T^{\text{ad}}$  and  $\zeta \in \mu_e^*$ , we have  $T_0^{\tau} = T_0^{\sigma(\zeta)} = T_0^{\sigma} = T$ , therefore  $C_M(T) = T$  and T is a maximal torus in M.

2.2.3. Let  $\mathfrak{M}$  be the set of pseudo-Levi subgroups of  $(G, \sigma)$ . Let  $\mathfrak{M} = G \setminus \mathfrak{M}$  be the G-conjugacy classes of pseudo-Levi subgroups. Let  $\mathfrak{M}_T$  be the set of pseudo-Levi subgroups of  $(G, \sigma)$  of the form  $G^{\tau}$  for some  $\tau \in T^{\mathrm{ad}} \times \sigma(\mu_e^*)$ . Then  $\mathfrak{M}_T \to \mathfrak{M}$  is surjective by Lemma 2.2.2(1). The Weyl group W acts on  $\mathfrak{M}_T$  by conjugation.

2.2.4. Root system of a pseudo-Levi. When  $M \in \mathfrak{M}_T$ , we have the root system  $\Phi(M,T)$ . For each  $\alpha \in \Phi(M,T)$ , the one-dimensional root space  $\mathfrak{m}(\alpha)$  lies in some  $\mathfrak{g}^i$  for a unique  $i \in \mathbb{Z}/e\mathbb{Z}$  (for if M is the centralizer of  $(t, \sigma(\zeta_e)) \in T^{\mathrm{ad}} \times \sigma(\mu_e)$ , then  $\mathfrak{m}(\alpha) \neq 0$  if and only if  $\alpha(t)\sigma(\zeta_e)$  acts on  $\mathfrak{g}$  with eigenvalue 1, in

which case there is a unique  $i \in \mathbb{Z}/e\mathbb{Z}$  such that  $\alpha(t) \in \zeta_e^{-i}$  and  $\mathfrak{g}^i(\alpha) = \mathfrak{m}(\alpha) \neq 0$ ). Therefore each  $\alpha \in \Phi(M,T)$  determines some  $i \in \mathbb{Z}/e\mathbb{Z}$ . In other words the inclusion  $\Phi(M,T) \subset \Phi(G,T)$  can be canonically lifted to an injective map  $\Phi(M,T) \hookrightarrow \widetilde{\Phi}(G,T) \subset \Phi(G,T) \times \mathbb{Z}/e\mathbb{Z}$ , and we denote its image by  $\widetilde{\Phi}(M,T) \subset \widetilde{\Phi}(G,T)$ .

2.2.5. Relevant affine subspaces and pseudo-Levi subgroups. Let  $\mathbb{E} \in \mathfrak{E}$  be a relevant affine subspace of  $\mathfrak{a}$ . Let  $R_{\mathbb{E}} \subset \widetilde{\Phi}(G,T)$  be the set of pairs  $(\alpha,i)$  such that  $\langle \alpha, y \rangle + i/e \in \mathbb{Z}$  for all  $y \in \mathbb{E}$ . If  $(\alpha,i) \in R_{\mathbb{E}}$ , then  $\alpha$  determined  $i \in \mathbb{Z}/e\mathbb{Z}$ , hence the projection  $R_{\mathbb{E}} \to \Phi(G,T)$  is injective. In particular,  $R_{\mathbb{E}}$  is a reduced root system. There is a unique reductive subgroup  $G_{\mathbb{E}} \subset G$  containing T as a maximal torus with Lie algebra

$$\mathfrak{g}_{\mathbb{E}} = \mathfrak{t} \oplus (\bigoplus_{(\alpha,i)\in R_{\mathbb{E}}} \mathfrak{g}^i(\alpha)).$$

In other words  $\Phi(G_{\mathbb{E}}, T) = R_{\mathbb{E}}$ .

We claim that  $G_{\mathbb{E}} \in \mathfrak{M}_T$ . In fact, we pick any  $y \in \mathbb{E}$  that does not lie in any facet with dimension strictly smaller than that of  $\mathbb{E}$ , and write  $y = \lambda/n$  for some  $\lambda \in \mathbb{X}_*(T)$  and  $n \in e\mathbb{Z}_{>0}$ . Then  $R_{\mathbb{E}} = \{(\alpha, i) \in \widetilde{\Phi}(G, T) | \langle \alpha, y \rangle + i/e \in \mathbb{Z}\}$ . Let  $\tau = \lambda(\zeta_n) \cdot \sigma(\zeta_n^{n/e}) \in T \times \sigma(\mu_e^*)$  for some primitive *n*th root of unity  $\zeta_n$ . Then one can check by examining root spaces that  $G_{\mathbb{E}} = G^{\tau}$ , and therefore  $G_{\mathbb{E}} \in \mathfrak{M}_T$ . This way we have defined a map

(2.2) 
$$\Gamma: \mathfrak{E} \to \mathfrak{M}_T$$

given by  $\mathbb{E} \mapsto G_{\mathbb{E}}$ . Since all elements  $\tau \in T \times \sigma(\mu_e^*)$  can be obtained from an element  $y = \lambda/n \in \mathfrak{a}$  as above,  $\Gamma$  is surjective. It is also easy to see that  $\Gamma$  is  $\mathbb{X}_*(T^{\mathrm{ad}})$ -invariant with respect to its translation action on  $\mathfrak{E}$ .

2.2.6. Suppose  $y \in \mathfrak{a}$ ; then y lies in a unique facet F which spans a relevant affine subspace  $\mathbb{E}$ . We define  $G_y$  and  $G_F$  to be  $G_{\mathbb{E}}$ . We denote the Lie algebra of  $G_y$  and  $G_F$  by  $\mathfrak{g}_y$  and  $\mathfrak{g}_F$ . Consider the composition

$$\underline{\Gamma}_{\mathfrak{F}}:\mathfrak{F}\to\mathfrak{E}\xrightarrow{\Gamma}\mathfrak{M}_T\to\mathfrak{M}_T$$

sending a facet F to  $G_{\mathbb{E}}$  where  $\mathbb{E} = \text{Span}(F)$ . For  $[M] \in \mathfrak{M}$  (the *G*-conjugacy class of a pseudo-Levi M), let  $\mathfrak{F}^{[M]} \subset \mathfrak{F}$  denote the preimage of [M] under the map  $\underline{\Gamma}_{\mathfrak{F}}$ .

2.2.7. Let  $\mathbb{E}$  be a relevant affine subspace of  $\mathfrak{a}$ , and let  $M = G_{\mathbb{E}}$  be the corresponding pseudo-Levi subgroup. Let  $Z_M$  be the center of M, and let  $Z_M^0$  be the neutral component of  $Z_M$ . Let  $\mathfrak{a}^M = \mathbb{X}_*(Z_M^0) \otimes_{\mathbb{Z}} \mathbb{Q} \subset \mathfrak{a}$ . Then  $\mathbb{E}$  is a torsor under  $\mathfrak{a}^M$ .

Let  $\mathfrak{a}_M = \mathbb{X}_*(T/Z_M) \otimes_{\mathbb{Z}} \mathbb{Q} = \mathfrak{a}/\mathfrak{a}^M$ . Let  $\pi_M : \mathfrak{a} \to \mathfrak{a}_M$  be the natural projection. The subgroup  $W_{\mathbb{E}}$  of  $W_{\text{aff}}$  fixes  $\mathbb{E}$  pointwise and induces an affine action on  $\mathfrak{a}_M$  fixing the point  $\pi_M(\mathbb{E})$ . Shifting by  $-\pi_M(\mathbb{E})$ , we may identify  $W_{\mathbb{E}}$  with the Weyl group of M with respect to T, which acts linearly on  $\mathfrak{a}_M$ .

2.2.8. The Killing form on  $\mathfrak{a}$  restricts to a Euclidean structure on  $\mathbb{E}_{\mathbb{R}} = \mathbb{E} \otimes \mathbb{R}$ . For each  $H \in \mathfrak{H}$ , the intersection  $H \cap \mathbb{E}$  is either empty, or the whole  $\mathbb{E}$ , or a hyperplane in  $\mathbb{E}$ . Let  $\mathfrak{H}(\mathbb{E})$  be the set of hyperplanes in  $\mathbb{E}$  of the form  $H \cap \mathbb{E}$ , where  $H \in \mathfrak{H}$ . These hyperplanes give a stratification of  $\mathbb{E}$  into  $\mathbb{E}$ -facets. This stratification is simply the restriction to  $\mathbb{E}$  of the stratification of  $\mathfrak{a}$  by facets, and an  $\mathbb{E}$ -facet is simply a facet of  $\mathfrak{a}$  that is contained in  $\mathbb{E}$ . Let  $\mathfrak{F}(\mathbb{E})$  denote the set of  $\mathbb{E}$ -facets.

We call the open  $\mathbb{E}$ -facets  $\mathbb{E}$ -alcoves. These are in natural bijection with connected components of  $\mathbb{E}'_{\mathbb{R}} = \mathbb{E}_{\mathbb{R}} - \bigcup_{H \in \mathfrak{H}(\mathbb{E})} H_{\mathbb{R}}$ . The set of  $\mathbb{E}$ -alcoves is denoted  $\operatorname{Alc}(\mathbb{E})$ .

# 2.3. Action of $\mathbb{G}_m^{\mathbf{rot}}$ .

2.3.1. Let  $\mathbb{G}_m^{\text{rot}}$  be a one-dimensional torus. For each relevant affine subspace  $\mathbb{E} \subset \mathfrak{a}$ , we define an action of  $\mathbb{G}_m^{\text{rot}}$  on the corresponding pseudo-Levi subgroup  $G_{\mathbb{E}}$  as follows. It acts trivially on T. For each  $(\alpha, i) \in R_{\mathbb{E}}$ , we let  $\mathbb{G}_m^{\text{rot}}$  act on the root space  $\mathfrak{g}^i(\alpha)$  of  $\mathfrak{g}_{\mathbb{E}}$  with weight  $-e\langle \alpha, y \rangle$  for any  $y \in \mathbb{E}$  (note here that  $-e\langle \alpha, y \rangle \in \mathbb{Z}$  is independent of the choice of  $y \in \mathbb{E}$ ).

If H is a subgroup of  $G_{\mathbb{E}}$  stable under the action of  $\mathbb{G}_m^{\text{rot}}$ , we denote  $H_{\diamondsuit} = H \rtimes \mathbb{G}_m^{\text{rot}}$ .

2.3.2. We denote the canonical generator of  $\mathbb{X}_*(\mathbb{G}_m^{\mathrm{rot}})$  by ed; dually the canonical generator of  $\mathbb{X}^*(\mathbb{G}_m^{\mathrm{rot}})$  is  $\delta/e$ . We then identify  $\mathfrak{a}_{\diamond} = \mathfrak{a} \oplus \mathbb{Q}d$  (see §2.1.8) with  $\mathbb{X}_*(T_{\diamond})_{\mathbb{Q}}$ , where  $T_{\diamond} = T \times \mathbb{G}_m^{\mathrm{rot}}$ .

2.3.3. For a relevant affine subspace  $\mathbb{E} \subset \mathfrak{a}$ , let  $\mathbb{E}_{\diamond}$  be the  $\mathbb{Q}$ -linear span of the affine subspace  $\mathbb{E} + d \subset \mathfrak{a}_{\diamond}$ . Let  $M = G_{\mathbb{E}}$ . Then  $\delta \in \mathfrak{a}_{\diamond}^*$  restricts to a linear function  $\delta : \mathbb{E}_{\diamond} \to \mathbb{Q}$ , and we have

$$\mathbb{E} = \delta^{-1}(1); \quad \mathfrak{a}^M = \delta^{-1}(0).$$

The center  $Z_{M_{\diamond}}$  of  $M_{\diamond} = M \rtimes \mathbb{G}_m^{\mathrm{rot}}$  fits into an exact sequence

$$1 \to Z_M \to Z_{M_{\diamond}} \to \mathbb{G}_m^{\mathrm{rot}} \to 1.$$

The neutral component  $Z_{M_{\Diamond}}^{0}$  is a subtorus of  $T_{\Diamond} = T \times \mathbb{G}_{m}^{\text{rot}}$ . Therefore  $\mathbb{X}_{*}(Z_{M_{\Diamond}}^{0})_{\mathbb{Q}}$  is a linear subspace of  $\mathbb{X}_{*}(T_{\Diamond})_{\mathbb{Q}} = \mathfrak{a}_{\Diamond}$ . The following lemma is a direct calculation.

2.3.4. Lemma. We have  $\mathbb{X}_*(Z^0_{M_{\triangle}})_{\mathbb{Q}} = \mathbb{E}_{\Diamond}$  as subspaces of  $\mathfrak{a}_{\Diamond}$ .

## 2.4. Relative affine Weyl group.

2.4.1. We fix a *G*-conjugacy class [M] of pseudo-Levi subgroups of  $(G, \sigma)$ , i.e.,  $[M] \in \underline{\mathfrak{M}}$ . We assume that  $\mathfrak{m} = \operatorname{Lie} M$  carries an *M*-equivariant cuspidal local system  $\mathcal{L}$  supported on some nilpotent orbit  $\mathcal{O}$ . This means that the perverse sheaf  $\mathcal{L}^{\sharp}[\dim \mathcal{O}]$  is cuspidal for the semisimple Lie algebra  $\mathfrak{m}^{\operatorname{der}} = [\mathfrak{m}, \mathfrak{m}]$  in the sense of [4, 2(a)]. By the classification of cuspidal local systems in [3], such a nilpotent orbit  $\mathcal{O}$ , if it exists, is unique (i.e., independent of the cuspidal local system  $\mathcal{L}$ ).

The possible conjugacy classes of pseudo-Levi subgroups M supporting cuspidal local systems on a nilpotent orbit are classified in [6, §7] in the case e = 1 and in [8, §11] in the case e = 2 or 3. If we choose a standard alcove  $A_0$  in  $\mathfrak{a}$ , one can find a standard facet  $F \subset \overline{A}_0$  such that  $F \in \mathfrak{F}^{[M]}$  (which may not be unique). Then Fcorresponds to a subdiagram of the (twisted) extended Dynkin diagram of  $(G, \sigma)$ . In loc. cit., all such subdiagrams arising from M supporting cuspidal local systems are shown in the boxes inside the ambient affine Dynkin diagrams.

2.4.2. Let  $A \in \mathfrak{F}^{[M]}$  and  $\mathbb{E} = \text{Span}(A)$ . Recall this means that  $G_{\mathbb{E}}$  is in the *G*-conjugacy class [M]. For notational convenience we may assume  $G_{\mathbb{E}} = M$ .

For each  $H \in \mathfrak{H}(\mathbb{E})$ , there is a unique orthogonal reflection  $r_H$  across H. Let  $W_{\text{aff}}(\mathbb{E})$  be the group of affine isometries of  $\mathbb{E}$  generated by the reflections  $r_H$  for all  $H \in \mathfrak{H}(\mathbb{E})$ . Let I(A) be the set of hyperplanes  $H \in \mathfrak{H}(\mathbb{E})$  spanned by codimension one facets on the boundary of A.

## 2.4.3. **Proposition** ([6]).

- (1) The group  $W_{\text{aff}}(\mathbb{E})$  acts simply transitively on Alc( $\mathbb{E}$ ).
- (2) If dim  $\mathbb{E} > 0$ , then  $W_{\text{aff}}(\mathbb{E})$  is an affine Coxeter group with simple reflections  $\{r_H\}_{H \in I(A)}$ .

*Proof.* Let I be any system of simple generators of  $W_{\text{aff}}$  containing I(A). By [6, §5.6] (which uses the classification of cuspidal local systems given in [5]), in the case e = 1, I(A) is an *excellent* subset of I in the sense of [6, §2.4]. In the case e = 2 or 3, I(A) is again an excellent subset of I. This follows by examining the tables in [8, §11] case by case. Then the desired statements follow from [6, §2.9, §2.11(a)].

2.4.4. For  $A \in \mathfrak{F}^{[M]}$  which spans  $\mathbb{E}$ , let

$$W^A_{\text{aff}} := (W_{\text{aff}}(\mathbb{E}), I(A))$$

be the resulting Coxeter group.

For each  $H \in I(A)$ , we have its stabilizer  $W_H$  under  $W_{\text{aff}}$ . For a pair  $H \neq H' \in I(A)$ , the intersection  $H \cap H'$  is also a relevant affine subspace, hence  $W_{H \cap H'} \subset W_{\text{aff}}$  is also defined. For any of the finite Weyl groups  $W_{\mathbb{E}}, W_H$ , and  $W_{H \cap H'}$ , under any system of Coxeter generators, the length of the longest element is well-defined, and we denote it by  $\ell(W_{\mathbb{E}}), \ell(W_H)$  and  $\ell(W_{H \cap H'})$ .

By [6, §2.28(a)], for  $H, H' \in I(A)$ , the order  $m_{H,H'}$  of  $r_H r_{H'}$  in  $W_{\text{aff}}(\mathbb{E})$  is given by

$$m_{H,H'} = \frac{2(\ell(W_{H\cap H'}) - \ell(W_{\mathbb{E}}))}{\ell(W_H) + \ell(W_{H'}) - 2\ell(W_{\mathbb{E}})}.$$

Let  $N_{W_{\text{aff}}}(W_{\mathbb{E}})$  be the normalizer of  $W_{\mathbb{E}}$  in  $W_{\text{aff}}$ . Since  $W_{\mathbb{E}}$  is the pointwise stabilizer of  $\mathbb{E}$  and  $\mathbb{E}$  is the fixed point locus of  $W_{\mathbb{E}}$ ,  $N_{W_{\text{aff}}}(W_{\mathbb{E}})$  is also the stabilizer of  $\mathbb{E}$  under  $W_{\text{aff}}$ . The quotient  $N_{W_{\text{aff}}}(W_{\mathbb{E}})/W_{\mathbb{E}}$  acts on  $\mathbb{E}$  by affine isometries.

2.4.5. Proposition ([6, Lemma 2.29(a)]). The natural homomorphism

 $N_{W_{\mathrm{aff}}}(W_{\mathbb{E}})/W_{\mathbb{E}} \to \mathrm{Aut}(\mathbb{E})$ 

is injective, and its image is  $W_{\text{aff}}(\mathbb{E})$ . In particular, we have a canonical isomorphism

$$N_{W_{\mathrm{aff}}}(W_{\mathbb{E}})/W_{\mathbb{E}} \cong W_{\mathrm{aff}}(\mathbb{E}).$$

2.4.6. The linear action of  $W_{\text{aff}}$  on  $\mathfrak{a}_{\diamond}$  (see §2.1.8) restricts to a linear action of  $W_{\text{aff}}(\mathbb{E}) \cong N_{W_{\text{aff}}}(W_{\mathbb{E}})/W_{\mathbb{E}}$  on  $\mathbb{E}_{\diamond}$ . We denote the action of  $w \in W_{\text{aff}}(\mathbb{E})$  on  $\mathbb{E}_{\diamond}$  and its dual space  $\mathbb{E}^*_{\diamond}$  by  $\xi \mapsto {}^w \xi$ .

2.4.7. For each hyperplane  $H \in \mathfrak{H}(\mathbb{E})$ , we shall introduce a root  $\alpha_H \in \mathfrak{a}^{M,*}$  and a coroot  $\alpha_H^{\vee} \in \mathfrak{a}^M$ . Since H itself is a relevant affine subspace, we have a pseudo-Levi subgroup  $L := G_H \subset G$  containing M. The connected component of  $\mathbb{E}_{\mathbb{R}} - H_{\mathbb{R}}$  containing A gives a parabolic subgroup  $Q \subset L$  containing M as a Levi subgroup. This gives a triangular decomposition of  $\mathfrak{l} = \text{Lie } L$ ,

$$(2.3) l = n^+ \oplus m \oplus n^-,$$

where  $\mathfrak{n}^+$  is the nilradical of Lie Q. Then  $\mathfrak{m}$  acts on  $\mathfrak{n}^{\pm}$  by the adjoint action.

The  $Z_M^0$ -weights on  $\mathfrak{n}^+$  are multiples of each other (because their vanishing loci are parallel to H). Let  $\alpha_H \in \mathbb{X}^*(Z_M^0)$  be the shortest weight that appears in  $\mathfrak{n}^+$ . This defines an assignment  $\mathfrak{H}(\mathbb{E}) \to \mathbb{X}^*(Z_M^0)$  given by  $H \mapsto \alpha_H$ . The vanishing locus of  $\alpha_H$  (as a hyperplane in  $\mathfrak{a}^M$ ) is parallel to H. There is a unique affine function  $\widetilde{\alpha}_H$  on  $\mathbb{E}$  whose linear part is  $\alpha_H$  and whose vanishing locus is H.

Recall we have the reflection  $r_H$  across H. We define  $\alpha_H^{\vee} \in \mathfrak{a}^M$  to be the unique element such that

$$r_H(x) = x - \widetilde{\alpha}_H(x)\alpha_H^{\vee} \quad \forall x \in \mathbb{E}.$$

2.4.8. For each  $H \in \mathfrak{H}(\mathbb{E})$ , we introduce a positive integer  $c_H$  following [5, 2.10]. We use the same notation as §2.4.7. Recall that  $\mathcal{O} \subset \mathfrak{m}$  is the nilpotent orbit that supports a cuspidal local system. Pick any  $e \in \mathcal{O}$ , and denote its adjoint action on  $\mathfrak{n}^+$  (see (2.3)) by ad(e). We define  $c_H$  to be the largest integer c such that  $\mathrm{ad}(e)^{c-2} \neq 0$  as an endomorphism of  $\mathfrak{n}^+$ . In other words,  $c_H - 1$  is the largest size of a Jordan block of  $\mathfrak{n}^+$  under ad(e). Clearly  $c_H$  is independent of the choice of  $e \in \mathcal{O}$ . The assignment  $H \mapsto c_H$  defines a function

$$c:\mathfrak{H}(\mathbb{E})\to\mathbb{Z}_{>0}.$$

It is also clear that the function c is invariant under the permutation action of  $W_{\text{aff}}(\mathbb{E})$  on  $\mathfrak{H}(\mathbb{E})$ .

2.4.9. Canonicity of  $(\mathbb{E}, W_{\text{aff}}(\mathbb{E}), I(A))$ . So far we have used the choice of  $A \in \mathfrak{F}^{[M]}$  to define the affine Coxeter group  $W_{\text{aff}}^A = (W_{\text{aff}}(\mathbb{E}), I(A))$ . If  $A_1 \in \mathfrak{F}^{[M]}$  is another facet with  $\mathbb{E}_1 = \text{Span}(A_1)$ , then the affine Coxeter group  $W_{\text{aff}}^{A_1}(W_{\text{aff}}(\mathbb{E}_1), I(A_1))$  is also defined and it acts on  $\mathbb{E}_1$ . Suppose  $A_1$  and A are in the same  $W_{\text{aff}}$ -orbit; we claim that there is a *canonical* affine isometry  $\mathbb{E} \cong \mathbb{E}_1$  and a canonical isomorphism of Coxeter groups  $W_{\text{aff}}^A \cong W_{\text{aff}}^{A_1}$  under which the action of  $W_{\text{aff}}(\mathbb{E})$  on  $\mathbb{E}$  and the action of  $W_{\text{aff}}(\mathbb{E}_1)$  on  $\mathbb{E}_1$  are intertwined.

In fact, let  $w \in W_{\text{aff}}$  be such that  $A_1 = wA$ . Then w is well-defined in the coset  $W_{\text{aff}}/W_A = W_{\text{aff}}/W_{\mathbb{E}}$ . The element w defines an affine isometry  $e_w : \mathbb{E} \cong \mathbb{E}_1$ . Conjugation by w gives an isomorphism  $\operatorname{Ad}(w) : W_{\text{aff}}(\mathbb{E}) \cong N_{W_{\text{aff}}}(W_{\mathbb{E}})/W_{\mathbb{E}} \xrightarrow{\sim} N_{W_{\text{aff}}}(W_{\mathbb{E}_1})/W_{\mathbb{E}_1} \cong W_{\text{aff}}(\mathbb{E}_1)$ , sending I(A) bijectively to  $I(A_1)$ . The isomorphisms  $e_w$  and  $\operatorname{Ad}(w)$  clearly intertwine the action of  $W_{\text{aff}}(\mathbb{E})$  on  $\mathbb{E}$  and the action of  $W_{\text{aff}}(\mathbb{E}_1)$  on  $\mathbb{E}_1$ . Changing w into  $ww_1$  for some  $w_1 \in W_{\mathbb{E}}$ , we have  $e_{ww_1} = e_w \circ e_{w_1}$  and  $\operatorname{Ad}(ww_1) = \operatorname{Ad}(w)\operatorname{Ad}(w_1)$ . Since  $w_1 \in W_{\mathbb{E}}$  fixes  $\mathbb{E}$  pointwise, both  $e_{w_1}$  and  $\operatorname{Ad}(w_1)$  are the identity isomorphisms. Therefore both  $e_w$  and  $\operatorname{Ad}(w)$  are independent of the choice of  $w \in W_{\text{aff}}$  sending A to  $A_1$ .

Under the canonical isomorphism  $(\mathbb{E}, W_{\text{aff}}(\mathbb{E}), I(A)) \cong (\mathbb{E}_1, W_{\text{aff}}(\mathbb{E}_1), I(A_1))$  described above, the set of hyperplanes  $\mathfrak{H}(\mathbb{E})$  maps bijectively to  $\mathfrak{H}(\mathbb{E}_1)$ , and the assignments  $H \mapsto \alpha_H, \alpha_H^{\vee}$  and  $c_H$  are compatible with such bijections. Moreover, the isomorphism  $\mathbb{E} \cong \mathbb{E}_1$  extends uniquely to a linear isomorphism  $\mathbb{E}_{\diamond} \cong \mathbb{E}_{1,\diamond}$  preserving the linear function  $\delta$ , under which the linear actions of  $W_{\text{aff}}(\mathbb{E})$  and  $W_{\text{aff}}(\mathbb{E}_1)$  are intertwined.

2.5. The graded DAHA attached to  $W_{\text{aff}}^A$ . We retain the setup as in the beginning of §2.4.

2.5.1. Fix  $A \in \mathfrak{F}^{[M]}$  and let  $\mathbb{E} = \text{Span}(A)$ ,  $M = G_{\mathbb{E}}$ . For each  $i \in I(A)$  (which is a subset of  $\mathfrak{H}(\mathbb{E})$ ) indexing a simple reflection  $s_i \in W_{\text{aff}}(\mathbb{E})$ , we have defined the simple root  $\alpha_i \in \mathfrak{a}^{M,*}$ , simple coroot  $\alpha_i^{\vee} \in \mathfrak{a}^M$ , and a positive integer  $c_i$  in §2.4.7 and §2.4.8.

We define a graded Q-algebra  $\mathbb{H}_c(W^A_{\text{aff}})$  as follows. As a Q-vector space it is a tensor product

(2.4) 
$$\mathbb{H}_{c}(W_{\mathrm{aff}}^{A}) = \mathbb{Q}[u] \otimes \mathrm{Sym}(\mathbb{E}_{\Diamond}^{*}) \otimes \mathbb{Q}[W_{\mathrm{aff}}(\mathbb{E})].$$

The grading on  $\mathbb{H}_c(W^A_{\mathrm{aff}})$  is determined by assigning u and  $\mathbb{E}^*_{\Diamond}$  degree 2, and assigning  $W_{\mathrm{aff}}(\mathbb{E})$  degree 0. The algebra structure on  $\mathbb{H}_c(W^A_{\mathrm{aff}})$  is determined by:

- Each of the tensor factors in (2.4) is a subalgebra.
- u is central in  $\mathbb{H}_c(W^A_{\text{aff}})$ .
- For each simple reflection  $s_i \in W_{\text{aff}}(\mathbb{E})$   $(i \in I(A))$ , we have

 $(2.5) \ (1 \otimes 1 \otimes s_i) \cdot (1 \otimes v \otimes 1) - (1 \otimes s_i v \otimes 1) \cdot (1 \otimes 1 \otimes s_i) = c_i \langle v, \alpha_i^{\vee} \rangle u \otimes 1 \otimes 1 \quad \forall v \in \mathbb{E}_{\diamond}^*.$ 

Here the pairing  $\langle v, \alpha_i^{\vee} \rangle$  makes sense because  $\mathfrak{a}^M$  is a subspace of  $\mathbb{E}_{\Diamond}$ .

2.5.2. Specialization. The element  $\delta \in \mathbb{E}^*_{\Diamond}$  is invariant under the action of  $W_{\text{aff}}(\mathbb{E})$ , therefore it is central in  $\mathbb{H}_c(W^A_{\text{aff}})$  by the relation (2.5). For  $\nu \in \mathbb{Q}$ , we define a quotient of  $\mathbb{H}_c(W^A_{\text{aff}})$  given by

$$\mathbb{H}_{c,\nu}(W^A_{\text{aff}}) = \mathbb{H}_c(W^A_{\text{aff}})/(u+\nu,\delta-1).$$

Now  $\mathbb{H}_{c,\nu}(W_{\mathrm{aff}}^A)$  is no longer graded but only filtered: if we let  $F_{\leq i} = F_{\leq i} \mathbb{H}_{c,\nu}(W_{\mathrm{aff}}^A)$ be the image of degree  $\leq i$  elements in  $\mathbb{H}_c(W_{\mathrm{aff}}^A)$  for  $i \in \mathbb{Z}_{\geq 0}$ , then  $F_{\leq i} \cdot F_{\leq j} \subset F_{\leq i+j}$ for all  $i, j \in \mathbb{Z}_{\geq 0}$ .

2.5.3. By the discussion in §2.4.9, if  $A, A_1 \in \mathfrak{F}^{[M]}$  are in the same  $W_{\text{aff}}$ -orbit, then there is a *canonical* isomorphism of  $\mathbb{Q}[u, \delta]$ -algebras  $\mathbb{H}_c(W_{\text{aff}}^A) \cong \mathbb{H}_c(W_{\text{aff}}^{A_1})$ , and a canonical isomorphism of algebras  $\mathbb{H}_{c,\nu}(W_{\text{aff}}^A) \cong \mathbb{H}_{c,\nu}(W_{\text{aff}}^{A_1})$  for any  $\nu \in \mathbb{Q}$ .

#### 3. Facets and spirals

In this section we describe how to obtain spirals for the  $\mathbb{Z}/m\mathbb{Z}$ -graded Lie algebra  $\mathfrak{g}$  from facets in  $\mathfrak{a}$ . The results in this section are closely related to, and sometimes reformulations of results in [10, §10] (see §3.4.11 for comparison with the notation in [10]).

# 3.1. $\mathbb{Z}/m\mathbb{Z}$ -gradings on $\mathfrak{g}$ .

3.1.1. Let  $G, \mathfrak{g}$  and the  $\mathbb{Z}/m\mathbb{Z}$ -grading on  $\mathfrak{g}$  be given as in the Introduction. Such a  $\mathbb{Z}/m\mathbb{Z}$ -grading on  $\mathfrak{g}$  is the same datum as a homomorphism  $\theta : \mu_m \to \operatorname{Aut}(G)$ . Given such a  $\theta$ , the graded piece  $\mathfrak{g}_i$  is the subspace of  $\mathfrak{g}$  where  $\theta(\zeta)$  acts by  $\zeta^i$ , for all  $\zeta \in \mu_m$ .

Consider the composition  $\theta_{\text{Out}} : \mu_m \xrightarrow{\theta} \text{Aut}(G) \to \text{Out}(G)$ . Since elements in Out(G) have order 1, 2, or 3, there is a unique  $e \in \{1, 2, 3\}$  and an injective homomorphism  $\sigma : \mu_e \hookrightarrow \text{Out}(G)$  such that  $\theta_{\text{Out}}$  factors as

$$\theta_{\text{Out}}: \mu_m \xrightarrow{[m/e]} \mu_e \xrightarrow{\sigma} \text{Out}(G),$$

where [m/e] is the map  $\zeta \mapsto \zeta^{m/e}$ . We call  $\sigma$  the *outer class* of the  $\mathbb{Z}/m\mathbb{Z}$ -grading  $\theta$ . The discussions in §2 are then applicable to the pair  $(G, \sigma)$ . In particular, we fix a maximal torus  $T \subset G^{\sigma}$  as in §2.1.2.

3.1.2. By Lemma 2.2.2(1), for a primitive element  $\zeta \in \mu_m$ , the element  $\theta(\zeta)$  is *G*-conjugate to an element in  $T^{\mathrm{ad}} \times \sigma(\zeta^{m/e})$ . Therefore,  $\theta$  is *G*-conjugate to a homomorphism

(3.1) 
$$\theta_{\overline{x}} = (\overline{x}, [m/e]) : \mu_m \to T^{\mathrm{ad}} \times \mu_e,$$

where the first factor is given by an element  $\overline{x} \in \mathbb{X}_*(T^{\mathrm{ad}}) \otimes \mathbb{Z}/m\mathbb{Z}$ . Therefore, without loss of generality, we may assume that  $\theta$  takes the form (3.1) for some element  $\overline{x} \in \mathbb{X}_*(T^{\mathrm{ad}}) \otimes \mathbb{Z}/m\mathbb{Z}$ .

3.1.3. For the rest of the section, we fix an element  $x \in \mathbb{X}_*(T^{\mathrm{ad}})$  with image  $\overline{x} \in \mathbb{X}_*(T^{\mathrm{ad}}) \otimes \mathbb{Z}/m\mathbb{Z}$ . In other words, x is a cocharacter  $\mathbb{G}_m \to T^{\mathrm{ad}}$  whose restriction to  $\mu_m$  gives the first factor of  $\theta_{\overline{x}}$ .

3.1.4. Lemma. The point  $x/m \in \mathbb{X}_*(T^{\mathrm{ad}})_{\mathbb{Q}} = \mathfrak{a}$  defines a pseudo-Levi subgroup  $G_{x/m}$  as in §2.2.6 and §2.2.5. We have  $G_{\underline{0}} = G_{x/m}$ , and T is a maximal torus of  $G_{\underline{0}}$ .

*Proof.* Let  $\zeta$  be a primitive *m*th root of unity. By definition  $G_{\underline{0}} = G^{\tau}$  for  $\tau = x(\zeta)\sigma(\zeta^{m/e})$ . The calculation in §2.2.5 shows that  $G^{\tau} = G_{x/m}$ . By Lemma 2.2.2(2), T is a maximal torus of  $G^{\tau} = G_{\underline{0}}$ .

The proof above shows that

(3.2) 
$$\mathfrak{g}_{\underline{n}} = \bigoplus_{(\alpha,i)\in\widetilde{\Phi}(G,T);\langle\alpha,x\rangle/m+i/e\in n/m+\mathbb{Z}}\mathfrak{g}^{i}(\alpha)$$

The Weyl group  $W_{G_{\underline{0}}}$  of  $G_{\underline{0}}$  can be identified with  $W_{x/m}$ , the stabilizer of x/m under the affine Weyl group  $W_{\text{aff}}$ .

## 3.2. Graded pseudo-Levi subgroups.

3.2.1. Let  $\mathfrak{m} \subset \mathfrak{g}$  be a subalgebra. A  $\mathbb{Z}$ -grading  $\mathfrak{m}_*$  on  $\mathfrak{m}$  is said to be *compatible* with the given  $\mathbb{Z}/m\mathbb{Z}$ -grading on  $\mathfrak{g}$  if for any  $i \in \mathbb{Z}/m\mathbb{Z}$  we have

$$\mathfrak{g}_i\cap\mathfrak{m}=igoplus_{n\equiv i(m)}\mathfrak{m}_n.$$

3.2.2. Let  $\mathfrak{M}^{\mathbb{Z}-\mathrm{gr}}$  be the set of quadruples  $(M, M_0, \mathfrak{m}, \mathfrak{m}_*)$  where M is a pseudo-Levi subgroup of  $(G, \sigma)$  with Lie algebra  $\mathfrak{m}, \mathfrak{m} = \bigoplus_{n \in \mathbb{Z}} \mathfrak{m}_n$  is a  $\mathbb{Z}$ -grading on  $\mathfrak{m}$ compatible with the  $\mathbb{Z}/m\mathbb{Z}$ -grading on  $\mathfrak{g}$ , and  $M_0 = e^{\mathfrak{m}_0}$ . The adjoint representation of  $G_0$  on  $\mathfrak{g}$  induces an action of  $G_0$  on  $\mathfrak{M}^{\mathbb{Z}-\mathrm{gr}}$ , and we denote the quotient  $G_0 \setminus \mathfrak{M}^{\mathbb{Z}-\mathrm{gr}}$ by  $\mathfrak{M}^{\mathbb{Z}-\mathrm{gr}}$ . Let  $\mathfrak{M}_T^{\mathbb{Z}-\mathrm{gr}} \subset \mathfrak{M}^{\mathbb{Z}-\mathrm{gr}}$  be the subset of quadruples  $(M, M_0, \mathfrak{m}, \mathfrak{m}_*)$  such that  $M \in \mathfrak{M}_T$  and  $T \subset M_0$ .

If  $(M, M_0, \mathfrak{m}, \mathfrak{m}_*) \in \mathfrak{M}_T^{\mathbb{Z}^{-}\mathrm{gr}}$ , then  $T \subset M_0$ , hence the  $\mathbb{Z}$ -grading on  $\mathfrak{m}$  is given by a unique element  $j \in \mathbb{X}_*(T/Z_M)$ . Hence we get a bijection

$$\mathfrak{M}_T^{\mathbb{Z}-\mathrm{gr}} \cong \{ (M, j) | M \in \mathfrak{M}_T, j \in \mathbb{X}_*(T/Z_M) \}.$$

The Weyl group  $W_{G_{\underline{0}}}$  of  $G_{\underline{0}}$  acts on  $\mathfrak{M}_T^{\mathbb{Z}-\mathrm{gr}}$ .

3.2.3. Lemma. The natural map  $\mathfrak{M}_T^{\mathbb{Z}-\mathrm{gr}} \to \underline{\mathfrak{M}}^{\mathbb{Z}-\mathrm{gr}}$  induces an injective map

$$W_{G_{\underline{0}}} \backslash \mathfrak{M}_{T}^{\mathbb{Z}-\mathrm{gr}} \hookrightarrow \underline{\mathfrak{M}}^{\mathbb{Z}-\mathrm{gr}}$$

Proof. Suppose  $(M, M_0, \mathfrak{m}, \mathfrak{m}_*)$  and  $(M', M'_0, \mathfrak{m}', \mathfrak{m}'_*)$  are both in  $\mathfrak{M}_T^{\mathbb{Z}-\mathrm{gr}}$ , and  $g \in G_{\underline{0}}$  is such that  $\operatorname{Ad}(g)(M, M_0, \mathfrak{m}, \mathfrak{m}_*) = (M', M'_0, \mathfrak{m}', \mathfrak{m}'_*)$ . Since  $\operatorname{Ad}(g)M_0 = M'_0$ ,  $\operatorname{Ad}(g)T$  and T are both maximal tori of  $M'_0$ , there exists  $m' \in M'_0$  such that  $\operatorname{Ad}(m')\operatorname{Ad}(g)T = T$ . Replacing g by m'g, we have  $\operatorname{Ad}(g)T = T$ , hence  $g \in N_{G_{\underline{0}}}(T)$ . Since T acts trivially on  $\mathfrak{M}_T^{\mathbb{Z}-\mathrm{gr}}$ , letting  $w \in W_{G_{\underline{0}}}$  be the image of g, then  $w \cdot (M, M_0, \mathfrak{m}, \mathfrak{m}_*) = (M', M'_0, \mathfrak{m}', \mathfrak{m}'_*)$ . This shows that the fibers of  $\mathfrak{M}_T^{\mathbb{Z}-\mathrm{gr}} \to \mathfrak{M}_T^{\mathbb{Z}-\mathrm{gr}}$  are  $W_{G_{\underline{0}}}$ -orbits.

## 3.3. Spirals and splittings.

3.3.1. We recall from [9] some basic definitions about spirals. Recall  $\epsilon \in \{\pm 1\}$  is the sign of  $\eta$ . An element  $\lambda \in Y_{G_{\underline{0}}, \underline{\mathbb{Q}}}$  (see [9, §0.11], it means  $\lambda$  is a formal expression  $\lambda'/N$  where  $\lambda' : \mathbb{G}_m \to G_{\underline{0}}$  and  $N \in \mathbb{Z}_{>0}$ ) induces a  $\mathbb{Q}$ -grading  $\mathfrak{g} = \bigoplus_{r \in \mathbb{Q}} {\lambda \choose r} \mathfrak{g}$  on  $\mathfrak{g}$ such that each  $\mathfrak{g}_i$  is the direct sum of  ${\lambda \choose r} \mathfrak{g}_i = {\lambda \choose r} \mathfrak{g} \cap \mathfrak{g}_i$  for various  $r \in \mathbb{Q}$ . Using  $\lambda$  we define

(3.3) 
$${}^{\epsilon}\mathfrak{p}_n^{\lambda} := {}^{\lambda}_{\geq \epsilon n}\mathfrak{g}_{\underline{n}} = \bigoplus_{r \in \mathbb{Q}, r \geq \epsilon n} ({}^{\lambda}_r \mathfrak{g}_{\underline{n}}) \quad \forall n \in \mathbb{Z}.$$

Recall from [9, §2.5] that an  $\epsilon$ -spiral for the  $\mathbb{Z}/m\mathbb{Z}$ -graded  $\mathfrak{g}$  is a collection of subspaces  $\mathfrak{p}_* = \{\mathfrak{p}_n | n \in \mathbb{Z}\}$  of the form  ${}^{\epsilon}\mathfrak{p}_*^{\lambda}$  for some  $\lambda \in Y_{G_{\underline{0}}, \mathbb{Q}}$ .

We also recall

$${}^{\epsilon}\mathfrak{l}_n^{\lambda} := {}^{\lambda}_{\epsilon n}\mathfrak{g}_{\underline{n}}$$

Since  $\epsilon$  is fixed throughout, we shall omit the left superscript  $\epsilon$  from the notation such as  ${}^{\epsilon}\mathfrak{p}_{n}^{\lambda}$  and  ${}^{\epsilon}\mathfrak{l}_{n}^{\lambda}$ .

The direct sum  $\mathfrak{l}^{\lambda} = \bigoplus_{n \in \mathbb{Z}} \mathfrak{l}_{n}^{\lambda}$  is a reductive subalgebra of  $\mathfrak{g}$ , and we let  $L^{\lambda} = e^{\mathfrak{l}^{\lambda}}$ ,  $L_{0}^{\lambda} = e^{\mathfrak{l}_{0}^{\lambda}}$ . Note that in [9],  $\mathfrak{l}_{n}^{\lambda}$  was denoted  $\widetilde{\mathfrak{l}}_{n}^{\lambda}$ ,  $L^{\lambda}$  was denoted  $\widetilde{\Lambda}^{\lambda}$ , etc.; in this paper we omit the tildes.

Recall from [9, §2.6] that a *splitting* of the spiral  $\mathfrak{p}_*$  is a system of the form  $(L^{\lambda}, L_0^{\lambda}, \mathfrak{l}^{\lambda}, \mathfrak{l}^{\lambda}_*)$  for some  $\lambda \in Y_{G_0, \mathbb{Q}}$  such that  $\mathfrak{p}_* = \mathfrak{p}_*^{\lambda}$ . The following statement is proved in [9, 2.6(c)].

3.3.2. Lemma. For  $\lambda \in Y_{G_0,\mathbb{Q}}$ , the group  $L^{\lambda}$  is a pseudo-Levi subgroup of  $(G, \sigma)$ .

3.3.3. Let  $\underline{\mathfrak{P}}$  be the set of  $G_{\underline{0}}$ -conjugacy classes of  $\epsilon$ -spirals for the  $\mathbb{Z}/m\mathbb{Z}$ -graded  $\mathfrak{g}$ . By Lemma 3.3.2, assigning to an  $\epsilon$ -spiral  $\mathfrak{p}_*$  any of its splittings (which are conjugate under  $P_0 = e^{\mathfrak{p}_0}$  by [9, §2.7(a)]), we get a map

$$\underline{\Lambda}: \underline{\mathfrak{P}} \to \underline{\mathfrak{M}}^{\mathbb{Z}-\mathrm{gr}}.$$

3.4. **Spirals and facets.** In this subsection we discuss the relationship between spirals and facets in  $\mathfrak{a}$ . The key ingredients are summarized in the diagram (3.6). Some results here are reformulations of results in [10, §10]. We also point out the relationship between spirals and parahoric subalgebras in the loop Lie algebra in §3.4.13.

3.4.1. Let  $\mathfrak{P}_T$  be the set of  $\epsilon$ -spirals  $\mathfrak{p}_*$  of the form  $\mathfrak{p}_*^{\lambda}$  for some  $\lambda \in \mathfrak{a}$  (note that  $\mathfrak{a} = Y_{T,\mathbb{Q}}$  is naturally a subset of  $Y_{G_{\underline{0}},\mathbb{Q}}$ ). The Weyl group  $W_{G_{\underline{0}}}$  acts on  $\mathfrak{P}_T$  by conjugation.

3.4.2. Lemma. The natural map  $\mathfrak{P}_T \to \mathfrak{P}$  induces a bijection

$$W_{G_{\underline{0}}} \setminus \mathfrak{P}_T \xrightarrow{\sim} \mathfrak{P}.$$

*Proof.* First, the map  $\mathfrak{P}_T \to \underline{\mathfrak{P}}$  is surjective because every  $\lambda \in Y_{G_{\underline{0}},\mathbb{Q}}$  is  $G_{\underline{0}}$ -conjugate to one in  $\mathfrak{a}$ .

Now suppose for  $\lambda, \lambda' \in \mathfrak{a}$ ,  $\mathfrak{p}_*^{\lambda}$  and  $\mathfrak{p}_*^{\lambda'}$  are  $G_0$ -conjugate to each other. Let  $g \in G_0$  be such that  $\operatorname{Ad}(g)\mathfrak{p}_*^{\lambda'} = \mathfrak{p}_*^{\lambda}$ , i.e.,  $\mathfrak{p}_*^{\operatorname{Ad}(g)\lambda'} = \mathfrak{p}_*^{\lambda}$ . Let  $P_0 = e^{\mathfrak{p}_0^{\lambda}}$ ,  $L_0 = e^{\mathfrak{l}_0^{\lambda}}$  and  $P_0' = e^{\mathfrak{p}_0^{\lambda'}}$ ,  $L_0' = e^{\mathfrak{l}_0^{\lambda'}}$ . Then  $P_0$  is a parabolic subgroup of  $G_0$  with Levi subgroup  $L_0$ ;  $P_0'$  is a parabolic subgroup of  $G_0$  are Levi subgroups of  $P_0$ . Therefore, there exists  $p \in P_0$  such that  $\operatorname{Ad}(p)\operatorname{Ad}(g)L_0' = L_0$ . Replacing g by pg, we have  $\operatorname{Ad}(g)(P_0', L_0') = (P_0, L_0)$ . Since  $\mathfrak{l}_0 = \operatorname{Ad}(g)\mathfrak{l}_0' = \frac{\operatorname{Ad}(g)\lambda'}{\mathfrak{g}_0}$  contains  $\mathfrak{t}$ , then for any integer N such that  $N\lambda'$  defines a homomorphism  $N\lambda' : \mathbb{G}_m \to G_0$ , the image of  $\operatorname{Ad}(g)(N\lambda')$  commutes with T. Therefore  $\operatorname{Ad}(g)(N\lambda')$  has image in T because T is a maximal torus in  $G_0$ , hence  $\operatorname{Ad}(g)\lambda' \in \mathfrak{a}$ . Since  $\lambda'$  and  $\operatorname{Ad}(g)\lambda'$  are both in  $\mathfrak{a}$  and are conjugate under  $\overline{G_0}$ , they are in the same  $W_{G_0}$ -orbit. Therefore there exists  $w \in W_{G_0}$  such that  $\operatorname{Ad}(g)\lambda' = w\lambda'$ , hence  $\mathfrak{p}_*^{\lambda} = \mathfrak{p}_*^{\operatorname{Ad}(g)\lambda'} = \mathfrak{p}_*^{w\lambda'} = w\mathfrak{p}_*^{\lambda'}$ , i.e., the spirals  $\mathfrak{p}_*^{\lambda}$  and  $\mathfrak{p}_*^{\lambda'}$  are in the same  $W_{G_0}$ -orbit.

3.4.3. By (3.2), we have

$$\mathfrak{l}^{\lambda} = \bigoplus_{(\alpha,i) \in \widetilde{\Phi}(G,T); \langle \alpha, x - \epsilon \lambda \rangle / m + i/e \in \mathbb{Z}} \mathfrak{g}^{i}(\alpha)$$

Comparing with the root system of  $G_{(x-\epsilon\lambda)/m}$  (see §2.2.5), we see that

(3.4) 
$$\mathfrak{l}^{\lambda} = \mathfrak{g}_{(x-\epsilon\lambda)/m}; \quad L^{\lambda} = G_{(x-\epsilon\lambda)/m}$$

For  $y \in \mathfrak{a}$ , we let

$$\lambda_y := \epsilon(x - my);$$

then we have  $L^{\lambda_y} = G_y$ .

3.4.4. Lemma. Let  $y, y' \in \mathfrak{a}$ ; then  $\mathfrak{p}_*^{\lambda_y} = \mathfrak{p}_*^{\lambda_{y'}}$  if and only if y and y' are in the same facet of  $\mathfrak{a}$ .

Proof. We argue for the case  $\epsilon = 1$  and the case  $\epsilon = -1$  is similar. From (3.2) and (3.3) we see that  $\mathfrak{p}_n^{\lambda_y}$  is the sum of  $\mathfrak{g}^i(\alpha)$  where  $(\alpha, i) \in \widetilde{\Phi}(G, T)$  such that  $(\langle \alpha, x \rangle - n)/m + i/e \in \mathbb{Z}$  and that  $(\langle \alpha, x \rangle - n)/m \ge \langle \alpha, y \rangle$ . In other words the condition for  $\mathfrak{g}^i(\alpha)$  to appear in  $\mathfrak{p}_n^{\lambda_y}$  is that for some  $\ell \in \mathbb{Z}$  with residue class  $i \mod e$ , we have  $(\langle \alpha, x \rangle - n)/m + \ell/e = 0 \ge \langle \alpha, y \rangle + \ell/e$ . I.e.,  $\mathfrak{g}^i(\alpha)$  appears in  $\mathfrak{p}_n^{\lambda_y}$  if and only if there exists an affine root  $\widetilde{\alpha} = \alpha + \ell/e$  with image  $(\alpha, i) \in \widetilde{\Phi}(G, T)$  such that

(3.5) 
$$\widetilde{\alpha}(x/m) - n/m = 0 \ge \widetilde{\alpha}(y).$$

Suppose y and y' are in the same facet; then for any affine real root  $\tilde{\alpha}$ ,  $\tilde{\alpha}(y)$  and  $\tilde{\alpha}(y')$  are either both positive or both negative or both zero. Therefore the condition (3.5) holds for y if and only if it holds for y'. Hence  $\mathfrak{p}_*^{\lambda_y} = \mathfrak{p}_*^{\lambda_{y'}}$ .

Conversely, suppose y and y' do not lie in the same facet; then there exists a real affine root  $\tilde{\alpha} \in \Phi_{\text{aff}}$  such that  $\tilde{\alpha}(y) \leq 0$  but  $\tilde{\alpha}(y') > 0$ . Let  $(\alpha, i) \in \tilde{\Phi}(G, T)$  be the image of  $\tilde{\alpha}$ . Let  $n = m\tilde{\alpha}(x/m) \in \mathbb{Z}$ ; then by (3.5) we see that  $\mathfrak{g}^i(\alpha)$  is contained in  $\mathfrak{p}_n^{\lambda_y}$  but not in  $\mathfrak{p}_n^{\lambda_y'}$ . Therefore  $\mathfrak{p}_*^{\lambda_y}$  is not equal to  $\mathfrak{p}_*^{\lambda_{y'}}$  in this case.

3.4.5. Let  $F \in \mathfrak{F}$ . We define the  $\epsilon$ -spiral  $\mathfrak{p}_*^F$  as  $\mathfrak{p}_*^{\lambda_y}$  for any  $y \in F$ . This is welldefined by Lemma 3.4.4. The map  $F \mapsto \mathfrak{p}_*^F$  then gives a bijection

$$\Delta_x:\mathfrak{F}\xrightarrow{\sim}\mathfrak{P}_T.$$

The notation  $\Delta_x$  suggests that it depends on x and not just on  $\overline{x}$ .

We define a splitting  $(L^F, L_0^F, \mathfrak{l}^F, \mathfrak{l}^F)$  of  $\mathfrak{p}_*^F$  as  $(L^{\lambda_y}, L_0^{\lambda_y}, \mathfrak{l}^{\lambda_y}, \mathfrak{l}^{\lambda_y})$  for any  $y \in F$ . Then the assignment  $F \mapsto (L^F, L_0^F, \mathfrak{l}^F, \mathfrak{l}^F_*) \in \mathfrak{M}_T$  gives a map

$$\Lambda_T:\mathfrak{P}_T\xrightarrow{\Delta_x^{-1}}\mathfrak{F}\to\mathfrak{M}_T^{\mathbb{Z}-\mathrm{gr}}.$$

Alternatively, for  $\mathfrak{p}_* \in \mathfrak{P}_T$ , the element  $\Lambda_T(\mathfrak{p}_*)$  may be characterized as the unique splitting  $(L, L_0, \mathfrak{l}, \mathfrak{l}_*)$  of  $\mathfrak{p}_*$  such that  $T \subset L_0$ , but we shall not need this statement.

3.4.6. Lemma. For any  $\mathbb{E} \in \mathfrak{E}$ , denoting  $G_{\mathbb{E}}$  by M, we define  $j_x \in \mathfrak{a}_M$  by

$$j_x = m(\pi_M(x/m) - \pi_M(\mathbb{E})),$$

where we recall  $\pi_M : \mathfrak{a} \to \mathfrak{a}_M = \mathfrak{a}/\mathfrak{a}^M$  is the projection. Then  $j_x \in \mathbb{X}_*(T/Z_M) \subset \mathfrak{a}_M$ , and therefore  $j_x$  defines a  $\mathbb{Z}$ -grading  $\mathfrak{m}_*$  on  $\mathfrak{m} = \text{Lie } M$ .

Proof. For  $F \in Alc(\mathbb{E})$ , we denote  $(L^F, L_0^F, \mathfrak{l}^F, \mathfrak{l}^F_*)$  by  $(M, M_0, \mathfrak{m}, \mathfrak{m}_*) \in \mathfrak{M}_T^{\mathbb{Z}-\mathrm{gr}}$ . Let  $y \in F$ , and  $\lambda_y = \epsilon(x - my)$ . Then by §3.4.5 and (3.4), we have  $M = G_y$ . By definition, the  $\mathbb{Z}$ -grading on  $\mathfrak{m} = \mathfrak{l}^F$  is the restriction of the  $\mathbb{Q}$ -grading  $\epsilon \lambda_y = x - my$  of  $\mathfrak{g}$ . Therefore, the  $\mathbb{Z}$ -grading on  $\mathfrak{m}$  is given by the image  $\pi_M(\lambda_y) = \pi_M(x - my) = m(\pi_M(x/m) - \pi_M(\mathbb{E})) = j_x$ . This shows that  $j_x$  has integral weights on  $\mathfrak{m}$  and hence  $j_x \in \mathbb{X}_*(T/Z_M)$ .

3.4.7. By Lemma 3.4.6, the assignment  $\mathbb{E}\mapsto (M=G_{\mathbb{E}},\jmath_x)$  defines a map

$$\Gamma_x: \mathfrak{E} \to \mathfrak{M}_T^{\mathbb{Z}-\mathrm{gr}}$$

We claim that this map is injective. In fact, if both  $\mathbb{E}$  and  $\mathbb{E}' \in \mathfrak{E}$  give the same  $(M, M_0, \mathfrak{m}, \mathfrak{m}_*) \in \mathfrak{M}_T^{\mathbb{Z}-\mathrm{gr}}$ , then  $\mathbb{E}$  is parallel to  $\mathbb{E}'$  because they are both parallel to  $\mathfrak{a}^M$ . Moreover, the gradings on  $\mathfrak{m}$  induced by  $m(\pi_M(x/m) - \pi_M(\mathbb{E}))$  and  $m(\pi_M(x/m) - \pi_M(\mathbb{E}'))$  are the same, hence  $\pi_M(\mathbb{E}) = \pi_M(\mathbb{E}')$ . Therefore  $\mathbb{E} = \mathbb{E}'$ .

3.4.8. Recall that an *admissible system* is  $(M, M_0, \mathfrak{m}, \mathfrak{m}_*, \mathcal{L})$ , where  $(M, M_0, \mathfrak{m}, \mathfrak{m}_*) \in \mathfrak{M}^{\mathbb{Z}-\mathrm{gr}}$  which lies in the image of  $\Lambda : \mathfrak{P} \to \mathfrak{M}^{\mathbb{Z}-\mathrm{gr}}$  (see [9, §3.1(c)]), and  $\mathcal{L}$  is an  $M_0$ -equivariant cuspidal local system on the open  $M_0$ -orbit  $\mathfrak{m}_\eta$  of  $\mathfrak{m}_\eta$  in the sense of [7, §4.4]. Our notation is slightly different from that in [9, §3.1], where  $(M, M_0, \mathfrak{m}, \mathfrak{m}_*, \mathcal{L}^{\sharp})$  is called an admissible system. Let  $\mathfrak{T}_\eta$  be the set of admissible systems, and let  $\mathfrak{T}_\eta = G_0 \backslash \mathfrak{T}_\eta$ . Forgetting the local system  $\mathcal{L}$  gives a map  $\mathfrak{T}_\eta \to \mathfrak{M}^{\mathbb{Z}-\mathrm{gr}}$ , and passes to the  $G_0$ -quotients  $\mathfrak{T}_\eta \to \mathfrak{M}^{\mathbb{Z}-\mathrm{gr}}$ .

3.4.9. The constructions of the maps  $\Delta_x, \Gamma_x$  and  $\Lambda_T$  gives a commutative diagram

Here we have used the isomorphism  $W_{x/m} \cong W_{G_0}$ , the injectivity of  $\Gamma_x$  shown in §3.4.7, Lemma 3.2.3, and the isomorphism proved in Lemma 3.4.2. The composition of the first row is given by  $[F] \mapsto [\mathfrak{p}^F_*]$ .

For  $\xi \in \underline{\mathfrak{T}}_{\eta}$ , let  $\overline{\xi} \in \underline{\mathfrak{M}}^{\mathbb{Z}-\mathrm{gr}}$  be its image. Let  $\mathfrak{M}_{T}^{\mathbb{Z}-\mathrm{gr},\xi}, \mathfrak{P}_{T}^{\xi}, \mathfrak{F}^{\xi}$ , and  $\mathfrak{E}^{\xi}$  be the subsets of  $\mathfrak{M}_{T}^{\mathbb{Z}-\mathrm{gr}}, \mathfrak{P}_{T}, \mathfrak{F}$ , and  $\mathfrak{E}$  which is the preimage of  $\overline{\xi}$  (these subsets only depend on  $\overline{\xi}$ , but we use the superscripts  $\xi$  for simplicity of notation). The diagram (3.6) restricts to a diagram

3.4.10. Corollary. Let  $\xi \in \underline{\mathfrak{T}}_n$ .

- (1) All elements in  $\mathfrak{F}^{\xi}$  are in the same  $W_{\text{aff}}$ -orbit.
- (2) Let  $\mathbb{E} \in \mathfrak{E}^{\xi}$ . Then the inclusion  $\operatorname{Alc}(\mathbb{E}) \subset \mathfrak{F}^{\xi}$  induces a bijection

$$W_{x/m}(\mathbb{E}) \setminus \operatorname{Alc}(\mathbb{E}) \xrightarrow{\sim} W_{x/m} \setminus \mathfrak{F}^{\xi} \cong \underline{\mathfrak{P}}^{\xi},$$

where  $W_{x/m}(\mathbb{E})$  is the image of  $W_{x/m} \cap N_{W_{aff}}(W_{\mathbb{E}})$  in

$$W_{\mathrm{aff}}(\mathbb{E}) \cong N_{W_{\mathrm{aff}}}(W_{\mathbb{E}})/W_{\mathbb{E}}$$

Proof. By the diagram (3.7),  $W_{x/m} \setminus \mathfrak{E}^{\xi}$  is a singleton, i.e., the action of  $W_{x/m}$  on  $\mathfrak{E}^{\xi}$  is transitive. Let  $\mathbb{E} \in \mathfrak{E}^{\xi}$  and  $F \in \mathfrak{F}^{\xi}$ . Then  $\operatorname{Span}(F)$  is in the  $W_{x/m}$ -orbit of  $\mathbb{E}$ , therefore, the facet F is in the  $W_{x/m}$ -orbit of an alcove of  $\mathbb{E}$ . In other words, the map  $\operatorname{Alc}(\mathbb{E}) \to W_{x/m} \setminus \mathfrak{F}^{\xi}$  is surjective. Since  $N_{W_{\operatorname{aff}}}(W_{\mathbb{E}})$  is the stabilizer of  $\mathbb{E}$  under  $W_{\operatorname{aff}}$  (see the discussion before Proposition 2.4.5), two facets in  $\operatorname{Alc}(\mathbb{E})$  are in the same  $W_{x/m}$ -orbit if and only if they are in the same  $W_{x/m} \cap N_{W_{\operatorname{aff}}}(W_{\mathbb{E}})$ -orbit, which implies (2).

If  $A, A_1 \in \mathfrak{F}^{\xi}$ , then  $\mathbb{E} = \operatorname{Span}(A)$  and  $\mathbb{E}_1 = \operatorname{Span}(A_1)$  are both in  $\mathfrak{E}^{\xi}$ . Hence there exists  $w \in W_{x/m}$  such that  $w\mathbb{E} = \mathbb{E}_1$ . Now both  $A_1$  and wA are  $\mathbb{E}_1$ -alcoves. By Proposition 2.4.3(1), there exists  $w_1 \in W_{\operatorname{aff}}(\mathbb{E}_1)$  such that  $w_1wA = A_1$ . Hence  $A_1$  is in the same  $W_{\operatorname{aff}}$ -orbit as A, which proves (1).

3.4.11. Comparison with [10, §10]. We explain the relation between the notation used here and that in [10, §10]. Let  $\mathbb{E} \in \mathfrak{E}^{\xi}$  and  $\Gamma_x(\mathbb{E}) = (M, M_0, \mathfrak{m}, \mathfrak{m}_*)$ . In [10, §10.1], the vector space  $\mathfrak{a}^M$  was denoted by  $\mathbf{E}$ , and we introduced a set of affine hyperplanes (denoted by  $\mathfrak{H}_{\alpha,n,N}$ ) in it. To each  $\varpi \in \mathbf{E}$  we attached an  $\epsilon$ spiral denoted by  $\mathfrak{p}_{\ast}^{\overline{\omega}}$  in [10, §10.2]. We claim that there is an affine isomorphism  $\tau : \mathbf{E} \cong \mathbb{E}$ , sending the set of hyperplanes  $\{\mathfrak{H}_{\alpha,n,N}\}$  bijectively to  $\mathfrak{H}(\mathbb{E})$ , such that if  $\varpi \in \mathbf{E}$  corresponds to  $y \in \mathbb{E}$ ; then  $\mathfrak{p}_{\ast}^{\overline{\omega}} = \mathfrak{p}_{\ast}^{\lambda_y}$ .

First note that in the notation of [10], the grading on  $\mathfrak{m}$  is induced by  $\eta\iota/2 \in \mathbb{X}_*(T)_{\mathbb{Q}}$ , which lies in  $\mathbb{X}_*(T \cap M^{\mathrm{der}})_{\mathbb{Q}}$  because  $\iota$  comes from a homomorphism  $\phi$ : SL<sub>2</sub>  $\to M$ . Therefore  $\pi_M(\eta\iota/2) = \jmath$ . In particular,  $\pi_M(x - \eta\iota/2)/m = (\pi_M(x) - \jmath)/m$  therefore  $(x - \eta\iota/2)/m \in \mathbb{E}$ . Therefore the map  $\varpi \mapsto y = (x - \frac{\eta}{2}(\iota + \varpi))/m$  gives an affine isomorphism  $\tau$  :  $\mathbf{E} = \mathfrak{a}^M \xrightarrow{\sim} \mathbb{E}$ . If  $\varpi \mapsto y$  under this map, then  $\lambda_y = \frac{\epsilon\eta}{2}(\iota + \varpi) = \frac{|\eta|}{2}(\iota + \varpi) = \underline{\varpi}$ , therefore  $\mathfrak{p}_*^{\overline{\varpi}} = \mathfrak{p}_*^{\lambda_y}$ .

Under the isomorphism  $\tau$ , the open subset  $\mathbf{E}'$  introduced in [10, §10.1] (which is the complement of the hyperplanes  $\mathfrak{H}_{\alpha,n,N}$ ) is characterized by the following property: for  $\varpi \in \mathbf{E}$ ,  $\mathfrak{l}_{*}^{\underline{\varpi}} = \mathfrak{m}_{*}$  if and only if  $\varpi \in \mathbf{E}'$  (see [10, 10.2(b)]). On the other hand, the argument of Lemma 3.4.4 shows that for  $y \in \mathbb{E}$ ,  $\mathfrak{l}_{*}^{\lambda_{y}} = \mathfrak{m}_{*}$  if and only if  $G_y = M$ , i.e.,  $y \in \mathbb{E}'$ . Therefore  $\tau(\mathbf{E}') = \mathbb{E}' := \mathbb{E} - \bigcup_{H \in \mathfrak{H}(\mathbb{E})} H$ , hence the set of hyperplanes  $\{\mathfrak{H}_{\alpha,n,N}\}$  in  $\mathbf{E}'$  corresponds to  $\mathfrak{H}(\mathbb{E})$  under  $\tau$ .

The following statement will not be used in the rest of the paper.

3.4.12. Lemma. Let  $\mathbb{E} \in \mathfrak{E}^{\xi}$  and  $(M, M_0, \mathfrak{m}, \mathfrak{m}_*) = \Gamma_x(\mathbb{E})$ . Let  $\mathcal{W}$  be the finite Weyl group introduced in [10, §10.11] which acts faithfully on  $\mathbb{E}$ . Then under the isomorphism  $\tau : \mathbb{E} \cong \mathbb{E}$  in §3.4.11,  $\mathcal{W}$  corresponds to the group  $W_{x/m}(\mathbb{E})$  introduced in Corollary 3.4.10.

Proof. Recall the group H' introduced in [10, §10.9]: it is the normalizer of  $(M, M_0, \mathfrak{m}, \mathfrak{m}_*)$  in  $G_0$ . By [10, §10.9(e)], the group  $M_0$  is the identity component of H', and  $H'/M_0 \cong \mathcal{W}$ . Since T is a maximal torus of  $M_0$ , we have  $N_{H'}(T)/N_{M_0}(T) \cong H'/M_0 \cong \mathcal{W}$ . Since  $W_{x/m}(\mathbb{E}) = (W_{x/m} \cap N_{W_{\mathrm{aff}}}(W_{\mathbb{E}}))/(W_{x/m} \cap W_{\mathbb{E}})$ , the desired isomorphism  $\mathcal{W} \cong W_{x/m}(\mathbb{E})$  follows from two canonical isomorphisms

(3.8) 
$$N_{H'}(T)/T \cong W_{x/m} \cap N_{W_{\text{aff}}}(W_{\mathbb{E}});$$

$$(3.9) N_{M_0}(T)/T \cong W_{x/m} \cap W_{\mathbb{E}}.$$

Since  $H', M_0$  are subgroups of  $G_{\underline{0}}$  containing T, both  $N_{H'}(T)/T$  and  $N_{M_0}(T)/T$  are subgroups of  $W_{G_{\underline{0}}}$ , which is identified with  $W_{x/m}$ . The above isomorphisms will in fact be equalities of subgroups of  $W_{x/m}$ .

We show (3.9). Since  $M_0$  is the subgroup of M that fixes the grading element  $j_x \in \mathfrak{a}_M$ ,  $N_{M_0}(T)/T$  is the subgroup of  $N_M(T)/T = W_{\mathbb{E}}$  that fixes  $j_x$  in  $\mathfrak{a}_M$ . Now suppose  $w \in W_{\mathbb{E}}$ . By the definition of  $j_x$  in Lemma 3.4.6, it is fixed by w if and only if w(x/m) and x/m have the same image under  $\pi_M$ . Since w fixes  $\mathbb{E}$  pointwise, the points w(x/m) and x/m have the same orthogonal projection to  $\mathbb{E}$  as well. Therefore  $w \in N_{M_0}(T)/T \subset W_{\mathbb{E}}$  if and only if w(x/m) = x/m.

We show (3.8). Let  $w \in N_{H'}(T)/T \subset W_{x/m}$ . Now we view x/m as the origin of  $\mathfrak{a}$ , so that w acts linearly on  $\mathfrak{a}$ . Let  $\mathbb{E}'$  be the affine subspace parallel to  $\mathfrak{a}^M$  that passes through x/m. Since w normalizes M, it normalizes  $Z_M^{\circ}$ , hence stabilizes  $\mathbb{E}'$ . Since w normalizes each graded piece of  $\mathfrak{m}$ , it fixes the grading element  $j_x \in \mathfrak{a}_M$ , and therefore fixes  $\langle x/m \rangle_{\mathbb{E}}$  (the projection of x/m to  $\mathbb{E}$ ). These imply that w stabilizes  $\mathbb{E}$  (which is characterized as the affine subspace parallel to  $\mathbb{E}'$  and passing through  $\langle x/m \rangle_{\mathbb{E}}$ ). Therefore  $N_{H'}(T)/T \subset W_{x/m} \cap N_{W_{aff}}(W_{\mathbb{E}})$ .

Conversely, if  $w \in W_{x/m}$  also lies in  $N_{W_{aff}}(W_{\mathbb{E}})$ , then w stabilizes  $\mathbb{E}$ . Again let  $\mathbb{E}'$  be the affine subspace parallel to  $\mathfrak{a}^M$  that passes through x/m; then w stabilizes  $\mathbb{E}'$ , hence induces an action  $\overline{w}$  on the quotient  $\mathfrak{a}_M = \mathfrak{a}/\mathbb{E}'$ . Let  $M' = C_G(Z_M^0)$ ; then M' is the smallest Levi subgroup of G containing M. Now  $\overline{w}$  fixes both  $\pi_M(\mathbb{E}') = \pi_M(x/m)$  and  $\pi_M(\mathbb{E})$  in  $\mathfrak{a}_M$ . Note that  $\mathfrak{a}_M$  has a set of affine hyperplanes  $\mathfrak{H}(\mathfrak{a}_M)$  given by  $\pi_M(H)$  for those  $H \in \mathfrak{H}$  (whenver  $\pi_M(H)$  is an affine hyperplane in  $\mathfrak{a}_M$ ). The root system  $\Phi(M, T)$  up to  $\pm 1$  can be identified with the linear parts of those  $H \in \mathfrak{H}(\mathfrak{a}_M)$  passing through  $\pi_M(\mathbb{E})$ . Since  $\overline{w}$  preserves  $\mathfrak{H}(\mathfrak{a}_M)$  and fixes  $\pi_M(\mathbb{E})$ , it stabilizes  $\Phi(M, T)$ , hence any lift of w to  $N_{G_0}(T)$  stabilizes M. Since  $\overline{w}$  fixes the grading element  $j_x \in \mathfrak{a}_M$  (because it fixes  $\pi_M(x/m)$ ) for  $\mathfrak{m}$ , it normalizes  $(M, M_0, \mathfrak{m}, \mathfrak{m}_*)$ . This finishes the proof of (3.8).

The combination of Corollary 3.4.10, §3.4.11, and Lemma 3.4.12 implies that  $\underline{\mathfrak{P}}^{\xi}$  is in natural bijection with  $W \setminus \operatorname{Alc}(\mathbf{E})$  (where  $\operatorname{Alc}(\mathbf{E})$  as the set of alcoves in  $\mathbf{E}$  defined by the hyperplanes  $\{\mathfrak{H}_{\alpha,n,N}\}$  in [10, §10]). This is essentially a restatement of [10, §10.11(a)].

3.4.13. Spirals and parahoric subalgebras. We clarify the relationship between spirals and parahoric subalgebras in the loop algebra of  $\mathfrak{g}$ . This discussion is independent of the rest of the paper. From  $(\mathfrak{g}, \sigma)$ , we may form the  $\sigma$ -twisted loop algebra  $\widetilde{\mathfrak{g}}$  as in [2, §8.2]:  $\widetilde{\mathfrak{g}} = \mathcal{L}(\mathfrak{g}, \sigma, e)$  is the fixed point subalgebra of  $\mathfrak{g} \otimes_k k(t^{1/e})$  under the diagonal action  $\mu_e \ni \zeta : X \otimes t^{n/e} \mapsto \sigma(\zeta)\zeta^{-n}X \otimes t^{n/e}$ . We can then talk about affine roots of  $\widetilde{\mathfrak{g}}$  as in [2], which includes the real affine roots  $\Phi_{\text{aff}}$  as in §2.1.5 and imaginary roots  $\{n\delta|n \in \mathbb{Z} - \{0\}\}$ . Then  $\widetilde{\mathfrak{g}}$  is the Lie algebra of a quasi-split form  $\mathcal{G}$  of  $\mathcal{G}$  over F = k(t), and  $\mathfrak{a}$  is the apartment attached to the maximal split torus  $T_F$  of  $\mathcal{G}$ . By Bruhat-Tits theory, each point  $y \in \mathfrak{a}$  defines a parahoric subalgebra  $\widetilde{\mathfrak{g}}_{y,\geq 0}$  of  $\widetilde{\mathfrak{g}}$  which is the t-adic completion of the span of the affine root spaces  $\widetilde{\mathfrak{g}}(\widetilde{\alpha})$  such that  $\widetilde{\alpha}(y) \ge 0$  (including imaginary roots). For any  $r \in \mathbb{Q}$  we define  $\widetilde{\mathfrak{g}}_{y,r}$  to be the direct sum of those affine root spaces  $\widetilde{\mathfrak{g}}(\widetilde{\alpha})$  such that  $\widetilde{\alpha}(y) = r$ .

With the choice of  $x \in \mathbb{X}_*(T^{\mathrm{ad}})$  lifting  $\overline{x}$ , the equality (3.2) gives a canonical isomorphism

(3.10) 
$$\mathfrak{g}_n \cong \widetilde{\mathfrak{g}}_{x/m,n/m} \quad \text{for } n \in \mathbb{Z}.$$

We may reinterpret (-1)-spirals  ${}^{-1}\mathfrak{p}_*^{\lambda}$  for the  $\mathbb{Z}/m\mathbb{Z}$ -graded Lie algebra  $\mathfrak{g}$  using parahoric subalgebras of  $\tilde{\mathfrak{g}}$  as follows. Under the isomorphism (3.10), we have for any  $y \in \mathfrak{a}$ 

$${}^{-1}\mathfrak{p}_n^{my-x} = \widetilde{\mathfrak{g}}_{x/m,n/m} \cap \widetilde{\mathfrak{g}}_{y,\geq 0},$$
$${}^{1}\mathfrak{l}_n^{x-my} = {}^{-1}\mathfrak{l}_n^{my-x} = \widetilde{\mathfrak{g}}_{x/m,n/m} \cap \widetilde{\mathfrak{g}}_{y,0} \quad \text{for } n \in \mathbb{Z}.$$

#### 4. The graded DAHA action on spiral inductions

In this section we give the proof of Theorem 1.2.3.

#### 4.1. Some preparation.

4.1.1. We keep the setup of §3.1. In particular, we fix a maximal torus  $T \subset G_{\underline{0}}$ ; the  $\mathbb{Z}/m\mathbb{Z}$ -grading on  $\mathfrak{g}$  is given by  $\theta_{\overline{x}}$  as in (3.1) for an element  $\overline{x} \in \mathbb{X}_*(T^{\mathrm{ad}}) \otimes \mathbb{Z}/m\mathbb{Z}$ . We also fix a lifting  $x \in \mathbb{X}_*(T^{\mathrm{ad}})$  of  $\overline{x}$ .

4.1.2. Canonical graded DAHA. For any  $A \in \mathfrak{F}^{\xi}$ , we have defined an affine Coxeter group  $W_{\mathrm{aff}}^A$  and a graded DAHA  $\mathbb{H}_c(W_{\mathrm{aff}}^A)$  in §2.4 and §2.5. By Corollary 3.4.10, all facets  $A \in \mathfrak{F}^{\xi}$  are in the same  $W_{\mathrm{aff}}$ -orbit. By the discussion in §2.4.9 and §2.5.3, we may canonically identify the various  $W_{\mathrm{aff}}^A$ ,  $\mathbb{E} = \mathrm{Span}(A)$  and  $\mathbb{H}_c(W_{\mathrm{aff}}^A)$  for  $A \in \mathfrak{F}^{\xi}$ . We denote the resulting affine Coxeter group by  $W_{\mathrm{aff}}^{\xi}$  with simple reflections indexed by  $I^{\xi}$  and reflection representation on a canonical Q-vector space  $\mathbb{E}^{\xi}_{\diamond}$ . We have a canonical linear function  $\delta : \mathbb{E}^{\xi}_{\diamond} \to \mathbb{Q}$  and define  $\mathbb{E}^{\xi} = \delta^{-1}(1) \subset \mathbb{E}^{\xi}_{\diamond}$ . The corresponding graded DAHA is denoted by  $\mathbb{H}_c(W_{\mathrm{aff}}^{\xi})$ . We define the specialization  $\mathbb{H}_{c,\nu}(W_{\mathrm{aff}}^{\xi}) = \mathbb{H}_c(W_{\mathrm{aff}}^{\xi})/(u + \nu, \delta - 1)$ .

4.1.3. Spiral induction. Let  $A \in \mathfrak{F}^{\xi}$ . We have two maps from the twisted product  $X^A = G_{\underline{0}} \overset{P_0^A}{\times} \mathfrak{p}_{\eta}^A$  (see §1.5.7)

(4.1) 
$$\mathfrak{g}_{\underline{\eta}} \xleftarrow{\alpha^A} X^A = G_{\underline{0}} \overset{P_0^A}{\times} \mathfrak{p}_{\eta}^A \xrightarrow{\beta^A} [\mathfrak{l}_{\eta}^A/L_0^A] .$$

Here  $\alpha^A(g,v) = \operatorname{Ad}(g)v$  for  $g \in G_{\underline{0}}, v \in \mathfrak{p}^A_\eta$ ;  $\beta^A(g,v)$  is the image of v under  $\mathfrak{p}^A_\eta \to \mathfrak{l}^A_\eta \to \mathfrak{l}^A_\eta \to [\mathfrak{l}^A_\eta/L_0^A]$ .

By definition, there exists  $g \in G_{\underline{0}}$  such that  $\operatorname{Ad}(g)(L^{A}, L_{0}^{A}, \mathfrak{l}^{A}, \mathfrak{l}_{\underline{n}}^{A}) = (M, M_{0}, \mathfrak{m}, \mathfrak{m}_{\underline{n}})$ . Then  $\operatorname{Ad}(g)$  induces an isomorphism of stacks  $\operatorname{Ad}(g) : [\mathfrak{l}_{\eta}^{A}/L_{0}^{A}] \cong [\mathfrak{m}_{\eta}/M_{0}]$ , and the  $L_{0}^{A}$ -equivariant local system  $\operatorname{Ad}(g)^{*}\mathcal{L}$  on the open  $L_{0}^{A}$ -orbit  $\mathring{\mathfrak{l}}_{\eta}^{A}$  is defined. We claim that the isomorphism class of  $\operatorname{Ad}(g)^{*}\mathcal{L}$  is independent of the choice of g. In fact, for a different choice of  $g' \in G_{\underline{0}}$  such that  $\operatorname{Ad}(g')(L^{A}, L_{0}^{A}, \mathfrak{l}^{A}, \mathfrak{l}_{\underline{n}}) = (M, M_{0}, \mathfrak{m}, \mathfrak{m}_{\underline{n}}), g'g^{-1} \in G_{\underline{0}}$  normalizes  $(M, M_{0}, \mathfrak{m}, \mathfrak{m}_{\underline{n}})$ , i.e.,  $h = g'g^{-1} \in H'$  in the notation introduced in [10, §10.9]. By [10, §10.10(a)],  $\operatorname{Ad}(h)^{*}\mathcal{L} \cong \mathcal{L}$ , therefore  $\operatorname{Ad}(g')^{*}\mathcal{L} \cong \operatorname{Ad}(g)^{*}\mathcal{L}$ .

The above discussion shows that there is a canonical cuspidal local system  $\mathcal{L}^A$  on the open  $\mathring{\mathfrak{l}}^A_\eta$  of  $\mathfrak{l}^A_\eta$  that corresponds to  $\mathcal{L}$  under any isomorphism  $(L^A, L^A_0, \mathfrak{l}^A, \mathfrak{l}^A_*) \cong$  $(M, M_0, \mathfrak{m}, \mathfrak{m}_*)$  induced by an element in  $G_{\underline{0}}$ . The spiral induction from  $\mathfrak{p}^A_*$  and  $\mathcal{L}$ is

$$\mathbf{I}_A := \mathbf{I}_{\mathfrak{p}^A_*} = \alpha^A_! \beta^{A,*} \mathcal{L}^{A,\sharp} \in \mathcal{D}_{G_{\underline{0}}}(\mathfrak{g}^{nil}_\eta)_{\xi}.$$

4.1.4. If  $\mathfrak{p}_*, \mathfrak{p}'_* \in \mathfrak{P}^{\xi}$  are conjugate by an element  $g \in G_{\underline{0}}$ , then g induces an isomorphism  $\mathbf{I}_{\mathfrak{p}_*} \cong \mathbf{I}_{\mathfrak{p}'_*}$ . This isomorphism is independent of the choice of g because when  $\mathfrak{p}_* = \mathfrak{p}'_*, g$  must lie in  $P_0 = e^{\mathfrak{p}_0}$  which then induces the identity automorphisms on  $\mathbf{I}_{\mathfrak{p}_*}$ . Therefore, for a  $G_{\underline{0}}$ -orbit  $[\mathfrak{p}_*] \in \mathfrak{P}^{\xi}$ , we have a *canonical* object  $\mathbf{I}_{[\mathfrak{p}_*]} \in \mathcal{D}_{G_{\underline{0}}}(\mathfrak{g}^{nil}_{\underline{n}})_{\xi}$ .

Since  $W_{x/m} \setminus \mathfrak{F}^{\xi} \cong \mathfrak{P}^{\xi}$ , we see that  $\mathbf{I}_A = \mathbf{I}_{\mathfrak{p}^A_*}$  only depends on the  $W_{x/m}$ -orbit of  $A \in \mathfrak{F}^{\xi}$ . For  $[A] \in W_{x/m} \setminus \mathfrak{F}^{\xi}$ , we define  $\mathbf{I}_{[A]}$  to be  $\mathbf{I}_A$  for any  $A \in [A]$ .

4.1.5. Action of the torus  $\mathbb{G}_m^{\text{rot}}$ . For any  $F \in \mathfrak{F}$ , we define an action of  $\mathbb{G}_m^{\text{rot}}$  on the  $\epsilon$ -spiral  $\mathfrak{p}_*^F$  in the following way. For any root space  $\mathfrak{g}^i(\alpha) \subset \mathfrak{p}_n^F$ , let  $\mathbb{G}_m^{\text{rot}}$  act on it by weight  $e(n - \langle \alpha, x \rangle)/m$ . Under this definition, the  $\mathbb{G}_m^{\text{rot}}$ -action on  $\mathfrak{l}^F$  is consistent with the definition in §2.3.

We also define the action of  $\mathbb{G}_m^{\text{rot}}$  on  $\mathfrak{g}_0$  (resp.,  $\mathfrak{g}_{\underline{\eta}}$ ) in a similar way: it acts on a root space  $\mathfrak{g}^i(\alpha)$  by weight  $-e\langle \alpha, x/m \rangle$  (resp.,  $e(\eta - \langle \alpha, x \rangle)/m$ ). Note that the  $\mathbb{G}_m^{\text{rot}}$ -action on  $\mathfrak{g}_{\eta}$  depends on  $\eta$ , and not just  $\eta$ .

4.1.6. Action of the torus  $\mathbb{G}_m^{\flat}$ . Consider the homomorphism

$$s: \mathbb{G}_m \to T^{\mathrm{ad}} \times \mathbb{G}_m^{\mathrm{rot}} \times \mathbb{G}_m^{\mathrm{dil}}$$
$$t \mapsto (x(t), t^{m/e}, t^{-\eta}).$$

We denote its image by  $\mathbb{G}_m^{\flat}$ .

For each facet  $F \in \mathfrak{F}$ , the group  $T^{\mathrm{ad}} \times \mathbb{G}_m^{\mathrm{rot}} \times \mathbb{G}_m^{\mathrm{dil}}$  acts on  $P_0^F, L^F$ , and  $L_0^F$  with the trivial action of  $\mathbb{G}_m^{\mathrm{dil}}$ ; it also acts on the vector spaces  $\mathfrak{p}_*^F$  and  $\mathfrak{l}_*^F$  where  $\mathbb{G}_m^{\mathrm{dil}}$ acts by scaling. Therefore  $\mathbb{G}_m^{\flat}$  acts on  $P_0^F, L^F, \mathfrak{p}_*^F$ , and  $\mathfrak{l}_*^F$ . Note that the adjoint action of  $L^F$  on  $\mathfrak{l}^F$  is still  $\mathbb{G}_m^{\flat}$ -equivariant. Similarly, the group  $T^{\mathrm{ad}} \times \mathbb{G}_m^{\mathrm{rot}} \times \mathbb{G}_m^{\mathrm{dil}}$ acts on  $G_0$  and  $\mathfrak{g}_{\underline{\eta}}$  (with the action of  $\mathbb{G}_m^{\mathrm{rot}}$  defined in §4.1.5), hence  $\mathbb{G}_m^{\flat}$  also acts on  $G_0$  and  $\mathfrak{g}_{\eta}$ .

A direct calculation shows the following.

## 4.1.7. Lemma.

- (1) The torus  $\mathbb{G}_m^{\flat}$  acts trivially on  $G_{\underline{0}}$ ,  $\mathfrak{g}_{\eta}$ ,  $P_0^F$ , and  $L_0^F$ .
- (2) For any  $n \in \mathbb{Z}$ , the torus  $\mathbb{G}_m^{\flat}$  acts on  $\mathfrak{p}_n^F$  and  $\mathfrak{l}_n^F$  by weight  $n \eta$ .

4.1.8. Enhanced spiral induction. Let  $A \in \mathfrak{F}^{\xi}$ . The diagram (4.1) defining the spiral induction consists of  $\mathbb{G}_m^{\mathrm{rot}} \times \mathbb{G}_m^{\mathrm{dil}}$ -equivariant maps. Therefore we have an enhancement of the spiral induction  $\mathbf{I}_A$  into a  $G_{\underline{0},\Diamond} \times \mathbb{G}_m^{\mathrm{dil}} = (G_{\underline{0}} \rtimes \mathbb{G}_m^{\mathrm{rot}}) \times \mathbb{G}_m^{\mathrm{dil}}$ -equivariant object

$$\mathbf{J}_A = \alpha_!^A \beta^{A,*} \mathcal{L}^{A,\sharp} \in \mathcal{D}_{G_{0,\diamond} \times \mathbb{G}_m^{\mathrm{dil}}}(\mathfrak{g}_\eta).$$

# 4.2. Localization to $\mathbb{G}_m$ -fixed points.

4.2.1. Let X be a k-scheme of finite type with an action of  $\mathbb{G}_m$ . Let Y be another k-scheme of finite type and let  $f : X \to Y$  be a  $\mathbb{G}_m$ -invariant morphism. Let  $\overline{f} : [X/\mathbb{G}_m] \to Y$  be the induced map. Let  $i : X^{\mathbb{G}_m} \to X$  be the closed embedding of the fixed points, and let  $f_0 : X^{\mathbb{G}_m} \to Y$  be the restriction of f to the fixed point locus  $X^{\mathbb{G}_m}$ .

Let  $\mathcal{K} \in \mathcal{D}_{\mathbb{G}_m}(X)$  be a constructible  $\mathbb{G}_m$ -equivariant  $\overline{\mathbb{Q}}_{\ell}$ -complex on X, also viewed as a complex on  $[X/\mathbb{G}_m]$ . The direct image complex  $\overline{f}_*\mathcal{K}$  is bounded from below on Y. Let  $\mathcal{K}_0 = i^*\mathcal{K}$ .

4.2.2. The equivariant parameter  $v \in \mathrm{H}^{2}_{\mathbb{G}_{m}}(\mathrm{pt})$  induces a map  $v : \overline{f}_{*}\mathcal{K} \to \overline{f}_{*}\mathcal{K}[2]$ . Passing to perverse cohomology sheaves we get a map  $v : {}^{p}\mathrm{H}^{n}\overline{f}_{*}\mathcal{K} \to {}^{p}\mathrm{H}^{n+2}\overline{f}_{*}\mathcal{K}$ . Restricting to fixed points gives a map

$$i^*: \overline{f}_*\mathcal{K} \to f_{0,*}\mathcal{K}_0 \otimes \mathrm{H}^*_{\mathbb{G}_m}(\mathrm{pt}) = \bigoplus_{\ell \in \mathbb{Z}_{\geq 0}} f_{0,*}\mathcal{K}_0[-2\ell]v^\ell.$$

Passing to perverse cohomology sheaves we get a map

$${}^{p}\mathrm{H}^{n}(i^{*}):{}^{p}\mathrm{H}^{n}\overline{f}_{*}\mathcal{K}\to\bigoplus_{\ell\in\mathbb{Z}_{\geq0}}{}^{p}\mathrm{H}^{n-2\ell}f_{0,*}\mathcal{K}_{0}v^{\ell}.$$

Specializing v to a non-zero value  $v_0 \in \overline{\mathbb{Q}}_{\ell}^{\times}$ , we get

$${}^{p}\mathrm{H}^{n}(i^{*})_{v=v_{0}}:{}^{p}\mathrm{H}^{n}\overline{f}_{*}\mathcal{K}\to\bigoplus_{\ell\in\mathbb{Z}_{\geq0}}{}^{p}\mathrm{H}^{n-2\ell}f_{0,*}\mathcal{K}_{0}\subset\begin{cases} {}^{p}\mathrm{H}^{\mathrm{odd}}f_{0,*}\mathcal{K}_{0}, & n \text{ odd};\\ {}^{p}\mathrm{H}^{\mathrm{ev}}f_{0,*}\mathcal{K}_{0}, & n \text{ even}. \end{cases}$$

4.2.3. **Lemma.** Suppose f is proper. Let  $v_0 \in \overline{\mathbb{Q}}_{\ell}^{\times}$ . If  $n \in \mathbb{Z}$  is sufficiently large (depending on f and  $\mathcal{K}$ ), the map  $v : {}^{p}\mathrm{H}^{n}\overline{f}_{*}\mathcal{K} \to {}^{p}\mathrm{H}^{n+2}\overline{f}_{*}\mathcal{K}$  is an isomorphism, and the map  ${}^{p}\mathrm{H}^{n}(i^{*})_{v=v_0}$  gives an isomorphism

$${}^{p}\mathrm{H}^{n}\overline{f}_{*}\mathcal{K} \xrightarrow{\sim} \begin{cases} {}^{p}\mathrm{H}^{\mathrm{odd}}f_{0,*}\mathcal{K}_{0}, & n \text{ odd}; \\ {}^{p}\mathrm{H}^{\mathrm{ev}}f_{0,*}\mathcal{K}_{0}, & n \text{ even.} \end{cases}$$

Proof. Let  $U = X - X^{\mathbb{G}_m}$  and let  $\overline{g} : [U/\mathbb{G}_m] \to Y$  be the restriction of  $\overline{f}$ . Let  $\mathcal{K}_U$  be the restriction of  $\mathcal{K}$  to  $[U/\mathbb{G}_m]$ . Since  $\mathbb{G}_m$  has no fixed point on U, it acts on U with finite stabilizers, hence  $\overline{g}_!\mathcal{K}_U$  is a constructible complex on Y bounded both in cohomological degrees and perverse cohomological degrees. In particular, there exists  $N_1 \in \mathbb{Z}$  such that  ${}^p \mathrm{H}^n \overline{g}_! \mathcal{K}_U = 0$  for  $n \geq N_1$ . The exact triangle  $\overline{g}_! \mathcal{K}_U \to \overline{f}_* \mathcal{K} \to \overline{f}_{0,*} \mathcal{K}_0 \to$  (using that f is proper) gives a long exact sequence on the level perverse cohomology sheaves, which implies that for  $n \geq N_1$ , the restriction map  ${}^p \mathrm{H}^n(i^*)$  is an isomorphism.

On the other hand, the complex  $f_{0,*}\mathcal{K}_0$  also has bounded perverse cohomological degrees. Let  $N_2 \in \mathbb{Z}$  be such that  ${}^{p}\mathrm{H}^{n}f_{0,*}\mathcal{K}_0 = 0$  for  $n \geq N_2$ . Then for  $n \geq N_2$ , the perverse sheaf  $\bigoplus_{\ell \in \mathbb{Z}_{>0}} {}^{p}\mathrm{H}^{n-2\ell}f_{0,*}\mathcal{K}_0 v^{\ell}$  is equal to either  ${}^{p}\mathrm{H}^{\mathrm{odd}}f_{0,*}\mathcal{K}_0$  or

 ${}^{p}\mathrm{H}^{\mathrm{ev}}f_{0,*}\mathcal{K}_{0}$  according to the parity of n. In conclusion, when  $n \geq \max\{N_{1}, N_{2}\}$ , the map  ${}^{p}\mathrm{H}^{n}(i^{*})$  gives the desired isomorphism between  ${}^{p}\mathrm{H}^{n}\overline{f}_{*}\mathcal{K}$  and  ${}^{p}\mathrm{H}^{\mathrm{odd}}f_{0,*}\mathcal{K}_{0}$  or  ${}^{p}\mathrm{H}^{\mathrm{ev}}f_{0,*}\mathcal{K}_{0}$  according to the parity of n.

Multiplication by v gives a commutative diagram

For  $n \ge N_2$ , the right vertical map is obviously an isomorphism. Therefore for  $n \ge \max\{N_1, N_2\}$ , the left vertical map is also an isomorphism.

4.3. Action of the polynomial part. In this subsection we construct an action of the polynomial part

$$\mathbb{S}^{\xi}_{\eta/m} := \mathbb{Q}[u] \otimes \operatorname{Sym}(\mathbb{E}^{\xi,*}_{\diamondsuit})/(u+\eta/m,\delta-1)$$

of  $\mathbb{H}_{c,\eta/m}(W_{\mathrm{aff}}^{\xi})$  on  ${}^{p}\mathrm{HI}_{A}$  for each  $A \in \mathfrak{F}^{\xi}$ .

In the rest of the subsection, we fix  $A \in \mathfrak{F}^{\xi}$  and let  $\mathbb{E} = \text{Span}(A)$ . We identify  $\mathbb{E}^{\xi}_{\Diamond}$  with  $\mathbb{E}_{\Diamond}$ , hence  $\mathbb{S}^{\xi}_{\eta/m} \cong \mathbb{Q}[u] \otimes \text{Sym}(\mathbb{E}^{*}_{\Diamond})/(u + \eta/m, \delta - 1)$ .

4.3.1. We first construct an action of  $\mathbb{Q}[u] \otimes \operatorname{Sym}(\mathbb{E}^*_{\Diamond})$  on the enhanced spiral induction  $\mathbf{J}_A$  defined in §4.1.8.

Let  $\mathcal{C}^A$  be the cuspidal  $L^A$ -equivariant local system on the nilpotent orbit  $\mathcal{O}_{\mathfrak{l}^A}$ of  $\mathfrak{l}^A$  containing  $\mathring{\mathfrak{l}}^A_\eta$  whose restriction to  $\mathring{\mathfrak{l}}^A_\eta$  gives  $\mathcal{L}^A$  (this is well-defined by [7, Prop 4.2(c)]). Clearly  $\mathcal{C}^A$  is  $L^A_{\diamondsuit} \times \mathbb{G}^{\mathrm{dil}}_m$ -equivariant. We view the middle extension  $\mathcal{C}^{A,\sharp}$ as an object in  $\mathcal{D}_{L^A_{\diamondsuit} \times \mathbb{G}^{\mathrm{dil}}_m}(\mathfrak{l}^A)$ , therefore it carries an action of  $\mathrm{H}^*_{L^A_{\diamondsuit} \times \mathbb{G}^{\mathrm{dil}}_m}(\mathrm{pt})$ .

Let  $L^{A,\mathrm{ab}} = L^A / L^{A,\mathrm{der}}$ . Let  $L^{A,\mathrm{ab}}_{\Diamond} = L^A_{\Diamond} / L^{A,\mathrm{der}}$ ; then we have an exact sequence  $1 \to L^{A,\mathrm{ab}} \to L^{A,\mathrm{ab}}_{\Diamond} \to \mathbb{G}_m^{\mathrm{rot}} \to 1$ . Lemma 2.3.4 implies a canonical isomorphism

$$\mathbb{X}_*(L^{A,\mathrm{ab}}_{\diamondsuit})_{\mathbb{Q}} \cong \mathbb{X}_*(Z^0_{L^A_{\diamondsuit}})_{\mathbb{Q}} = \mathbb{E}_{\diamondsuit}.$$

Hence,  $\mathcal{C}^{A,\sharp}$  carries an action of

$$\mathrm{H}^{*}_{L^{A,\mathrm{ab}}_{\diamondsuit}\times\mathbb{G}^{\mathrm{dil}}_{m}}(\mathrm{pt})\cong\overline{\mathbb{Q}}_{\ell}[u]\otimes\mathrm{Sym}(\mathbb{E}^{*}_{\diamondsuit}\otimes\overline{\mathbb{Q}}_{\ell}).$$

Here we denote by u the canonical generator of  $\mathbb{X}^*(\mathbb{G}_m^{\mathrm{dil}})$ , which gives a basis for  $\mathrm{H}^2_{\mathbb{G}^{\mathrm{dil}}}(\mathrm{pt})$ .

Consider the  $\mathbb{G}_m^{\mathrm{rot}} \times \mathbb{G}_m^{\mathrm{dil}}$ -equivariant map

$$\eta^A: G_{\underline{0}} \overset{P_0^A}{\times} \mathfrak{p}^A_\eta \to [\mathfrak{l}^A_\eta/L_0^A] \to [\mathfrak{l}^A/L^A].$$

We have  $\mathbf{J}_A = \alpha_!^A \beta^{A,*} \mathcal{L}^{A,\sharp} \cong \alpha_!^A \eta^{A,*} \mathcal{C}^{A,\sharp} \in \mathcal{D}_{G_{\underline{0},\Diamond} \times \mathbb{G}_m^{\mathrm{dil}}}(\mathfrak{g}_{\underline{\eta}})$ . Therefore,  $\mathbf{J}_A$  also carries an action of  $\mathbb{Q}[u] \otimes \mathrm{Sym}(\mathbb{E}_{\Diamond}^*)$ . By construction, u acts on  $\mathbf{J}_A$  by cupping with the equivariant parameter of  $\mathbb{G}_m^{\mathrm{dil}}$ , and  $\delta/e \in \mathbb{E}_{\Diamond}^*$  acts on  $\mathbf{J}_A$  by cupping with the equivariant parameter of  $\mathbb{G}_m^{\mathrm{rot}}$ .

4.3.2. Now we construct an action of  $\mathbb{S}_{\eta/m}^{\xi}$  on  ${}^{p}\mathrm{HI}_{A}$ .

Choose  $r \geq 1$  such that  $rx \in \mathbb{X}_*(T)$ . We denote by  $\widetilde{\mathbb{G}}_m^{\flat}$  the one-dimensional torus equipped with the map

(4.2) 
$$\widetilde{s}: \widetilde{\mathbb{G}}_{m}^{\flat} \to T \times \mathbb{G}_{m}^{\mathrm{rot}} \times \mathbb{G}_{m}^{\mathrm{dil}}$$
$$t \mapsto ((rx)(t), t^{mr/e}, t^{-r\eta}),$$

i.e.,  $\tilde{s}$  is the composition of the embedding s of  $\mathbb{G}_m^{\flat}$  (see §4.1.6) with the rth power map. We have a homomorphism

 $G_{\underline{0}} \times \widetilde{\mathbb{G}}_m^\flat \to G_{\underline{0},\diamondsuit} \times \mathbb{G}_m^{\mathrm{dil}}$ 

which is the natural embedding  $G_{\underline{0}} \hookrightarrow G_{\underline{0},\Diamond}$  on  $G_{\underline{0}}$  and the homomorphism  $\widetilde{\mathbb{G}}_{m}^{\flat} \xrightarrow{\widetilde{s}} T \times \mathbb{G}_{m}^{\mathrm{rot}} \times \mathbb{G}_{m}^{\mathrm{dil}} \subset G_{\underline{0},\Diamond} \times \mathbb{G}_{m}^{\mathrm{dil}}$ . This map is well-defined because the images of  $G_{\underline{0}}$  and  $\widetilde{\mathbb{G}}_{m}^{\flat}$  in  $G_{\underline{0},\Diamond} \times \mathbb{G}_{m}^{\mathrm{dil}}$  commute with each other by Lemma 4.1.7(1).

We have the forgetful functor for equivariant derived categories

$$\mathcal{D}_{G_{\underline{0}},\diamond} \times \mathbb{G}_m^{\mathrm{dil}}(\mathfrak{g}_{\underline{\eta}}) \to \mathcal{D}_{G_0} \times \widetilde{\mathbb{G}}_m^{\flat}(\mathfrak{g}_{\underline{\eta}}).$$

Let  $\mathbf{J}_{A}^{\flat} \in \mathcal{D}_{G_{\underline{0}} \times \widetilde{\mathbb{G}}_{m}^{\flat}}(\mathfrak{g}_{\underline{\eta}})$  be the image of  $\mathbf{J}_{A}$ . Let

$$\mathbf{S}_v = (\mathbb{Q}[u] \otimes \operatorname{Sym}(\mathbb{E}^*_{\diamondsuit}))/(u + \eta \delta/m).$$

We denote the image of -u and  $\eta\delta/m$  in  $\mathbf{S}_v$  by v. Since  $\mathbb{Q}[u] \otimes \operatorname{Sym}(\mathbb{E}^*_{\Diamond})$  acts on  $\mathbf{J}_A$ , it follows that  $\mathbf{S}_v$  acts on  $\mathbf{J}_A^{\flat}$  with v acting as a non-zero element in  $\operatorname{H}^2_{\widetilde{\mathbb{G}}_m^{\flat}}(\operatorname{pt})$ . Consider the morphism

$$\omega: [\mathfrak{g}_{\underline{\eta}}/(G_{\underline{0}}\times\widetilde{\mathbb{G}}_m^{\flat})] \cong [\mathfrak{g}_{\underline{\eta}}/G_{\underline{0}}] \times [\mathrm{pt}/\widetilde{\mathbb{G}}_m^{\flat}] \to [\mathfrak{g}_{\underline{\eta}}/G_{\underline{0}}].$$

The first isomorphism follows from the fact that  $\widetilde{\mathbb{G}}_m^{\flat}$  acts trivially on  $\mathfrak{g}_{\underline{\eta}}$  (Lemma 4.1.7(1)). We then have the direct image functor

$$\omega_*: \mathcal{D}_{G_{\underline{0}} \times \widetilde{\mathbb{G}}_m^\flat}(\mathfrak{g}_{\underline{\eta}}) \to \mathcal{D}_{G_{\underline{0}}}(\mathfrak{g}_{\underline{\eta}})$$

and its left adjoint  $\omega^*$ . By Lemma 4.1.7, the torus  $\mathbb{G}_m^{\flat}$  acts trivially on  $\mathfrak{g}_{\underline{\eta}}, G_{\underline{0}} \overset{P_0^A}{\times} \mathfrak{p}_{\eta}^A$ and  $[\mathfrak{l}_n^A/L_0^A]$ , therefore  $\mathbf{J}_{\underline{\lambda}}^{\flat} \cong \omega^* \mathbf{I}_A$ . Hence

$$\omega_* \mathbf{J}_A^{\flat} \cong \omega_* \omega^* \mathbf{I}_A \cong \mathbf{I}_A \otimes \mathrm{H}^*_{\widetilde{\mathbb{G}}_m^{\flat}}(\mathrm{pt}) \cong \mathbf{I}_A \otimes \overline{\mathbb{Q}}_{\ell}[v].$$

Therefore, the algebra  $\mathbf{S}_v$  acts on  $\mathbf{I}_A \otimes \overline{\mathbb{Q}}_{\ell}[v]$  in a  $\mathbb{Q}[v]$ -linear way. The degree 2 element  $v = -u = \eta \delta/m \in \mathbf{S}_v$  induces a map

$${}^{p}\mathrm{H}^{n}(v):{}^{p}\mathrm{H}^{n}(\mathbf{I}_{A}\otimes\overline{\mathbb{Q}}_{\ell}[v])\to{}^{p}\mathrm{H}^{n+2}(\mathbf{I}_{A}\otimes\overline{\mathbb{Q}}_{\ell}[v]).$$

For n even and sufficiently large, the map defined by specializing v to  $\eta/m$ 

$${}^{p}\mathrm{H}^{n}(\mathbf{I}_{A}\otimes\overline{\mathbb{Q}}_{\ell}[v]) = \bigoplus_{i\geq 0}{}^{p}\mathrm{H}^{n-2i}\mathbf{I}_{A}\cdot v^{i} \xrightarrow{v\mapsto \eta/m} \bigoplus_{i\geq 0}{}^{p}\mathrm{H}^{n-2i}\mathbf{I}_{A}\subset {}^{p}\mathrm{H}^{\mathrm{ev}}\mathbf{I}_{A}$$

is an isomorphism. Therefore  ${}^{p}\mathbf{H}^{n}(v)$  defines an endomorphism of  ${}^{p}\mathbf{H}^{\mathrm{ev}}\mathbf{I}_{A}$ , under which  $v = -u = \eta\delta/m$  acts as  $\eta/m$ . Similarly, for n odd and sufficiently large,  ${}^{p}\mathbf{H}^{n}(v)$  defines an endomorphism of  ${}^{p}\mathbf{H}^{\mathrm{odd}}\mathbf{I}_{A}$  under which  $v = -u = \eta\delta/m$  acts as  $\eta/m$ . We have thus constructed an action of  $\mathbf{S}_{v}/(v - \eta/m) \cong \mathbb{S}_{\eta/m}^{\xi}$  on  ${}^{p}\mathbf{H}\mathbf{I}_{A}$ preserving its  $\mathbb{Z}/2\mathbb{Z}$ -grading.

## This is a free offprint provided to the author by the publisher. Copyright restrictions may apply.

110

The above construction uses an auxiliary number r (to define a lifting of x to a cocharacter of T), but the resulting action is clearly independent of this choice. Moreover, if  $A_1$  is in the same  $W_{x/m}$ -orbit of A, the action of  $\mathbb{S}^{\xi}_{\eta/m}$  on  $\mathbf{I}_{A_1}$  and on  $\mathbf{I}_A$  are the same, under the canonical isomorphism  $\mathbf{I}_{A_1} \cong \mathbf{I}_A$ . Therefore, for  $[A] \in W_{x/m} \setminus \mathfrak{F}^{\xi}$ , we have a well-defined action of  $\mathbb{S}^{\xi}_{n/m}$  on  $\mathbf{I}_{[A]}$ .

4.3.3. Spectra of the polynomial ring action. Recall  $A \in \mathfrak{F}^{\xi}$  and  $\mathbb{E} = \operatorname{Span}(A)$ . The orthogonal projection of x/m onto  $\mathbb{E}$  gives a point  $\langle x/m \rangle_{\mathbb{E}} \in \mathbb{E}$ . Under the canonical isomorphism  $\iota_A : \mathbb{E} \cong \mathbb{E}^{\xi}$  (depending on A), we denote the image of  $\langle x/m \rangle_{\mathbb{E}}$  by  $\langle x/m \rangle_A$ . Note that  $\langle x/m \rangle_A \in \mathbb{E}^{\xi}$  depends only on the  $W_{x/m}$ -orbit of A, therefore we may denote it by  $\langle x/m \rangle_{[A]}$ .

We have constructed an action of  $\mathbb{S}^{\xi}_{\eta/m}$  on  ${}^{p}\mathrm{HI}_{[A]}$ . Then the subscheme Spec  $\mathbb{S}^{\xi}_{\eta/m}$  $\subset \mathbb{A}^{1}_{u} \times \mathbb{E}^{\xi}_{\Diamond}$  is defined by the equations  $u = -\eta/m$  and  $\delta = 1$ . Therefore the projection Spec  $\mathbb{S}^{\xi}_{\eta/m} \to \mathbb{E}^{\xi}_{\Diamond}$  gives an isomorphism of affine spaces over  $\mathbb{Q}$ 

Spec 
$$\mathbb{S}^{\xi}_{\eta/m} \xrightarrow{\sim} \mathbb{E}^{\xi}$$
.

4.3.4. **Proposition.** The action of  $\mathbb{S}^{\xi}_{\eta/m}$  on  ${}^{p}\mathrm{HI}_{[A]}$  has a single eigenvalue given by  $\langle x/m \rangle_{[A]} \in \mathbb{E}^{\xi} \cong \operatorname{Spec} \mathbb{S}^{\xi}_{n/m}$ .

Proof. We use the notation from §4.3.2. We write  $X^A = G_{\underline{0}} \overset{P_0^A}{\times} \mathfrak{p}_{\eta}^A$  as X, and let  $\mathcal{X} = [G_{\underline{0}} \backslash X]$ . Let  $p: L_{\Diamond}^{A,\mathrm{ab}} \times \mathbb{G}_m^{\mathrm{dil}} \to \mathbb{G}_m^{\mathrm{rot}} \times \mathbb{G}_m^{\mathrm{dil}}$  be the projection, let  $\mu: \mathbb{G}_m = \mathbb{G}_m^{\heartsuit} \to \mathbb{G}_m^{\mathrm{rot}} \times \mathbb{G}_m^{\mathrm{dil}}$  be given by  $t \mapsto (t^{rm/e}, t^{-r\eta})$ , and let  $H \to L_{\Diamond}^{A,\mathrm{ab}} \times \mathbb{G}_m^{\mathrm{dil}}$  be the base change of p along  $\mu$ . Then we have an exact sequence  $1 \to L^{A,\mathrm{ab}} \to H \to \mathbb{G}_m^{\heartsuit} \to 1$ . The map  $\rho: \mathbb{G}_p^{\circlearrowright} \xrightarrow{\tilde{s}} T \times \mathbb{G}_m^{\mathrm{rot}} \times \mathbb{G}_m^{\mathrm{dil}} \to L_{\Diamond}^{A,\mathrm{ab}} \times \mathbb{G}_m^{\mathrm{dil}}$  (for  $\tilde{s}$  see (4.2)) has image in H and gives a section to the projection  $H \to \mathbb{G}_m^{\heartsuit}$ . Therefore we may identify H with  $L^{A,\mathrm{ab}} \times \mathbb{G}_m^{\backsim}$ . The projection  $H \to L_{\Diamond}^{A,\mathrm{ab}}$  is an isogeny, which induces an isomorphism  $\mathbb{X}_*(H)_{\mathbb{Q}} \cong \mathbb{E}_{\Diamond}$ .

Let  $\pi: \widetilde{X} \to X$  be the  $L^{A,ab}$ -torsor corresponding to the map  $\eta: X \to [\mathfrak{l}^A/L^A] \to [\mathfrak{pt}/L^{A,ab}]$ . Then the  $G_{\underline{0}}$ -action lifts to  $\widetilde{X}$  canonically, and we form the stack  $\widetilde{\mathcal{X}} = [G_{\underline{0}} \setminus \widetilde{X}]$ . Also H acts on  $\widetilde{X}$  extending the  $L^{A,ab}$ -action, hence  $\widetilde{\mathbb{G}}_m^{\flat}$  also acts on  $\widetilde{X}$  via  $\rho: \widetilde{\mathbb{G}}_m^{\flat} \to H$ . A direct calculation shows that the  $\widetilde{\mathbb{G}}_m^{\flat}$ -action via  $\rho$  on  $\widetilde{X}$  coincides with the action  $\lambda$  given by  $\widetilde{\mathbb{G}}_m^{\flat} \xrightarrow{r_X} T \to G_{\underline{0}}$  and the action of  $G_{\underline{0}}$  on  $\widetilde{X}$ . In particular, the action of  $\rho(\widetilde{\mathbb{G}}_m^{\flat})$  on  $\widetilde{\mathcal{X}}$  is trivial. Let  $\chi: L^{A,ab} \to \mathbb{G}_m$  be a character, and we extend it to H by requiring it to

Let  $\chi : L^{A, \mathrm{ab}} \to \mathbb{G}_m$  be a character, and we extend it to H by requiring it to be trivial on  $\widetilde{\mathbb{G}}_m^{\flat}$ . Let  $X_{\chi} = \widetilde{X} \xrightarrow{L^{A, \mathrm{ab}}, \chi} \mathbb{G}_m$  be the  $\mathbb{G}_m$ -torsor over X using  $\chi_1$ . Let  $\overline{X}_{\chi} = \widetilde{X} \xrightarrow{H} \mathbb{G}_m$  be the  $\mathbb{G}_m$ -torsor over  $[\widetilde{X}/H] = [X/\rho(\widetilde{\mathbb{G}}_m^{\flat})]$  given by  $\chi$ . Then we have  $\overline{X}_{\chi} \cong [X_{\chi}/\rho(\widetilde{\mathbb{G}}_m^{\flat})]$ . The action of  $\rho(\widetilde{\mathbb{G}}_m^{\flat})$  on  $X_{\chi}$  is the same as the action  $\lambda$ given as the restriction of the left  $G_{\underline{0}}$ -action.

Let  $\mathcal{X}_{\chi} = [G_{\underline{0}} \setminus X_{\chi}]$  and  $\overline{\mathcal{X}}_{\chi} = [G_{\underline{0}} \setminus \overline{X}_{\chi}]$  be the  $\mathbb{G}_m$ -torsors over  $\mathcal{X}$  and  $\mathcal{X}/\rho(\widetilde{\mathbb{G}}_m^{\flat})$ , respectively. By the above discussion, the  $\rho$ -actions of  $\widetilde{\mathbb{G}}_m^{\flat}$  on  $\mathcal{X}_{\chi}$  and  $\mathcal{X}$  are trivial because they are the same as the  $\lambda$ -actions (hence part of the  $G_{\underline{0}}$ -actions on  $X_{\chi}$ and X) which we quotient out. Therefore  $\overline{\mathcal{X}}_{\chi} \cong [\mathcal{X}_{\chi}/\widetilde{\mathbb{G}}_m^{\flat}] \cong \mathcal{X}_{\chi} \times [\mathrm{pt}/\widetilde{\mathbb{G}}_m^{\flat}]$ .

The  $\mathbb{G}_m$ -torsor  $\mathcal{X}_{\chi} \to \mathcal{X}$  defines a Chern class  $c_1(\chi) \in \mathrm{H}^2_{G_{\underline{0}}}(X, \beta^{A,*}\mathcal{L}^{A,\sharp}) = \mathrm{H}^2(\mathcal{X}, \beta^{A,*}\mathcal{L}^{A,\sharp})$ . Similarly the  $\mathbb{G}_m$ -torsor  $\overline{\mathcal{X}}_{\chi} \to \overline{\mathcal{X}}$  defines a Chern class  $c_1^{\widetilde{\mathbb{G}}_m^{\flat}}(\chi) \in \mathrm{H}^2_{G_{\underline{0}} \times \widetilde{\mathbb{G}}_m^{\flat}}(X, \beta^{A,*}\mathcal{L}^{A,\sharp}) = \mathrm{H}^2_{\widetilde{\mathbb{G}}_m^{\flat}}(\mathcal{X}, \beta^{A,*}\mathcal{L}^{A,\sharp})$ . By the above discussion, we have

(4.3) 
$$c_1^{\widetilde{\mathbb{G}}_{2}^{\mathbb{b}}}(\chi) = c_1(\chi) \otimes 1$$

under the factorization  $\mathrm{H}^*_{\widetilde{\mathbb{G}}^{\flat}_m}(\mathcal{X}, \beta^{A,*}\mathcal{L}^{A,\sharp}) \cong \mathrm{H}^*(\mathcal{X}, \beta^{A,*}\mathcal{L}^{A,\sharp}) \otimes \overline{\mathbb{Q}}_{\ell}[v]$  (where  $v \in \mathrm{H}^2_{\widetilde{\mathbb{G}}^{\flat}}$  (pt) is an equivariant parameter).

By construction, the action of  $\chi \in \mathbf{S}_v$  on  ${}^p\mathbf{H}\mathbf{I}_A$  is induced from the map  $\mathbf{I}_A \otimes \overline{\mathbb{Q}}_\ell[v] \to \mathbf{I}_A[2] \otimes \overline{\mathbb{Q}}_\ell[v]$  given by  $\bigcup c_1^{\widetilde{\mathbb{Q}}_m^k}(\chi)$ . By (4.3), this map is  $\bigcup c_1(\chi)$  on  $\mathbf{I}_A$  and the identity on  $\overline{\mathbb{Q}}_\ell[v]$ . Since  $\bigcup c_1(\chi)$  is a nilpotent operator on  $\mathbf{I}_A$ , the action of  $\chi$  on  ${}^p\mathbf{H}\mathbf{I}_A$  is nilpotent. This being true for all  $\chi \in \mathbb{X}^*(L^{A,\mathrm{ab}})$ , it implies that the eigenvalues of the  $\mathbb{S}_{\eta/m}$ -action on  ${}^p\mathbf{H}\mathbf{I}_A$  are in the image of the map  $\ell : \mathbb{X}_*(\widetilde{\mathbb{G}}_m^b)_{\mathbb{Q}} \xrightarrow{\rho^*} \mathbb{X}_*(H)_{\mathbb{Q}} \cong \mathbb{E}_{\Diamond}$ . However, the image of  $\ell$  intersects  $\mathbb{E}$  in the unique point  $\langle x/m \rangle_{\mathbb{E}}$ , therefore the eigenvalue of the  $\mathbb{S}_{\eta/m}$ -action on  ${}^p\mathbf{H}\mathbf{I}_A$  is given by  $\langle x/m \rangle_{\mathbb{E}}$  if we identify  $\mathbb{S}_{\eta/m}$  with  $\mathcal{O}(\mathbb{E})$ . In other words, under the canonical isomorphism  $\mathbb{E} \cong \mathbb{E}^{\xi}$ , the eigenvalue of the  $\mathbb{S}_{\eta/m}$ -action on  ${}^p\mathbf{H}\mathbf{I}_A$  is given by  $\langle x/m \rangle_A$ .

### 4.4. Recollections from [5].

4.4.1. Let  $J \subset I^{\xi}$  be a proper subset of simple reflections. Let  $W_J \subset W^{\xi}_{\text{aff}}$  be the finite subgroup generated by J. For any  $A \in \mathfrak{F}^{\xi}$ , we may identify  $W^{\xi}_{\text{aff}}$  with  $W^{A}_{\text{aff}} = (W_{\text{aff}}(\mathbb{E}), I(A))$  for  $\mathbb{E} = \text{Span}(A)$  (see §2.4.9); then J gives a unique facet  $F^{A}_{J} \subset \partial A$  so that  $W_J$  is the stabilizer of  $F^{A}_{J}$  under  $W_{\text{aff}}(\mathbb{E})$ . Let  $\mathfrak{F}^{\xi}_{J} \subset \mathfrak{F}$  be the facets of the form  $F^{A}_{J}$  for various  $A \in \mathfrak{F}^{\xi}$ . By definition we have a surjective map

$$\Pi_J:\mathfrak{F}^\xi\to\mathfrak{F}^\xi_J$$

sending A to the facet  $F_J^A \subset \partial A$ . This map is equivariant under  $W_{x/m}$ , and it induces a maps

$$\underline{\Pi}_J: W_{x/m} \backslash \mathfrak{F}^{\xi} \to W_{x/m} \backslash \mathfrak{F}^{\xi}_J.$$

4.4.2. We define a subalgebra  $\mathbb{H}_{c}(W_{J})$  of  $\mathbb{H}_{c}(W_{\text{aff}}^{\xi})$  by taking the subspace

$$\mathbb{H}_c(W_J) = \mathbb{Q}[u] \otimes \operatorname{Sym}(\mathbb{E}^{\xi,*}_{\diamond}) \otimes \mathbb{Q}[W_J]$$

and the same relations, except that the commutation relation (2.5) is only required to hold for  $i \in J$ . For  $\nu \in \mathbb{Q}$ , let

$$\mathbb{H}_{c,\nu}(W_J) = \mathbb{H}_c(W_J)/(u+\nu,\delta-1).$$

4.4.3. Fix  $F \in \mathfrak{F}_{J}^{\xi}$ . Let  $\mathfrak{p}_{*} = \mathfrak{p}_{*}^{F}$ ,  $P_{0} = e^{\mathfrak{p}_{0}}$  and  $(L, L_{0}, \mathfrak{l}, \mathfrak{l}_{*}) = (L^{F}, L_{0}^{F}, \mathfrak{l}^{F}, \mathfrak{l}^{F}_{*})$ . We may identify  $W_{F} = \operatorname{Stab}_{W_{\operatorname{aff}}}(F)$  with the Weyl group  $W_{L}$  of L.

Let  $\Xi_F \subset \mathfrak{F}$  be the set of facets A in the same  $W_{\text{aff}}$ -orbit as those in  $\mathfrak{F}^{\xi}$  such that  $F \subset \partial A$ . Then  $\Xi_F$  carries a transitive action of  $W_F = W_L$ . Any  $A \in \Xi_F$  gives a parabolic subgroup  $Q^A \subset L$  whose Lie algebra is  $\mathfrak{q}^A = \bigoplus_{n \in \mathbb{Z}} \mathfrak{q}_n^A$  where  $\mathfrak{q}_n^A = \mathfrak{p}_n^A \cap \mathfrak{l}$ . Then  $L^A$  is a Levi subgroup of  $Q^A$ . For different  $A \in \Xi_F$ , the parabolics  $Q^A$  of L are in the same L-conjugacy class, therefore the quotient stacks  $[\mathfrak{l}^A/L^A]$  are *canonically* identified for various  $A \in \Xi_F$ . We denote these identical stacks by  $[\mathfrak{m}/M]$ .

The Weyl group  $W_J$  can be identified with the relative Weyl group  $W_F^A := N_{W_F}(W_A)/W_A$ .

4.4.4. Choose  $A \in \Xi_F$  and define  $\dot{\mathfrak{l}} = L \overset{Q^A}{\times} \mathfrak{q}^A$ . Note that  $\dot{\mathfrak{l}}$  is the partial Grothendieck alteration for  $\mathfrak{l}$  of type  $Q^A$ , and therefore is independent of the choice of  $A \in \Xi_F$ . We have a diagram (which is independent of the choice of  $A \in \Xi_F$ )

$$\mathfrak{l} \xleftarrow{a} \mathfrak{i} \xrightarrow{b} [\mathfrak{m}/M] ,$$

where a is the partial Grothendieck alteration and b is the composition of natural maps  $\dot{\mathfrak{l}} \to [\mathfrak{q}^A/Q^A] \to [\mathfrak{l}^A/L^A] \cong [\mathfrak{m}/M]$ . Let  $\mathcal{L}_M$  be the cuspidal *M*-equivariant local system on the nilpotent orbit  $\mathcal{O}$  of  $\mathfrak{m}$  whose restriction to  $\mathring{\mathfrak{m}}_{\eta}$  is  $\mathcal{L}$ .

The one-dimensional torus  $\mathbb{G}_m^{\text{dil}}$  acts on  $\mathfrak{l}, \mathfrak{q}$  and  $\mathfrak{m}$  by dilation, making the maps a and  $b \ \mathbb{G}_m^{\text{dil}}$ -equivariant. Recall the action of  $\mathbb{G}_m^{\text{rot}}$  on L, Q, and  $L^A$  introduced in §2.3, and the notation  $L_{\diamondsuit} = L \rtimes \mathbb{G}_m^{\text{rot}}$ , etc. By viewing  $L_{\diamondsuit}^A$  as Levi subgroups of  $L_{\diamondsuit}$ , we see that the canonical isomorphisms between  $[\mathfrak{l}^A/L^A]$  for  $A \in \Xi_F$  induce canonical isomorphisms between  $[\mathfrak{l}^A/L_{\diamondsuit}^A]$ , and we denote these identical stacks by  $[\mathfrak{m}/M_{\diamondsuit}]$ . The maps a and b induce maps of stacks

$$[\mathfrak{l}/(L_{\diamondsuit} \times \mathbb{G}_m^{\mathrm{dil}})] \xleftarrow{a} [\dot{\mathfrak{l}}/(L_{\diamondsuit} \times \mathbb{G}_m^{\mathrm{dil}})] \xrightarrow{b} [\mathfrak{m}/(M_{\diamondsuit} \times \mathbb{G}_m^{\mathrm{dil}})] .$$

We then have the parabolic induction

$$\operatorname{ind}^{\mathfrak{l}}_{\mathfrak{m}}(\mathcal{L}_{M}^{\sharp}) := a_{\mathfrak{l}} b^{*} \mathcal{L}_{M}^{\sharp} \in \mathcal{D}_{L_{\diamond} \times \mathbb{G}_{m}^{\operatorname{dil}}}(\mathfrak{l}).$$

Here we write  $\operatorname{ind}_{\mathfrak{m}}^{\mathfrak{l}}$  instead of  $\operatorname{ind}_{\mathfrak{q}^{A}}^{\mathfrak{l}}$  to indicate that it is independent of the choice of  $A \in \Xi_{F}$ .

Let  $\dot{\mathfrak{l}}_{\mathcal{O}}$  be the preimage of  $\mathcal{O}$  under the projection  $\dot{\mathfrak{l}} \to [\mathfrak{m}/M]$ . Let  $\mathring{a} : \dot{\mathfrak{l}}_{\mathcal{O}} \to \mathfrak{l}$ and  $\mathring{b} : \dot{\mathfrak{l}}_{\mathcal{O}} \to [\mathcal{O}/M]$  be the restrictions of a and b. Since the local system  $\mathcal{L}_M$  is clean and a is proper, we have

(4.4) 
$$\operatorname{ind}_{\mathfrak{m}}^{\mathfrak{l}}(\mathcal{L}_{M}^{\sharp}) \cong \mathring{a}_{!} \mathring{b}^{*} \mathcal{L}_{M} \cong \mathring{a}_{*} \mathring{b}^{*} \mathcal{L}_{M} \in \mathcal{D}_{L_{\Diamond} \times \mathbb{G}_{m}^{\mathrm{dil}}}(\mathfrak{l}).$$

We recall the following theorem from [5].

4.4.5. Theorem ([5]). There is a natural action of  $\mathbb{H}_{c}(W_{J})$  on

$$\operatorname{ind}_{\mathfrak{m}}^{\mathfrak{l}}(\mathcal{L}_{M}^{\sharp}) \in \mathcal{D}_{L_{\diamondsuit} \times \mathbb{G}_{m}^{\operatorname{dil}}}(\mathfrak{l}).$$

*Proof.* Let  $\ddot{\iota}_{\mathcal{O}} = \dot{\iota}_{\mathcal{O}} \times_{\mathfrak{l}} \dot{\iota}_{\mathcal{O}}$  with two projections  $\mathrm{pr}_1, \mathrm{pr}_2 : \ddot{\iota}_{\mathcal{O}} \to \dot{\iota}_{\mathcal{O}}$ . Let  $\ddot{\mathcal{L}}_M = \mathrm{pr}_1^* \dot{b}^* \mathcal{L}_M \otimes \mathrm{pr}_2^* \dot{b}^* \mathcal{L}_M^*$  is the dual local system of  $\mathcal{L}_M$ ), a local system on  $\ddot{\iota}_{\mathcal{O}}$ .

Let  $N = \dim \mathfrak{l}_{\mathcal{O}}$ . By (4.4) and proper base change, we have a graded isomorphism

$$(4.5) \qquad \operatorname{Ext}_{L_{\diamond}\times\mathbb{G}_{m}^{\operatorname{dil}}}^{*}(\operatorname{ind}_{\mathfrak{m}}^{\mathfrak{l}}(\mathcal{L}_{M}^{\sharp}), \operatorname{ind}_{\mathfrak{m}}^{\mathfrak{l}}(\mathcal{L}_{M}^{\sharp})) \cong \operatorname{Ext}_{L_{\diamond}\times\mathbb{G}_{m}^{\operatorname{dil}}}^{*}(\mathring{a}_{!}\dot{b}^{*}\mathcal{L}_{M}, \mathring{a}_{*}\dot{b}^{*}\mathcal{L}_{M}) \\ \cong \operatorname{Ext}_{L_{\diamond}\times\mathbb{G}_{m}^{\operatorname{dil}}}^{*}(\mathring{b}^{*}\mathcal{L}_{M}, \operatorname{pr}_{2,*}\operatorname{pr}_{1}^{!}\dot{b}^{*}\mathcal{L}_{M}) \cong \operatorname{H}_{2N-*}^{L_{\diamond}\times\mathbb{G}_{m}^{\operatorname{dil}}}(\ddot{\mathfrak{l}}_{\mathcal{O}}, \ddot{\mathcal{L}}_{M}).$$

Under (4.5),  $\mathrm{H}_{2N-*}^{L_{\Diamond} \times \mathbb{G}_{m}^{\mathrm{dil}}}(\ddot{\mathfrak{l}}_{\mathcal{O}}, \ddot{\mathcal{L}}_{M})$  carries a graded ring structure induced from the ring structure of  $\mathrm{Ext}_{L_{\Diamond} \times \mathbb{G}_{m}^{\mathrm{dil}}}^{*}(\mathrm{ind}_{\mathfrak{m}}^{\mathfrak{l}}(\mathcal{L}_{M}^{\sharp}), \mathrm{ind}_{\mathfrak{m}}^{\mathfrak{l}}(\mathcal{L}_{M}^{\sharp}))$ . We have a diagonal embedding  $\Delta : \dot{\mathfrak{l}}_{\mathcal{O}} \hookrightarrow \ddot{\mathfrak{l}}_{\mathcal{O}}$  and a canonical copy of the trivial local system inside  $\Delta^{*}\ddot{\mathcal{L}}_{M} \cong b^{*}\mathcal{L}_{M} \otimes b^{*}\mathcal{L}_{M}^{*}$ . Then the unit element **1** in the ring structure of  $\mathrm{H}_{2N-*}^{L_{\Diamond} \times \mathbb{G}_{m}^{\mathrm{dil}}}(\ddot{\mathfrak{l}}_{\mathcal{O}}, \ddot{\mathcal{L}}_{M})$  is given by the image of the fundamental class of  $\dot{\mathfrak{l}}_{\mathcal{O}}$  under the map

$$\mathrm{H}_{2N}^{L_{\Diamond}\times\mathbb{G}_{m}^{\mathrm{dil}}}(\dot{\mathfrak{l}}_{\mathcal{O}},\overline{\mathbb{Q}}_{\ell})\hookrightarrow\mathrm{H}_{2N}^{L_{\Diamond}\times\mathbb{G}_{m}^{\mathrm{dil}}}(\dot{\mathfrak{l}}_{\mathcal{O}},\Delta^{*}\ddot{\mathcal{L}}_{M})\xrightarrow{\Delta_{*}}\mathrm{H}_{2N}^{L_{\Diamond}\times\mathbb{G}_{m}^{\mathrm{dil}}}(\ddot{\mathfrak{l}}_{\mathcal{O}},\ddot{\mathcal{L}}_{M}).$$

This is consistent with the definition of 1 in  $[5, \S 6.1]$ .

For  $A \in \Xi_F$ , the group  $L^A_{\Diamond}$  is a Levi subgroup of  $L_{\Diamond}$  carrying a cuspidal local system  $\mathcal{C}_A$  which corresponds to  $\mathcal{L}_M$ . Moreover,  $W_J \cong W^A_F$  is the Weyl group of  $L_{\Diamond}$  relative to  $L^A_{\Diamond}$ . A graded affine Hecke algebra  $\mathbf{H} = \mathbf{H}_c(W_J)$  was introduced in [5, §0.1] for the triple  $(L_{\Diamond}, L^A_{\Diamond}, \mathcal{L}_M)$ . It takes the form  $\mathbf{H}_c(W_J) = \mathbb{Q}[u] \otimes \operatorname{Sym}(\mathbb{X}^*(Z^0_{L^A_{\Diamond}})_{\mathbb{Q}}) \otimes \mathbb{Q}[W_J]$ . By Lemma 2.3.4, we have a canonical  $W_J$ equivariant isomorphism  $\mathbb{X}^*(Z^0_{L^A_{\Diamond}})_{\mathbb{Q}} \cong \mathbb{E}^{\xi,*}_{\Diamond}$ . Moreover, our definitions of the coroots  $\alpha_j$  and the numbers  $c_j$  for  $j \in J$  are the same as their counterparts in [5]: we simply copied the definitions in loc. cit. We conclude that there is a canonical isomorphism  $\mathbf{H}_c(W_J) \cong \mathbb{H}_c(W_J)$ .

In [5, Corollary 6.4], a graded action of  $\mathbb{H}_c(W_J)$  on  $\mathrm{H}_{2N-*}^{L_{\diamond} \times \mathbb{G}_m^{\mathrm{dil}}}(\ddot{\mathfrak{l}}_{\mathcal{O}}, \ddot{\mathcal{L}}_M)$  was constructed (we use the action that was denoted by  $\Delta$  in [5]) and it was proved there that  $\mathrm{H}_{2N-*}^{L_{\diamond} \times \mathbb{G}_m^{\mathrm{dil}}}(\ddot{\mathfrak{l}}_{\mathcal{O}}, \ddot{\mathcal{L}}_M)$  is a free  $\mathbb{H}_c(W_J)$ -module of rank one with basis given by **1**. Therefore we have a graded isomorphism

$$\mathbb{H}_{c}(W_{J}) \cong \mathrm{H}_{2N-*}^{L_{\Diamond} \times \mathbb{G}_{m}^{\mathrm{dil}}}(\ddot{\mathfrak{l}}_{\mathcal{O}}, \ddot{\mathcal{L}}_{M})$$

sending **1** to **1**. It is also easy to check that the above isomorphism is indeed a ring isomorphism. Since  $\mathrm{H}_{2N-*}^{L_{\diamond} \times \mathbb{G}_{m}^{\mathrm{dil}}}(\ddot{\mathfrak{l}}_{\mathcal{O}}, \ddot{\mathcal{L}}_{M})$  acts on  $\mathrm{ind}_{\mathfrak{m}}^{\mathfrak{l}}(\mathcal{L}_{M}^{\sharp})$  by the isomorphism (4.5), so does  $\mathbb{H}_{c}(W_{J})$ .

4.4.6. Under the action of  $\mathbb{H}_{c}(W_{J})$  on  $\operatorname{ind}_{\mathfrak{m}}^{\mathfrak{l}}(\mathcal{L}_{M}^{\sharp})$ , the action of the polynomial part  $\mathbb{Q}[u] \otimes \operatorname{Sym}(\mathbb{E}_{\diamondsuit}^{\xi,*})$  has a similar description as in §4.3.1. This follows from the construction in [5, §4]. In particular, u acts via the equivariant parameter of  $\mathbb{G}_{m}^{\operatorname{dil}}$  and  $\delta/e$  acts via the equivariant parameter of  $\mathbb{G}_{m}^{\operatorname{rot}}$ .

4.4.7. We also need a compatibility of the construction in Theorem 1.2.3 with restriction from  $W_J$  to parabolic subgroups. Let  $J' \subset J$  be a subset. Choose  $A \in \Xi_F$ , and let  $F' = \prod_{J'}(A)$ . Then  $F \subset \partial F'$ , hence F' gives a parabolic subgroup  $Q^{F'}$  of L containing  $Q^A$ , and  $L' = L^{F'}$  is a Levi subgroup of  $Q^{F'}$ . Let  $Q^A_{F'}$  be the image  $Q^A \subset Q^{F'} \to L'$ ; then  $Q^A_{F'}$  is a parabolic subgroup of L' with Levi  $L^A \cong M$ . We have the transitivity of the induction

$$\operatorname{ind}_{\mathfrak{m}}^{\mathfrak{l}}(\mathcal{L}_{M}^{\sharp}) \cong \operatorname{ind}_{\mathfrak{l}'}^{\mathfrak{l}} \circ \operatorname{ind}_{\mathfrak{m}}^{\mathfrak{l}'}(\mathcal{L}_{M}^{\sharp}) \in \mathcal{D}_{L_{\diamond} \times \mathbb{G}_{m}^{\operatorname{dil}}}(\mathfrak{l}).$$

Theorem 4.4.5 gives an action of  $\mathbb{H}_c(W_{J'})$  on  $\operatorname{ind}_{\mathfrak{m}}^{\mathfrak{l}'}(\mathcal{L}_M^{\sharp})$ , which induces an action of  $\mathbb{H}_c(W_{J'})$  on  $\operatorname{ind}_{\mathfrak{m}}^{\mathfrak{l}}(\mathcal{L}_M^{\sharp})$  by the above isomorphism. The following statement follows from the construction in [5].

4.4.8. **Proposition.** The action of  $\mathbb{H}_{c}(W_{J'})$  on  $\operatorname{ind}_{\mathfrak{m}}^{\mathfrak{l}}(\mathcal{L}_{M}^{\sharp})$  induced from its action on  $\operatorname{ind}_{\mathfrak{m}}^{\mathfrak{l}'}(\mathcal{L}_{M}^{\sharp})$  coincides with the restriction of the action of  $\mathbb{H}_{c}(W_{J})$  on  $\operatorname{ind}_{\mathfrak{m}}^{\mathfrak{l}}(\mathcal{L}_{M}^{\sharp})$ constructed from Theorem 4.4.5.

4.5. Action of parabolic subalgebras. The goal of this subsection is to prove the following result.

4.5.1. **Proposition.** Let  $J \subset I^{\xi}$  be a proper subset and  $[F] \in W_{x/m} \setminus \mathfrak{F}_{J}^{\xi}$ . Then there is a canonical action of  $\mathbb{H}_{c,\eta/m}(W_J)$  on the perverse sheaf

$$\mathbb{I}_{[F]} := \bigoplus_{[A] \in \underline{\Pi}_J^{-1}[F]} {}^p \mathrm{HI}_{[A]}.$$

The proof occupies the rest of this subsection. We use the notation in  $\S4.4.3$ .

4.5.2. We have a morphism  $d: G_{\underline{0}} \overset{P_0}{\times} \mathfrak{p}_{\eta} \to [\mathfrak{p}_{\eta}/P_0] \to [\mathfrak{l}_{\eta}/L_0] \to [\mathfrak{l}/L]$ . We define  $\mathcal{X}$  by the following Cartesian diagram:



In other words, if we choose  $A \in \Xi_F$  and let  $U_0 = \ker(P_0 \to L_0)$ , then  $\mathcal{X} = \widetilde{\mathcal{X}}/L_0$ , where  $\widetilde{\mathcal{X}}$  consists of triples  $(v, gU_0, hQ^A) \in \mathfrak{g}_{\underline{\eta}} \times G_{\underline{0}}/U_0 \times L/Q^A$  such that  $\operatorname{Ad}(g^{-1})v$ lies in the preimage of  $\mathfrak{l}_{\eta} \cap \operatorname{Ad}(h)\mathfrak{q}^A \subset \mathfrak{l}_{\eta}$  under the projection  $\mathfrak{p}_{\eta} \twoheadrightarrow \mathfrak{l}_{\eta}$ , and  $\ell \in L_0$ acts on  $\widetilde{\mathcal{X}}$  diagonally by right translation by  $\ell^{-1}$  on  $G_{\underline{0}}/U_0$  and left translation by  $\ell$  on  $L/Q^A$ . There is a  $G_{\underline{0}}$ -action on  $\widetilde{\mathcal{X}}$  by diagonally acting on  $\mathfrak{g}_{\underline{\eta}}$  and  $G_{\underline{0}}/U_0$ , and this action descends to one on  $\mathcal{X}$ .

There is an action of  $\mathbb{G}_m^{\flat}$  on  $\mathcal{X}$  induced from the trivial action on  $G_{\underline{0}} \overset{P_0}{\times} \mathfrak{p}_{\eta}$  and the  $\mathbb{G}_m^{\flat}$ -action on  $[\dot{\mathfrak{l}}/L] = [\mathfrak{q}^A/Q^A]$ .

4.5.3. Lemma. The  $\mathbb{G}_m^{\flat}$ -fixed point locus of  $\mathcal{X}$  has a decomposition

$$\mathcal{X}^{\mathbb{G}_m^\flat} = \bigsqcup_{[A] \in W_{L_0} \setminus \Xi_F} G_{\underline{0}} \overset{P_0^A}{\times} \mathfrak{p}_{\eta}^A = \bigsqcup_{[A] \in W_{L_0} \setminus \Xi_F} X^A$$

*Proof.* Fix  $A \in \Xi_F$  which defines a parabolic subgroup  $Q = Q^A$  of L whose Levi subgroup  $L^A \cong M$ . Then the set  $\Xi_F \cong W_F/W_A = W_L/W_A$ . Therefore we reduce to showing that

(4.6) 
$$\mathcal{X}^{\mathbb{G}_m^\flat} = \bigsqcup_{[w] \in W_{L_0} \setminus W_L / W_A} X^{wA}.$$

The action of  $\mathbb{G}_m^{\flat}$  on  $\mathcal{X}$  lifts to  $\widetilde{\mathcal{X}}$ . Suppose  $(v, gU_0, hQ) \in \widetilde{\mathcal{X}}$  maps to a point in  $\mathcal{X}$  fixed by  $\mathbb{G}_m^{\flat}$ ; then for  $t \in \mathbb{G}_m^{\flat}$ , there is an  $\ell_0 \in L_0$  such that  $t \cdot (v, gU_0, hQ) = (v, g\ell_0^{-1}U_0, \ell_0 hQ)$ . This implies that  $g\ell_0^{-1}U_0 = gU_0$  hence  $\ell_0 = 1$ , i.e.,  $(v, gU_0, hQ) \in \widetilde{\mathcal{X}}^{\mathbb{G}_m^{\flat}}$ . Therefore  $\mathcal{X}^{\mathbb{G}_m^{\flat}} = \widetilde{\mathcal{X}}^{\mathbb{G}_m^{\flat}} / L_0$ .

If  $(v, gU_0, hQ) \in \widetilde{\mathcal{X}}^{\mathbb{G}_m^b}$ , then  $hQ \in L/Q$  is fixed under  $\mathbb{G}_m^b$ , i.e., there is a unique double coset  $[w] \in W_{L_0} \setminus W_L/W_A$  such that  $hQ \in L_0 wQ/Q \cong L_0/(L_0 \cap {}^wQ_0)$ . If we fix a representative w of [w] in  $W_L$ , then we may write  $hQ = \ell_0 wQ$  for a well-defined  $\ell_0 \in L_0/(L_0 \cap {}^wQ_0)$ . The condition for  $(v, gU_0, \ell_0 wQ)$  to be in  $\widetilde{\mathcal{X}}$ is that  $\operatorname{Ad}(g^{-1})v$  lies in the preimage of  $\mathfrak{l}_\eta \cap {}^{\ell_0 w}\mathfrak{q} = \operatorname{Ad}(\ell_0)(\mathfrak{l}_\eta \cap {}^w\mathfrak{q})$  under the projection  $\mathfrak{p}_\eta \to \mathfrak{l}_\eta$ . Note that for the  $\epsilon$ -spiral  $\mathfrak{p}_*^{wA}$ ,  $\mathfrak{p}_\eta^{wA}$  is exactly the preimage of  $\mathfrak{l}_\eta \cap {}^w\mathfrak{q}$  under the projection  $\mathfrak{p}_\eta \to \mathfrak{l}_\eta$ . Therefore the condition for  $(v, gU_0, \ell_0 wQ)$ to be in  $\widetilde{\mathcal{X}}^{\mathbb{G}_m^b}$  becomes  $\operatorname{Ad}(g^{-1})v \in \operatorname{Ad}(\ell_0)\mathfrak{p}_\eta^{wA}$ , or that  $v \in \operatorname{Ad}(g\ell_0)\mathfrak{p}_\eta^{wA}$ . Since g is well-defined up to right translation by  $U_0$ , and  $\ell_0$  is well-defined up to right translation by  $L_0 \cap {}^wQ_0$ , the element  $g\ell_0$  gives a well-defined coset in  $G_0/P_0^{wA}$ , because  $P_0^{wA} = U_0(L_0 \cap {}^wQ)$ . Therefore for fixed [w], the corresponding part of  $\mathcal{X}^{\mathbb{G}_m^b}$  consists of pairs  $(v, gP_0^{wA}) \in \mathfrak{g}_{\underline{\eta}} \times G_{\underline{0}}/P_0^{wA}$  such that  $v \in \operatorname{Ad}(g)\mathfrak{p}_{\eta}^{wA}$ , hence the description (4.6). To summarize we have a diagram in which the rhombus in the middle is Cartesian



4.5.4. Lemma. There is a canonical isomorphism in  $\mathcal{D}_{G_{0,\Diamond}\times\mathbb{G}_m^{\mathrm{dil}}}(\mathfrak{g}_\eta)$ 

$$f_{0,!}i^*p^*(\mathcal{L}_M^\sharp) \cong \bigoplus_{[A]\in\Pi_I^{-1}[F]} \mathbf{I}_{[A]}.$$

In particular, we have an isomorphism of perverse sheaves

$${}^{p}\mathrm{H}f_{0,!}i^{*}p^{*}(\mathcal{L}_{M}^{\sharp})\cong\mathbb{I}_{[F]}$$

*Proof.* The map  $\Pi_J : \mathfrak{F}^{\xi} \to \mathfrak{F}^{\xi}_J$  is  $W_{x/m}$ -equivariant, therefore we have a bijection  $\underline{\Pi}^{-1}([F]) \cong W_{x/m,F} \setminus \Pi_J^{-1}(F) = W_{L_0} \setminus \Pi_J^{-1}(F)$ . For  $A \in \Xi_F$  we have the maps

$$\mathfrak{g}_{\underline{\eta}} \xleftarrow{\alpha^A} X^A \xrightarrow{\gamma^A} [\mathfrak{l}^A/L^A] \cong [\mathfrak{m}/M] \;,$$

where  $\gamma^A$  is the restriction of  $p \circ i$  in diagram (4.7) to  $X^A$ . From the diagram (4.7) we see that the contribution of  $[A] \in W_{L_0} \setminus \Xi_F$  to the complex  $f_{0,!}i^*p^*(\mathcal{L}_M^{\sharp})$  is  $\alpha_!^A \gamma^{A,*} \mathcal{L}_M^{\sharp}$ . Therefore we reduce to showing the following two statements:

- (1) If  $A \in \mathfrak{F}^{\xi} \cap \Xi_F = \Pi_J^{-1}(F)$ , then there is a canonical isomorphism  $\alpha_!^A \gamma^{A,*} \mathcal{L}_M^{\sharp} \cong \mathbf{I}_A$ .
- (2) If  $A \notin \mathfrak{F}^{\xi}$ , then  $\alpha_{!}^{A} \gamma^{A,*} \mathcal{L}_{M}^{\sharp} = 0$ .

The morphism  $\gamma^A$  is the composition

$$\gamma^A: X^A \xrightarrow{\beta^A} [\mathfrak{l}^A_\eta/L^A_0] \xrightarrow{\iota} [\mathfrak{l}^A/L^A] \cong [\mathfrak{m}/M].$$

When  $A \in \mathfrak{F}^{\xi}$ , we have  $\iota^* \mathcal{L}_M^{\sharp} \cong \mathcal{L}^{A,\sharp}$ , and statement (1) follows from the definition of spiral induction  $I_A$ .

Now suppose  $A \in \Xi_F$  but  $A \notin \mathfrak{F}^{\xi}$ . Let  $j \in \mathbb{X}_*(T/Z_{L^A})$  be the element giving the  $\mathbb{Z}$ -grading  $\mathfrak{l}^A_*$  of  $\mathfrak{l}^A$ . Let  $\mathcal{O}^A \subset \mathfrak{l}^A$  be the nilpotent orbit corresponding to  $\mathcal{O}$  under the canonical isomorphism  $[\mathfrak{l}^A/L^A] \cong [\mathfrak{m}/M]$ . For any isomorphism  $L^A \cong M$ , the grading  $\mathfrak{m}_*$  induces a  $\mathbb{Z}$ -grading on  $\mathfrak{l}^A$ . Since the grading  $\mathfrak{m}_*$  is induced by  $\eta h_0/2$  for an  $\mathfrak{sl}_2$ -triple  $(e_0, h_0, f_0)$  in  $\mathfrak{m}$  with  $e_0 \in \mathcal{O}$ , the induced grading on  $\mathfrak{l}^A$  is induced by  $\eta h/2$  for an  $\mathfrak{sl}_2$ -triple (e, h, f) in  $\mathfrak{l}^A$  with  $e \in \mathcal{O}^A$ . Since  $A \notin \mathfrak{F}^{\xi}$ , the gradings  $\eta h/2$  and j are not conjugate under  $L^A$ .

We claim that  $\mathfrak{l}_{\eta}^{A} \cap \mathcal{O}^{A} = \emptyset$ , hence the vanishing of  $\iota^{*}\mathcal{L}_{M}^{\sharp}$  by the cleanness of  $\mathcal{L}_{M}$ , and statement (2) follows. We show the claim by contradiction. Suppose  $e' \in \mathfrak{l}_{\eta}^{A} \cap \mathcal{O}^{A}$ , we may complete it into an  $\mathfrak{sl}_{2}$ -triple (e', h', f') where  $h' \in \mathfrak{l}_{0}^{A}$  and  $f' \in \mathfrak{l}_{-\eta}^{A}$ . Then the adjoint action of h'/2 gives another  $\mathbb{Z}$ -grading  $\mathfrak{l}^{A} = \bigoplus_{n \in \mathbb{Z}} (\mathfrak{l}_{2n}^{h'} \mathfrak{l}^{A})$ , compatible with the original  $\mathbb{Z}$ -grading  $\mathfrak{l}_{*}^{A}$  (i.e.,  $\mathfrak{l}^{A}$  is the direct sum of  $\mathfrak{l}_{n}^{A} \cap \mathfrak{L}_{2n'}^{h'} \mathfrak{l}^{A}$  for  $n, n' \in \mathbb{Z}$ ) because  $h' \in \mathfrak{l}_{0}^{A}$ . We then have a third  $\mathbb{Z}$ -grading  $\mathfrak{l}^{A} = \bigoplus_{n \in \mathbb{Z}} \mathfrak{l}^{A}(n)$ 

given by the difference  $j - \eta h'/2$ :  $\mathfrak{l}^{A}(n) := \bigoplus_{n' \in \mathbb{Z}} (\mathfrak{l}_{\eta n'+n}^{A} \cap_{2n'}^{h'} \mathfrak{l}^{A})$ . Then e lies in the Levi subalgebra  $\mathfrak{l}^{A}(0)$  of  $\mathfrak{l}^{A}$ . Since  $\mathcal{O}^{A}$  is a distinguished nilpotent class of  $L^{A}$  (as it supports a cuspidal local system), we must have  $\mathfrak{l}^{A}(0) = \mathfrak{l}^{A}$ , hence j and  $\eta h'/2$  induce the same grading on  $\mathfrak{l}^{A}$ . However, this contradicts the assumption that j is not conjugate to any such grading. This completes the proof of statement (2) and hence of the corollary.

4.5.5. Finish of the proof of Proposition 4.5.1. By the diagram (4.7) and proper base change, we have

$$c_!d^*\mathrm{ind}^{\mathfrak{l}}_{\mathfrak{m}}(\mathcal{L}_M^{\sharp}) = c_!d^*a_!b^*(\mathcal{L}_M^{\sharp}) \cong f_!p^*(\mathcal{L}_M^{\sharp}) \in \mathcal{D}_{G_{\underline{0}}, \Diamond \times \mathbb{G}_m^{\mathrm{dil}}}(\mathfrak{g}_{\underline{\eta}}).$$

By Theorem 4.4.5, the complex  $\operatorname{ind}_{\mathfrak{m}}^{\mathfrak{l}}(\mathcal{L}_{M}^{\sharp})$  carries an action of  $\mathbb{H}_{c}(W_{J})$  under which the actions of u and  $\delta/e$  are given by the equivariant parameters of  $\mathbb{G}_{m}^{\operatorname{dil}}$  and  $\mathbb{G}_{m}^{\operatorname{rot}}$ , respectively (see §4.4.6). Therefore,  $f_{!}p^{*}(\mathcal{L}_{M}^{\sharp})$ , as an object in  $\mathcal{D}_{G_{\underline{0},\Diamond}\times\mathbb{G}_{m}^{\operatorname{dil}}}(\mathfrak{g}_{\underline{\eta}})$ , carries an action of  $\mathbb{H}_{c}(W_{J})$ .

Let  $\overline{f} : [\mathcal{X}/\mathbb{G}_m^{\flat}] \to \mathfrak{g}_{\underline{\eta}}$  be the induced map from f. Then  $\overline{f}_! p^*(\mathcal{L}_M^{\sharp}) \in \mathcal{D}_{G_{\underline{0}}}(\mathfrak{g}_{\underline{\eta}})$ carries an action of  $\mathbb{H}_c(W_J)$ . By the definition of  $\mathbb{G}_m^{\flat}$  and direct calculation, the map  $p: \mathcal{X} \to [\mathfrak{l}^A/L^A] \cong [\mathfrak{m}/M]$  is equivariant under the  $\mathbb{G}_m^{\flat}$  action on  $\mathcal{X}$  and its action on  $[\mathfrak{l}^A/L^A]$  through the embedding  $\mathbb{G}_m^{\flat} \hookrightarrow \mathbb{G}_m^{\mathrm{rot}} \times \mathbb{G}_m^{\mathrm{dil}}$  given by  $t \mapsto (t^{m/e}, t^{-\eta})$ . Therefore, both  $-u/\eta$  and  $\delta/m$  act as the equivariant parameter for  $\mathbb{G}_m^{\flat}$ , i.e., the  $\mathbb{H}_c(W_J)$ -action on  $\overline{f}_! p^*(\mathcal{L}_M^{\sharp}) \in \mathcal{D}_{G_{\underline{0}}}(\mathfrak{g}_{\eta})$  factors through the quotient

$$\mathbf{H}_v := \mathbb{H}_c(W_J)/(u + \eta \delta/m).$$

We denote the image of -u and  $\eta\delta/m$  in  $\mathbf{H}_v$  by v. In particular, the algebra  $\mathbf{H}_v$  acts on the  $\mathbb{Z}$ -graded  $G_{\underline{0}}$ -equivariant perverse sheaf  $\bigoplus_{n\in\mathbb{Z}}{}^{p}\mathbf{H}^{n}\overline{f}_{!}p^{*}(\mathcal{L}_{M}^{\sharp})$  on  $\mathfrak{g}_{\underline{\eta}}$ , with v acting as multiplication by  $\eta$  times the equivariant parameter of  $\mathbb{G}_{m}^{\flat}$ .

Let  $\mathcal{K} = p^*(\mathcal{L}_M^{\sharp}) \in \mathcal{D}_{G_{\underline{0}} \times \mathbb{G}_m^{\flat}}(\mathcal{X})$ . Let  $\mathcal{K}_0 = i^*\mathcal{K}$ . Applying Lemma 4.2.3 to the  $\mathbb{G}_m^{\flat}$ -equivariant morphism  $f : \mathcal{X} \to \mathfrak{g}_{\underline{\eta}}$ , using the specialization at  $v = \eta/m$ , we get the following isomorphism for n even and sufficiently large

(4.8) 
$${}^{p}\mathrm{H}^{n}(i^{*})_{v=\eta/m}: {}^{p}\mathrm{H}^{n}\overline{f}_{!}\mathcal{K} \xrightarrow{\sim} {}^{p}\mathrm{H}^{\mathrm{ev}}f_{0,!}\mathcal{K}_{0}$$

We define an action of  $\mathbf{H}_v$  on  ${}^{p}\mathbf{H}^{ev}f_{0,!}\mathcal{K}_0$  as follows. For  $a_{2s} \in \mathbf{H}_v$  homogeneous of degree  $2s \geq 0$ , choose large even n such that (4.8) holds. Then  $a_{2s}$  gives an action map

$$a_{2s}: {}^{p}\mathrm{H}^{\mathrm{ev}}f_{0,!}\mathcal{K}_{0} \xleftarrow{{}^{p}\mathrm{H}^{n}(i^{*})_{v=\eta/m}}{}^{p}\mathrm{H}^{n}\overline{f}_{!}\mathcal{K} \xrightarrow{a_{2s}}{}^{p}\mathrm{H}^{n+2s}\overline{f}_{!}\mathcal{K} \xrightarrow{{}^{p}\mathrm{H}^{n+2s}(i^{*})_{v=\eta/m}}{}^{p}\mathrm{H}^{\mathrm{ev}}f_{0,!}\mathcal{K}_{0}$$

It is easy to check that this map is independent of the choice of n and it defines an action of  $\mathbf{H}_v$  on  ${}^{p}\mathrm{H}^{\mathrm{ev}}f_{0,!}\mathcal{K}_0$ . Under this action v acts by  $\eta/m$ , therefore it descends to an action of  $\mathbf{H}_v/(v-\eta/m) \cong \mathbb{H}_{c,\eta/m}(W_J)$  on  ${}^{p}\mathrm{H}^{\mathrm{ev}}f_{0,!}\mathcal{K}_0$ . A similar argument gives an action of  $\mathbf{H}_v/(v-\eta/m) \cong \mathbb{H}_{c,\eta/m}(W_J)$  on  ${}^{p}\mathrm{H}^{\mathrm{odd}}f_{0,!}\mathcal{K}_0$ .

By Lemma 4.5.4,  ${}^{p}\text{H}f_{0,!}\mathcal{K}_{0} = {}^{p}\text{H}^{\text{ev}}f_{0,!}\mathcal{K}_{0} \oplus {}^{p}\text{H}^{\text{odd}}f_{0,!}\mathcal{K}_{0} \cong \mathbb{I}_{[F]}$ , therefore we have constructed an action of  $\mathbb{H}_{c,\eta/m}(W_{J})$  on  $\mathbb{I}_{[F]}$ . The action of the polynomial part of  $\mathbb{H}_{c,\eta/m}(W_{J})$  is the same as the one defined in §4.3.2. This finishes the proof of Proposition 4.5.1.

## 4.6. Finish of the proof of Theorem 1.2.3.

4.6.1. Changing J. Suppose  $J' \subset J$ ; then we have a map

$$\Pi_J^{J'}:\mathfrak{F}_{J'}^\xi\to\mathfrak{F}_J^\xi$$

sending  $F' \in \mathfrak{F}_{J'}^{\xi}$  to the unique facet  $F \in \mathfrak{F}_{J}^{\xi}$  on its boundary. We have  $\Pi_{J} = \Pi_{J}^{J'} \circ \Pi_{J'}$ . Let  $F' \in \mathfrak{F}_{J'}^{\xi}$  and  $F = \Pi_{J}^{J'}(F') \in \mathfrak{F}_{J}^{\xi}$ . Then  $\mathbb{I}_{[F']}$  is a direct summand of  $\mathbb{I}_{[F]}$  since  $\underline{\Pi}_{J'}([A]) = [F']$  implies  $\underline{\Pi}_{J}([A]) = [F]$ . On the other hand,  $\mathbb{H}_{c,\eta/m}(W_{J'})$  is a subalgebra of  $\mathbb{H}_{c,\eta/m}(W_{J})$ .

4.6.2. Proof of Theorem 1.2.3. For  $J' \subset J \subsetneq I^{\xi}$ , Proposition 4.4.8 implies that the inclusion  $\mathbb{I}_{[F']} \hookrightarrow \mathbb{I}_{[F]}$  is a map of  $\mathbb{H}_{c,\eta/m}(W_{J'})$ -modules (with the  $\mathbb{H}_{c,\eta/m}(W_{J'})$ -action on  $\mathbb{I}_{[F]}$  coming from its  $\mathbb{H}_{c,\eta/m}(W_J)$ -action by restriction). Since  $W_{\text{aff}}^{\xi}$  is generated by the subgroups  $W_J$  for various proper subsets  $J \subset I^{\xi}$ , Theorem 1.2.3 now follows from Propositions 4.5.1.

#### References

- Joseph Bernstein and Valery Lunts, Equivariant sheaves and functors, Lecture Notes in Mathematics, vol. 1578, Springer-Verlag, Berlin, 1994. MR1299527
- [2] Victor G. Kac, Infinite-dimensional Lie algebras, 3rd ed., Cambridge University Press, Cambridge, 1990. MR1104219
- G. Lusztig, Intersection cohomology complexes on a reductive group, Invent. Math. 75 (1984), no. 2, 205–272, DOI 10.1007/BF01388564. MR732546
- [4] George Lusztig, Fourier transforms on a semisimple Lie algebra over Fq, Algebraic groups Utrecht 1986, Lecture Notes in Math., vol. 1271, Springer, Berlin, 1987, pp. 177–188, DOI 10.1007/BFb0079237. MR911139
- [5] George Lusztig, Cuspidal local systems and graded Hecke algebras. I, Inst. Hautes Études Sci. Publ. Math. 67 (1988), 145–202. MR972345
- [6] George Lusztig, Classification of unipotent representations of simple p-adic groups, Internat. Math. Res. Notices 11 (1995), 517–589, DOI 10.1155/S1073792895000353. MR1369407
- [7] George Lusztig, Study of perverse sheaves arising from graded Lie algebras, Adv. Math. 112 (1995), no. 2, 147–217, DOI 10.1006/aima.1995.1031. MR1327095
- [8] G. Lusztig, Classification of unipotent representations of simple p-adic groups. II, Represent. Theory 6 (2002), 243–289, DOI 10.1090/S1088-4165-02-00173-5. MR1927955
- [9] George Lusztig and Zhiwei Yun, Z/m-graded Lie algebras and perverse sheaves, I, Represent. Theory 21 (2017), 277–321, DOI 10.1090/ert/500. MR3697026
- [10] George Lusztig and Zhiwei Yun, Z/m-graded Lie algebras and perverse sheaves, II, Represent. Theory 21 (2017), 322–353, DOI 10.1090/ert/501. MR3698042
- [11] Alexei Oblomkov and Zhiwei Yun, Geometric representations of graded and rational Cherednik algebras, Adv. Math. 292 (2016), 601–706, DOI 10.1016/j.aim.2016.01.015. MR3464031
- [12] Robert Steinberg, Endomorphisms of linear algebraic groups, Memoirs of the American Mathematical Society, No. 80, American Mathematical Society, Providence, R.I., 1968. MR0230728
- [13] Eric Vasserot, Induced and simple modules of double affine Hecke algebras, Duke Math. J. 126 (2005), no. 2, 251–323, DOI 10.1215/S0012-7094-04-12623-5. MR2115259

DEPARTMENT OF MATHEMATICS, MASSACHUSETTS INSTITUTE OF TECHNOLOGY, 77 MASSACHUSETTS AVENUE, CAMBRIDGE, MASSACHUSETTS 02139

Email address: gyuri@math.mit.edu

DEPARTMENT OF MATHEMATICS, MASSACHUSETTS INSTITUTE OF TECHNOLOGY, 77 MASSACHUSETTS AVENUE, CAMBRIDGE, MASSACHUSETTS 02139

Email address: zyun@mit.edu